

Universidad Miguel Hernández de Elche

Programa de Doctorado en Deporte y Salud

Effects of injury prevention programs on physical performance and neuromuscular control in youth soccer

Doctoral Thesis

A dissertation presented by

Carlos Pomares Noguera

Directed by

Dr. José Luis López Elvira and codirected by Dr. Francisco Ayala Rodríguez

Elche, 2021





The current doctoral thesis, titled *Effects of injury prevention programs on physical performance and neuromuscular control in youth soccer*, is presented with the following general quality indicators:

Article 1:

Ayala, F., Calderón-López, A., Delgado-Gosálbez, J. C., Parra-Sánchez, S., **Pomares-Noguera, C.**, Hernández-Sánchez, S., López-Valenciano, A. & De Ste Croix, M. (2017). Acute effects of three neuromuscular warm-up strategies on several physical performance measures in football players. *PloS One*, 12(1), e0169660. https://doi.org/10.1371/journal.pone.0169660

Journal indexed in the *Journal Citation Reports catalogue* with an impact factor of 3.234 (Q1).

Article 2:

Ayala, F., **Pomares-Noguera, C.**, Robles-Palazón, F.J., García-Vaquero, M.P, Ruiz-Pérez, I., Hernández-Sánchez, S., & De Ste Croix, M. (2017). Training effects of the FIFA 11+ and Harmoknee on several neuromuscular parameters of physical performance measures. *International Journal of Sports Medicine*, 38(04), 278-289. https://doi.org/10.1055/s-0042-121260

Journal indexed in the *Journal Citation Reports catalogue* with an impact factor of 2.410 (Q2).

Article 3:

Pomares-Noguera, C., Ayala, F., Robles-Palazón, F.J., Alomoto-Burneo, J.F., López-Valenciano, A., Elvira, J.L., Hernández-Sánchez, S. & De Ste Croix, M. (2018).

Training effects of the FIFA 11+ Kids on physical performance in youth football players: a randomized control trial. *Frontiers in Pediatrics*, 6, 40. https://doi:10.3389/fped.2018.00040

Journal indexed in the *Journal Citation Reports catalogue* with an impact factor of 2.349 (Q1).





El Dr. Francisco Ayala Rodríguez junto al Dr. José Luis López Elvira, Titular de la Universidad Miguel Hernández de Elche hacen constar que el trabajo de investigación titulado "*Effects of injury prevention programs on physical performance and neuromuscular control in youth soccer*" realizado por el doctorando D. Carlos Pomares Noguera, ha sido supervisado bajo su dirección y autorizado para su defensa de Tesis Doctoral en esta Universidad ante el tribunal correspondiente.

Lo que firman para los efectos oportunos en:

Elche, a 25 de junio de 2021

Fdo.: José Luis López Elvira (director) Fdo.: Francisco Ayala Rodríguez (codirector)

Universidad Miguel Hernández de Elche





El Dr. Francisco Javier Moreno Hernández, coordinador del programa de doctorado en Deporte y Salud de la Universidad Miguel Hernández de Elche.

AUTORIZA:

Que el trabajo de investigación titulado: "Effects of injury prevention programs on physical performance and neuromuscular control in youth soccer" realizado por D. Carlos Pomares Noguera bajo la dirección del Dr. José Luis López Elvira y la codirección del Dr. Francisco Ayala Rodríguez, sea defendido como Tesis Doctoral en esta Universidad ante el tribunal correspondiente.

Lo que firma para los efectos oportunos en:

Elche, a 25 de junio de 2021

Fdo.: Francisco Javier Moreno Hernández

Coordinador del programa de doctorado

Universidad Miguel Hernández de Elche





UNIVERSIDAD MIGUEL HERNÁNDEZ DE ELCHE

Programa de Doctorado en Deporte y Salud

Título de la Tesis:

EFFECTS OF INJURY PREVENTION PROGRAMS ON PHYSICAL PERFORMANCE AND NEUROMUSCULAR CONTROL IN YOUTH SOCCER

Tesis Doctoral presentada por:

D. Carlos Pomares Noguera

Dirigida por el Dr. Francisco Ayala Rodríguez y del Dr. Jose Luis López Elvira

El director

El codirector

El doctorando

Fdo.: Jose Luis López Elvira

Fdo.: Francisco Ayala

Fdo.: Carlos Pomares

Rodríguez

Noguera

Elche, a 25 de junio de 2021



UNIVERSIDAD MIGUEL HERNÁNDEZ DE ELCHE

Programa de Doctorado en Deporte y Salud

Effects of injury prevention programs on physical performance and neuromuscular control in youth



Doctoral Thesis

A dissertation presented by Carlos Pomares Noguera

Director: Dr. José Luis López Elvira and codirector: Dr. Francisco Ayala Rodríguez



A mi yaya, María.





AGRADECIMIENTOS

Aunque durante el largo camino que ha llevado el desarrollo de la presente tesis haya intentado agradecer a cada una de las personas que me han ayudado a llegar hasta aquí, creo que es necesario agradecer y recordar a aquellos que han hecho posible todo esto.

Voy a empezar por la persona más importante de la tesis. No, evidentemente no soy yo. Esa persona es Fran Ayala. Líder incansable a la que en algunas ocasiones llamamos boss y no precisamente por la definición de jefe (que arroja autoridad y poder), sino todo lo contrario, porque desprende humildad y siempre tiene en cuenta a su equipo (y qué suerte formar parte de él). Más que un jefe, es un líder; un ejemplo en todo lo que hace. Todo comenzó cuando tenía que elegir temática para mi trabajo final de máster y repasando líneas de investigación vi una que se basaba en la eficacia de programas de entrenamiento en futbolistas jóvenes, a la cual acompañaba su nombre. Recuerdo como si fuera ayer el momento de elegir dicha línea por lo claro que lo tenía, aunque la veía tan difícilmente accesible que mi esperanza no era la más alta del mundo. Otro de los momentos que recuerdo con claridad es cuando, además de realizar la preinscripción para entrar al programa de doctorado de otras universidades a las cuales tuve la suerte de haber sido aceptado también, me dio por hacerle caso a una persona que me dijo: "Carlos, realiza la preinscripción aquí en Elche también, nunca se sabe". Evidentemente, esa persona que me empujó a realizar dicha preinscripción no fue otra que Fran, lo cual, aunque él no lo sepa, le sigo agradeciendo día tras día. Sabes de sobra que este trabajo es tuyo. Gracias de nuevo, boss.

Esta tesis también es de la familia del CID, comenzando por José Luis, mi otro director, al cual lo tuve de profesor en la carrera y desde el primer momento me sorprendió lo inteligente y buena persona que es. Además de esto, y aunque no hemos podido pasar mucho tiempo juntos de manera presencial debido a mi trabajo, remarcar su sentido del humor, el cual lo poco que lo conozco, me encanta. Acordarme también de Fran Vera, director del laboratorio de biomecánica, del cual el primer recuerdo que guardo de él fue ese lanzamiento de balón medicinal que casi saca del pabellón en el que realizamos las pruebas de acceso a la carrera (desgraciadamente también me acuerdo cuando yo lancé dicho balón...). Gracias también por ayudarme y ser un referente en el ámbito académico.

También quisiera acordarme de toda esa gente del CID que me ha echado una mano durante este camino de manera desinteresada. Gracias a Álvaro, Adri, Amaya, Belén, Carla, Casto, David, Diego, Miguel, Rafa, Pedro, Sarabia, Tomás, Xoxe...

Y qué decir de mis compañeros de laboratorio, los más cercanos y que más tiempo han estado conmigo. Gracias Alejandro, por tu ejemplo y por prestarme tu ayuda antes siquiera de que yo me pusiera manos a la obra. Gracias Maripili por tu ejemplo y constancia (sobre todo aguantándonos al resto). Gracias Sergio por tu alegría y acogida desde el primer día. Gracias a la pareja murciana que me acompañó desde el máster, Fran y May. Fran, gracias por tu inestimable ayuda (ya sea midiendo jugadores para el que más tarde sería nuestro primer artículo, como echándome una mano en cuestiones estadísticas de última hora). Y, sobre todo, gracias a Ignacio. Es broma, Iñaki. Cualquier palabra se queda corta para agradecerte todo lo que me has ayudado desde el primer día que pisé el laboratorio, enseñándome procedimientos, protocolos, ayudándome a preparar mi primer seminario para alumnos de máster y un largo etcétera. He de decir que echo de menos tus empanadillas que hacían las mediciones de horas y horas mucho más amenas. Gracias, maestro.

No quisiera olvidarme tampoco de mi amigo David C., el cual conocí en la carrera, hicimos el máster juntos y hasta el día de hoy, al cual considero un apoyo y amigo de los de verdad.

Gracias también a la Solfamilia que hace que mis ratos fuera del ámbito académicolaboral sean revitalizantes. Gracias David y Pedro por estar siempre.

Tampoco puedo olvidarme de los entrenadores y preparadores físicos que han hecho posible la recogida de los datos de la presente tesis. Gracias en especial al At. Crevillente y a Torcua, el cual desde el primer momento me trató de diez y me prestó a sus jugadores para lo que yo necesitara. No me olvido de tu ayuda.

También mencionar a la familia Lucentina que ha estado conmigo en estos últimos dos años. Gracias a todos y cada uno de ellos por dar sentido a la palabra familia. Yasmina y José, gracias en especial por confiar en mí y demostrármelo desde el primer día.

Por último y no por ello menos importante, sino todo lo contrario, dar gracias a mi familia. A Tere y Juan, por ser los mejores padres que uno pudiera tener y, aunque en ocasiones me enfado con vosotros, os quiero con locura. A Maite, la mejor hermana que cualquiera pudiera imaginar. Gracias por ser el ejemplo más bonito que los papás me han dado. Gracias también a Simba y Nala, que, aunque de momento no hay aprendizaje desarrollado para que un perro pueda leer esto, no me perdonaría que no quedaseis reflejados aquí ya que sois familia. Y, sobre todo, gracias a ti yaya. Gracias por haber estado conmigo más de la mitad de mi vida y haberme enseñado tanto y tan bien. Dicen que las personas tenemos secretos que se quedan dentro de cada uno de nosotros para toda la vida. Mi mayor secreto es que nadie sabrá lo mucho que te quise y te querré.





TABLE OF CONTENTS

Table of Contents 19
List of Figures
List of Tables
List of Supplementary Files
Preface
Abstract
Resumen
Abbreviations
Chapter 1. General Introduction
1.1. Health benefits of regular physical activity and sport participation in
children and adolescents
1.2. Epidemiology of injuries in youth soccer
1.3. Structured warm-up programs for injury prevention in youth soccer 40
1.4. Effects of FIFA 11+, Harmoknee and FIFA 11+ Kids programs on
physical performance and neuromuscular risk factors
1.5. Lines of action of the present PhD thesis
Chapter 2. Research Objectives and Hypotheses 57
2.1. General objectives 59
2.2. Specific objectives 59
2.3. Research hypotheses
Chapter 3. Study 1 63
3.1. Abstract
3.2. Introduction
3.3. Material and methods

3.4. Results	79
3.5. Discussion	83
3.6. Practical applications	86
3.7. Supplementary files	88
Chapter 4. Study 2	91
4.1. Abstract	93
4.2. Introduction	94
4.3. Methods	95
4.4. Results	105
4.5. Discussion	111
4.6. Supplementary files	116
Chapter 5. Study 3	121
5.1. Abstract	123
5.2. Introduction	125
5.3. Materials and methods	127
5.4. Results	135
5.5. Discussion	136
5.6. Perspectives and limitations	141
5.7. Conclusion	142
Chapter 6. Epilogue	143
6.1. General conclusions	145
6.2. Limitations and recommendations for future research	147
Capítulo 7. Epílogo	149
7.1. Conclusiones generales	151
7.2. Limitaciones y recomendaciones para futuras investigaciones	153
Chapter 8. References	155

LIST OF FIGURES

Figure 1.1. Graphical representation of the FIFA 11+ program.

Figure 1.2. Graphical representation of the FIFA 11+ Kids program.

Figure 3.1. Schematic representation of the study 1 design.

Figure 3.2. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the FIFA 11+ and Harmoknee. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

Figure 3.3. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the Dynamic warm-up and Harmoknee. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

Figure 3.4. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the Dynamic warm-up and FIFA 11+. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

Figure 4.1. Schematic representation of the study 2 design.

Figure 4.2. Net effects (expressed as percentage) of the intervention on neuromuscular parameters of physical performance for the "FIFA 11+" and control group.

Figure 4.3. Net effects (expressed as percentage) of the intervention on neuromuscular parameters of physical performance for the Harmoknee and control group.

Figure 5.1. Schematic representation of the study 3 design.

Figure 5.2. Standardized mean differences (90% confidence interval) between the FIFA 11+ Kids and control group in the physical performance parameters analyzed.



LIST OF TABLES

Table 1.1. Summary of the major benefits of childhood and adolescence physical activity.

Table 1.2. Description of the Harmoknee warm-up program.

Table 1.3. Summary of studies investigating the injury-preventing effects of FIFA 11+, FIFA 11+ Kids and Harmoknee programs.

Table 1.4. Training (chronic) effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs on measures of physical and neuromuscular performance in youth soccer players.

Table 3.1. Description of the FIFA 11+ warm-up program.

Table 3.2. Description of the Dynamic warm-up program.

Table 3.3. Post-intervention (Dynamic warm-up, Harmoknee and FIFA 11+) results (mean ± standard deviation [SD]) for physical performance outcomes.

Table 4.1. Baseline and post-intervention (FIFA 11+ and Control) results (mean ± standard deviation [SD]) for physical performance outcomes. The percentage differences between pre and post-test average values are also reported.

Table 4.2. Baseline and post-intervention (Harmoknee and Control) results (mean ± standard deviation [SD]) for physical performance outcomes. The percentage differences between pre and post-test average values are also reported.

Table 5.1. Demographic variables for the children soccer players (all values are mean ± SD).

Table 5.2. Descriptive pre-tests statistics (mean \pm standard deviation) and differences between pre-test 2 and pre-tests 1 (mean \pm 95% confidence intervals).

Table 5.3. Baseline (pre-test) and post-intervention (FIFA 11+ and Control) results (mean ± standard deviation [SD]) for physical performance outcomes. The differences between preand post-test average values are also reported (mean and 95% confidence intervals [CI]).



LIST OF SUPPLEMENTARY FILES

Supplementary file 3.1. The passive hip flexion (passive straight leg raise test [figure a]), extension (Thomas test [figure b]) and abduction (passive hip abduction with knee flexed over the edge of the plinth test [figure c]); knee flexion (Modified Thomas test [figure d]) and ankle dorsiflexion (weight-bearing lunge with knee extended [figure e] and flexed [figure f] tests) range of motions assessment.

Supplementary file 3.2. Isokinetic testing position.

Supplementary file 3.3. Dynamic postural control assessment (anterior [figure a], posteromedial [figure b] and posterolateral [figure c] directions).

Supplementary file 4.1. Dynamic postural control assessment (anterior [a], posteromedial [b] and posterolateral [c] directions).

Supplementary file 4.2. The passive hip flexion (passive straight leg raise test [a]), knee flexion (Modified Thomas test [b]) and ankle dorsiflexion (weight-bearing lunge with knee extended test [c]) range of motion assessment.

Supplementary file 4.3. Single legged hop limb symmetry assessment (single hop for distance [a]; and triple hop for distance [b]).

Supplementary file 4.4. Illinois agility run dimensions and completion route.



PREFACE

The doctoral thesis that the reader is holding in his/her hands is the result of the last five years of research on the exciting topic of injury risk mitigation in youth soccer by some members of the research group Biomechanics for Health and Sport Performance purposes (BIOMEC) (code: 10/0062E) based at Miguel Hernández University of Elche.

The thesis presented here is divided into four main parts. Part I provides an overview of the available standardized warm-up programs designed with the aim of reducing soccerrelated injuries in youth players, the issues that this thesis seeks to address, and the research objectives and hypotheses. Part II includes the three studies developed to address the research questions mentioned in Part I. This part is further divided into three chapters (numbers 3 to 5) which present each of the studies separately to the reader. Part III presents the main conclusions drawn from this thesis and sets out future lines of research. Finally, Part IV provides the reference list of the studies cited throughout the different chapters.

However, the scientific production of this research project is not limited to this document. During the five years up to the completion of this thesis, the author has also collaborated in other research projects and has presented the main findings of this thesis in different International and National Congresses. This period has enabled the author to complete also one international research stay in United Kingdom (one month in the University of Gloucestershire) to improve his scientific skills. Finally, the author and his research group have been awarded with some mentions and distinctions during these years thanks to their contributions to this line of investigation.

I hope you enjoy reading this thesis as much as I have enjoyed these years of work.



ABSTRACT

The practice of sports on a regular basis has proven to be a vital component for maintaining an active and healthy lifestyle that lasts a lifetime, reducing the risk of suffering from numerous non-communicable diseases (e.g., cardiovascular disease [atherosclerosis], certain types of cancer [breast and colon], metabolic disorders [diabetes mellitus] and respiratory disease [chronic obstructive pulmonary disease]) and helps maintain an optimal state of physical, psychological and social well-being in children and adolescents.

In this sense, given that soccer (i.e., associate football) is the most popular sport in the world, being most of its practitioners under the age of 18, it could have great potential to induce the aforementioned effects in young people. However, the high physical demands of the game of soccer alongside the decline in essential motor performance that often occurs during the years of maximal rate of growth and maturation may place young players at high risk of suffering an injury in comparison with their non-athlete counterparts. Thus, soccer-related injuries can counter the health-related beneficial effects of sports participation at a young age if a child or adolescent is unable to continue to participate because of the residual effects of injury.

Within the last years, a number of neuromuscular training programs have been designed with the aim of preventing and reducing the number and severity of soccerrelated injuries in children and adolescents. The effectiveness of these neuromuscular training programs to reduce non-contact and overuse injuries in young soccer players has been documented in systematic reviews and meta-analyses. Although the main purpose of these neuromuscular training programs is injury prevention, knowing the effects (acute and chronic) elicited by them on physical performance and neuromuscular control measures could help identify the potential mechanism underlying the reported reduction in the incidence of injuries. Therefore, the main objective of this doctoral thesis was to analyze the acute and chronic effects of some neuromuscular training programs on parameters of physical performance and neuromuscular training programs on parameters of achieve this objective, three randomized controlled studies were carried out in a sample of 83 young soccer players.

On the one hand, the results of these studies report that neither the FIFA 11+ nor the Harmoknee programs elicit superior acute effects on measures of physical performance (with the exception of the 10 and 20 m sprint time) and neuromuscular control to the dynamic warm-up routines currently performed by most soccer players prior to training sessions and matches. On the other hand, the studies that comprise this doctoral thesis also show that, unlike traditional dynamic warm-up routines, when the FIFA 11+ Kids and FIFA 11+ programs are systematically performed at the start of training sessions and matches produce clinically relevant and statistically significant chronic improvements in some measures of physical performance (10 and 20 m sprint time and jump height) and neuromuscular control (unilateral dynamic balance, bilateral strength ratios) in children and adolescent soccer players, respectively.

Therefore, the main findings of this doctoral thesis suggest that the mechanisms behind the promising reduction in injury risk by adhering to the FIFA 11+ kids and FIFA 11+ programs in young soccer players might be particularly associated with enhancements in tendon stiffness, balance, coordination and physical competency.

Keywords: injury, injury prevention, children, adolescents, motor competence, warm-up, agility, dynamic stability, performance, jumping height.

RESUMEN

La práctica deportiva de manera regular ha demostrado ser un componente de vital importancia para mantener un estilo de vida activo y saludable que perdure toda la vida, que reduzca el riesgo de sufrir numerosas enfermedades crónicas no trasmisibles (ej., enfermedad cardiovascular [aterosclerosis], determinados tipos de cáncer [mama y colon], trastornos metabólicos [diabetes mellitus] y enfermedades respiratorias y pulmonares) y ayude a mantener un estado óptimo de bienestar físico, psíquico y social en niños y adolescentes.

En este sentido, y dado que el fútbol es el deporte más popular en todo el mundo, siendo la mayor parte de sus practicantes menores de 18 años, éste podría poseer un gran potencial para inducir los efectos anteriormente citados en los jóvenes. Sin embargo, las altas demandas físicas que el fútbol impone a sus practicantes, unido a los desajustes en el control motor propios del proceso de crecimiento y maduración, colocan a los jóvenes jugadores en una situación más vulnerable para sufrir una lesión en comparación con sus homólogos no deportistas. Así, las lesiones podrían contrarrestar los efectos positivos que la práctica del fútbol presenta sobre la salud si un niño o adolescente no es capaz de continuar adherido a la misma como consecuencia de los efectos residuales derivados de estas.

En los últimos años, numerosos programas de entrenamiento neuromuscular (ej., FIFA 11+ y Harmoknee) han sido diseñados con el propósito de prevenir y reducir el número y la severidad de las lesiones asociadas a la práctica del fútbol en niños y adolescentes. La efectividad de estos programas de entrenamiento neuromuscular para reducir las tasas de lesiones por no contacto y sobreuso ha sido documentada en jóvenes jugadores por estudios de revisión y meta-análisis. Aunque el principal propósito de los programas de entrenamiento neuromuscular es la prevención de lesiones, conocer los efectos (agudos y crónicos) ocasionados por estos programas sobre el rendimiento físico y control neuromuscular podría ayudar a identificar el mecanismo potencial que subyace a la evidenciada reducción en la incidencia de lesiones. Por lo tanto, el objetivo principal de esta tesis doctoral fue analizar los efectos agudos y crónicos de ciertos programas de entrenamiento neuromuscular sobre parámetros de rendimiento físico y control neuromuscular en jugadores jóvenes de fútbol. Con el fin de conseguir estos objetivos, se realizaron tres estudios experimentales controlados y aleatorizados en una muestra de 83 jugadores jóvenes de fútbol.

Los resultados de los estudios experimentales informan, por un lado, de que los programas FIFA 11+ y Harmoknee no producen efectos agudos sobre medidas de rendimiento físico (con excepción del tiempo en el sprint de 10 y 20 m) y control neuromuscular distintos a los observados por los programas tradicionales de ejercicios dinámicos que muchos jugadores adolescentes de fútbol llevan a cabo al inicio de las sesiones de entrenamiento y los partidos. Por otro lado, estos estudios también muestran que la realización sistemática de los programas FIFA 11+ Kids y FIFA 11+ al inicio de las sesiones de entrenamiento y en sustitución de las tradicionales rutinas de calentamiento dinámico produce una mejora clínicamente relevante y estadísticamente significativa en ciertas medidas de rendimiento físico (tiempo en el sprint de 10 y 20 m y altura de salto) y control neuromuscular (estabilidad dinámica unipodal, ratios bilaterales de fuerza) en niños y adolescentes jugadores de fútbol, respectivamente.

Por lo tanto, los principales hallazgos de la presente tesis doctoral sugieren que el mecanismo que podría estar detrás de la reducción en el riesgo de lesión que la adherencia a los programas FIFA 11+ Kids y FIFA 11+ produce en jugadores jóvenes de fútbol podría estar particularmente asociado a mejoras en la rigidez tendinosa, estabilidad, coordinación y competencia física.

Palabras clave: lesión, prevención de lesiones, niños, adolescentes, competencia motriz, calentamiento, agilidad, estabilidad dinámica, rendimiento, altura de salto.

ABBREVIATIONS

3G: Third generation **ACL:** Anterior cruciate ligament **ASIS:** Anterior superior iliac spine **CI:** Confidence interval CG: Control group **CMJ**: Countermovement jump **CON:** Concentric **COP:** Center of pressure CV: Coefficient of variation **DJ**: Drop jump DWU: Dynamic warm-up **ECC:** Eccentric F-MARC: FIFA Medical Assessment and Research Centre FIFA: Fédération Internationale de Football Association H/Q: Hamstring to quadriceps strength ratio H/QCONV: Conventional hamstring to quadriceps strength ratio H/Qconv₆₀: Conventional hamstring to quadriceps strength ratio at 60^o/s H/QCONV180: Conventional hamstring to quadriceps strength ratio at 180º/s H/QFUNCT: Functional hamstring to quadriceps strength ratio H/QFUNC60: Functional hamstring to quadriceps strength ratio at 60^e/s H/QFUNC180: Functional hamstring to quadriceps strength ratio at 180º/s H/QFUNC30/240: Functional hamstring to quadriceps strength ratio between the torques 30°/s (eccentrically) and 240º/s (concentrically)

IG: Intervention group
KIPP: Knee injury prevention program
LESS: Landing error scoring system
LSI: Limb symmetry index
PAP: Post-activation potentiation
PEP: Prevent injury and enhance performance program
PHV: Peak height velocity
RCT: Randomized control trial
ROM: Range of motion
RR: Relative risk
RSA: Repeated sprint ability
SD: Standard deviation
SEM: Standard error of measure
SLJ: Standing long jump
SSC: Stretch-shortening cycle
TE: Typical error of measure
VJ: Vertical jump
WHO: World Health Organization

CHAPTER 1

General Introduction


1

GENERAL INTRODUCTION

1.1 HEALTH BENEFITS OF REGULAR PHYSICAL ACTIVITY AND SPORT PARTICIPATION IN CHILDREN AND ADOLESCENTS

Both regular physical activity and physical fitness are considered to be important prerequisites for the health of children (5-14 years) and adolescents (>14-18 years) [1–3]. Thus, the benefits provided by regular physical activity for both children and adolescents have been observed in several health domains (table 1.1) that include physiology, neuroimmune functioning, recovery from disease and disability, increased muscular strength and cardiorespiratory fitness, angiogenesis and neurogenesis, reduced risk for chronic disease, improved self-esteem and psychological well-being as well as reduced stress, anxiety and depression [4]. Regular physical activity may also reduce the cost of managing common noncommunicable diseases that are associated with inadequate physical activity in youth population (e.g., obesity, low-back pain) and fosters better performance in schools (World Health Organization [WHO] [5]).

Therefore, promoting a physically active lifestyle across youth is not only an imperative for public health but also vitally important for the social and economic development of the countries [1]. In this sense, the new WHO 2020 guidelines on physical activity and sedentary behavior strongly recommend that children and adolescents:

- Should do at least an average of 60 min per day of moderate-to-vigorous intensity, mostly aerobic, physical activity, across the week.
- Vigorous-intensity aerobic activities, as well as those that strengthen muscle and bone should be incorporated at least 3 days a week.

Sport may serve as a suitable and enjoyable physical activity setting for achieving these recommended levels of practice for health in youth population. Furthermore, participation in organized youth sport has been positively associated with higher levels of physical activity in adulthood [6,7].

Table 1.1: Summary of the major benefits of childhood and adolescence physical activity.

	1. Health benefits during childhood and adolescence							
•	Maintenance of energy balance and prevention of overweight and obesity.							
•	Promoted healthy growth and development of the musculoskeletal and cardiovascular system.							
•	Reduced risk factors for:							
	- Cardiovascular disease.							
	- Type 2 diabetes.							
	- Hypertension.							
	- Hypercholesterolemia.							
•	Improved mental health and psychological well-being through:							
	- Reduced anxiety and stress.							
	- Reduced depression.							
	- Higher self-esteem.							
	- Improved cognitive function.							
•	Improved social interaction.							
	2. Improved health during adulthood							
•	Reduced probability of becoming obese during adulthood.							
•	Reduced morbidity and mortality from chronic disease during adulthood.							
•	Improved bone mass reduces likelihood of osteoporosis in later life.							
	3. Establishment of lifetime activity patterns							
	Improved likelihood of becoming an active adult.							

In this sense, as soccer (i.e., associated football) is the world's most popular sport with more than 270 million registered players (at amateur and elite levels) in 2006 [8], most of them male younger than 18 years, it might have a great potential to induce beneficial health effects and to support an active and healthy lifestyle during the life-course [9]. However, sport (including soccer) has been considered a double-edged sword [10] as it may have also some negative aspects (mainly at elite levels) associated. These negative aspects should be mitigated to avoid that its health benefits elicited in children and adolescents could be dramatically dismissed or even overwhelmed by them (e.g., risk of failure leading to poor mental health [11,12], risk of injury [13,14], sudden death [15–17], eating disorders [18], burnout [19], and exercise-induced gastrointestinal tract discomfort [20]). Among the most frequently reported negative aspects associated with youth soccer it

can be found an elevated risk of injury [21]. This heighten injury risk, but particularly the standardized strategies designed to mitigate it, is the object of study of the present doctoral thesis. Although soccer practice increases the risk of injury in both male and female players (independently of their age and level of play), the current thesis is focused on youth male soccer players.

1.2 EPIDEMIOLOGY OF INJURIES IN YOUTH SOCCER

Soccer requires players to perform a substantive number of repeated high intensity movements such as sudden accelerations and decelerations, rapid changes of direction, tackling, jumping and landing tasks [22,23]. These high-intensity situations result in a notable increase in injury risk and are especially relevant in children where individual growth and maturation may predispose youth players to a higher risk [24–27].

In this sense, Rössler et al. [28], in a two-season descriptive epidemiological study carried out in players aged from 7 to 12 years, documented injury incidence rates of 0.6 (95% confidence interval [CI] = 0.5 - 0.7) and 4.6 (95% CI = 4.0 - 5.2) injuries per 1000 hour of soccer exposure during training sessions and match play, respectively. In addition, this study found that most (76.3%) injuries were located in the lower limbs and the most common types of injuries were joint and ligament injuries (mainly in the knee and ankle joints) (30.5%), contusions (22.5%), muscle and tendon injuries (mainly thigh [hamstrings, abductors and quadriceps] muscles) (18.5%), and fractures and bone injuries (15.4%). These authors also noticed a positive relationship between injury incidence and players' age.

For its part, a recently conducted systematic review and meta-analysis on epidemiology of injuries in youth soccer (under 19 years old) reported for male adolescent players (12 to 19 years) training and match injury incidence rates of 2.8 (95%CI = 2.0 - 3.5) and 14.4 (95%CI = 11.0 - 17.8) per 1000 hours of exposure, respectively. In addition, lower extremity injuries had the highest incidence rates (4.1 injuries/1000 hours of exposure) and the most common types of injuries were muscle/tendon (1.9 injuries/1000 hours of exposure) and joint (non-bone) and ligament injuries (1.0 injuries/1000 hours of exposure) [29].

From the epidemiological data just described, it might be stated that injury incidence rates are higher during the adolescence compared to the pre-pubertal stage [30]. In

particular, injury incidence seems to be aligned to peak height velocity (adolescence), when rapid disproportional growth is evident [31]. Likewise, the injury pattern seems to be different in children and adolescents, whereby younger soccer players seem to have more fractures and bone stress, fewer strains and sprains, and more injuries of the upper body than adolescent players (12-18 years).

The most frequently diagnosed soft tissue injuries (muscle, tendon and ligament) in youth male soccer players (i.e., thigh muscle strains, knee and ankle ligaments sprains and tears) may: a) lead to moderate absence from sport participation, b) negatively impact on short and long-term athlete development, c) cause long-term disability (development of knee osteoarthritis in adulthood) and d) increase medical costs [32,33]. Thus, soccer-related injuries can counter the health- related beneficial effects of sports participation at a young age if the child or adolescent is unable to continue to participate because of the residual effects of injuries [24].

Consequently, there is a clear need for implementing appropriated measures to prevent and reduce the number and severity of soccer-related injuries in youth male players.

1.3 STRUCTURED WARM-UP PROGRAMS FOR INJURY PREVENTION IN YOUTH SOCCER

Within the last decade, a substantive number of studies have been published describing different warm-up programs (e.g., FIFA 11 [34], FIFA 11+ [35], FIFA 11+ Kids [28], Harmoknee [36], Knäkontroll [37], KIPP [38] and PEP [39]) aimed at reducing the number and severity of sport-related injuries in children but mainly adolescent players. All of these programs include running exercises and specific dynamic movements focusing on enhancing physical competence and performance, as it has been suggested as a primary injury risk factor in youth team sport athletes [40,41] and are based on the injury profile and biological status (i.e., maturation phase) of the target population (children [<12 years] and adolescents [from 12 to 18 years]).

The three most popular standardized injury prevention programs that have been specifically developed to be implemented as pre-exercise warm-ups in youth soccer are described as follow.

FIFA 11+

The FIFA 11+ was developed in 2006 by the FIFA Medical Assessment and Research Centre (F -MARC) together with the Oslo Sports Trauma Research Centre and the Santa Monica Orthopaedic and Sports Medicine Research Foundation. FIFA 11+ is a complete warm-up program with running exercises at the beginning and end to activate the cardiovascular system. It features specific preventive exercises focusing on core and leg strength, balance and agility, each of them at three levels of increasing difficulty to provide variation and progression (figure 1.1). It takes about 20 to 25 minutes to complete and requires minimal equipment (a set of cones and balls). FIFA 11+ is encouraged to replace usual warm-ups for a few sessions (two or three) each week in amateur adolescent soccer.

2.13	1 RUNNING STRAIGHT AHEAD	ſ	2 RUNNING HIP OUT		3 RUNNING HIP IN
11	The sense is made up of 6 to 10 parts of portfollower, appendix for more space. Sense years multiple after size to sense to the 1 or part of them. By bapteries of the map to the bar part of cores, do the way tack, you can increase your good appendix to a part of the sense of the sense tack, you can increase your good appendix to a part of the sense of the sense tack.	1 Big	. Value or op saals, integroep as each par of cover to the power losses and refers prove Ng saatsweds . Alternate formwer losses will and right logs at taxonautor cores. 2 ares	·	Hall or pop soch, stopping at such par of over to till your love and reda high investig. Alternate between left and optifying at account is some 2 to
N	CIRCLING PARTNER CIRCLING HARTNER And		SOURCE CONTACT SHOULDER CONTACT And the set of the set		6 RUNNING QUICK FORWARDS & BACKWAF An and a service in the service of service in the service of
	STRENGTH · PLYOMETRICS	· BALANC	E · 10 MINUTES		
	EVER. 1 The BENCH Subscription of the second		2 THE BENCH 2 THE BENCH 3 THE BENCH 4 THE ADDRESS		The BENCH THE THE THE THE THE THE THE THE TH
	B SUPEWAYS BENCH STATE The second state of the second state of th		SIDEWAYS BENCH RAISE & LOWER HIP Provide State of the second state		B SUBJECT STATE STATE OF STATE ST
4	Adm/STRINGS BCDINER Decime and the set of the set of the set of the set of the set Decime and the set of the	4	HAMASTRINGS INTERCOMPACTIONS CONTRACTOR AND	<u>s</u> 6	HAMSTRINGS DOVARCE HAMMSTRINGS Automatical and a set of the set of
	10 SINGLE-LEG STANCE HOLD THE BALL The present of the stand of the standard	*	SINGLE-LEG STANCE THROWING BALL WITH PARTNER Device and the second 2-3 are the two parts, with a site of any carding the second part of the second and the part parts have been of the lower part of the second and the second parts have been of the two parts of the second and the second parts have been of the second parts of the second parts have been of the second transport of the second parts of the seco		SINGLE-LEG STANCE TEST YOUR PARTNER Service particles. Sind on one log consists your proton und a min't lo work. Service which pro take grave balance, with direct and the work.
	SQUATS WITH TOP RAISE Notes and the first fragments and the first start of the second		SQUATS WALKING LUNGES Define another such that has well start. Has pure funds are an interface another such that has well start. Has pure funds are and been applied and that a start and the such as an an interface and the such as an and the such as an an an an interface and the such as an and the such as an an an an interface and the such as an and the such as an an an an interface and the such as an an an an an an an an an interface and the such as an an an an an an an an an interface and the such as an an an an an an an an an an an an a		SQUATS ONE-LEG SQUATS Description of the set
	22 JUMPING VERTICAL JUMPS Comparison for the purchash parts from purchash parts for the purchash parts from purchash parts and purchash parts for the purchash parts and purchash parts	*	ULMPING LATERAL JUMPS Description of the set	*	UMMPING DOX JUMPS
	RUNNING EXERCISES · 2 MI	NUTES			
* *	RUINING ACROSS THE PITCH Across the pitch, with a 15-80% material pace. 2 Mar.	Â	10 RUNNING BOUNDING The valid high backing genes with a kigh breach, backag genes (or the kigh of the valid high backing genes (or the kigh breach), backag genes (or the kigh of the kigh and the well-by genes (or the kigh of the back of the kight of the back match, backag backage have and well and the or the back of the back of the back match, backage backage have and well and the or the back of the back of the back match, backage backage have and well and the back of the back of the back of the back match backage backage have and the back of the back		15 RUNNING PLANT & CUT Ing 45 ray, the starts on the calculate and calculate performance of the many plant A calculated in part two balant membra fragment of the many plant A calculated in part two balant membra fragment of the many plant A calculated in part of the start balant of the many plant A calculated in part of the start balant of the many plant and the start plant balant.

Figure 1.1. FIFA 11+ program.

Harmoknee

The Harmoknee injury prevention program, designed by Kiani et al. [36], aims to reduce knee injury incidence rates in youth soccer players. It is focused on improving the main modifiable risk factors for knee injuries but mainly those related to anterior cruciate ligament (ACL) tears and ruptures (e.g., global dynamic stability of the lower limb, knee control during jump and fall actions, eccentric strength, joints range of motion). It is a multi-component program, consisting of five well-defined sections or blocks of exercises: warm up, muscle activation, balance, strength and core stability. These blocks of exercises were designed according to the existing scientific evidence regarding non-contact knee injury risk factors and were also supported by several experienced soccer practitioners (table 1.2).

Exercise	Duration
Part 1: Warm up	10 minutes
1. Jogging	4 minutes
2. Backward jogging on the toes	1 minute
3. High-knee skipping	30 s
4. Defensive pressure technique	30 s
5. One and one	2 minutes
Part 2: Muscle activation	2 minutes
6. Calf	
7. Quadriceps	
8. Hamstrings	4 11 / 11
9. Hip flexor muscles	4 s each leg/side
10. Groin muscles	
11. Hip and lower back muscles	
Part 3: Balance	2 minutes
12. Forward and backward double leg jumps	
13. Lateral single leg jumps	20
14. Forward and backward single leg jumps	30 s
15. Double leg jump with or without ball	
Part 4: Strength	4 minutes (1 min each exercise)

Table 1.2. Harmoknee warm-up program*.

Effects of injury prevention programs on physical performance and neuromuscular control

16. Walking	lunges in place	15 repetitions each leg
17. Hamstrir	ıg curl	12 repetitions
18. Single-kn	ee squat with toe raises	12 repetitions
Part 5: Core stab	llity	4 minutes (1 min each exercise)
19. Sit-ups		2 sets x 12 repetitions
20. Plank on	elbows and toes	2 sets x 20 s
21. Bridging		2 sets x 12 repetitions

*: for more details see http://www.harmoknee.com

FIFA 11+ Kids

As with the FIFA 11+ program, the Kids' version was designed by a panel of international experts from the FIFA Medical Assessment and Research Centre (F-MARC), the Department of Sport, Exercise and Health (DSBG) in Basel (Switzerland), the University of Amsterdam (Netherlands), the University of Cincinnati (USA), the Charles University and Na Bulovce Hospital (Czech Republic), the University of Applied Science in Sion (Switzerland) and the Institute of Sport and Preventive Medicine of Saarland University.

The FIFA 11+ for Kids has been specially developed for children's soccer, and the program is aimed at 7 to 13-year-olds. There is a focus on:

- Spatial orientation, anticipation, and attention particularly while dual-tasking (to avoid unintended contact with other players or objects).
- Body stability and movement coordination (more general than specific).
- Learning appropriate fall techniques (to minimize the consequences of unavoidable falls).

Accordingly, this warm-up program may contribute to achieving the aims of the earlier stages (1. Active Start, 2. FUNdamentals, and 3. Learn to Train) of the long-term athletic development conceptual model for late specialization sports [42–44].

The program consists of seven exercises that should be performed in the specified order at the beginning of every training session (figure 1.2). Every exercise has five levels that become progressively harder (levels 1 to 5). It should take around 15 minutes to complete the program.



Figure 1.2. FIFA 11+ Kids program.

1.3.1 Evidence of injury prevention for FIFA 11+, Harmoknee and FIFA 11+ Kids programs in youth soccer

In the last decade, a substantive number of prospective studies have been conducted to explore the injury-preventing effect of FIFA 11+ in youth soccer (table 1.3). Epidemiological data from these individual studies has been also pooled into different meta-analyses with the aim of increasing statistical power and assess with a higher level of evidence whether the FIFA 11+ program when systematically performed during training is associated with injury reduction in adolescent soccer players [45]. In this sense, the FIFA 11+ program has demonstrated a substantial injury-preventing effect by reducing soccer injuries in recreational/subelite soccer by 39% [45]. Furthermore, this program has been able to reduce the top four most prevalent soccer injuries: hamstring, hip/groin, knee and ankle injuries by 60%, 41%, 48% and 32%, respectively. However, it should be highlighted that in order to achieve the desired injury-preventing effect, the FIFA 11+ must be performed at least twice a week during training sessions and/or match play for 16-24 weeks.

Only Kiani et al. [36] have assessed the efficacy of the Harmoknee program to reduce the number of knee injuries in a large cohort (n = 1506) of adolescent female soccer players (from 13 to 19 years old). In particular, this study documented a significant reduction in non-contact knee injury rates for the intervention (i.e., Harmoknee) group (0.01 injuries per 1000 hours of exposure) in comparison with the control group (0.15 injuries per 1000 hours of exposure), which performed the usual warm-up.

Despite the fact that both the FIFA 11+ and FIFA 11+ Kids have been designed by the same international group of experts, the efficacy of the latter program to reduce injury incidence rates in children soccer players has not been explored with the equal vigor that the FIFA 11+ (table 1.3). In this sense, the evidence available in this regard suggests that the implementation of the FIFA 11+ Kids as pre-exercise warm-up in training sessions and match play may result in a notable reduction of injury risk [42,43] and its associated health care costs [46] in children younger than 14 years. In particular, a major study of over 4,000 children in four countries showed that players who used the FIFA 11+ Kids program suffered 38% fewer injuries than children in the group that used a normal warm-up program, and where serious injuries were concerned (i.e., those involving a period of over 28 days without playing) the figure even exceeded 50% [42].

Therefore, the current scientific evidence available allows to state, but with different degree of certainty, that the FIFA 11+, Harmoknee and FIFA 11+ Kids, when performed systematically during everyday training sessions (2-3 times per week for at least 16 weeks) as standard warm-up, may present powerful injury-preventing effects in youth soccer players. Despite these injury-preventing effects, the implementations of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs in applied youth soccer settings remains a challenging task.

Grass-root coaches have been identified as the key instigators in performing injury prevention programs with their players. What are the performance benefits of such exercises? is one of the most common questions by soccer coaches, when exposed to an injury prevention program. Therefore, if these three injury prevention programs also demonstrated direct performance benefits it would help motivate grass-root coaches and players to implement them (as alternatives to traditional warm-ups) in their training sessions and match play.

Reference		Weekly		m / 1		Caralasiana	
Country	Participants	frequency		l otal exposure	N [®] of injuries	Conclusions	
FIFA 11+							
						In the IG, there was a significantly lower risk of	
	A total of 1892 female players (12-17					injury in general (RR = 0.68 [95%; CI = 0.48-0.98]),	
Soligard et al. [35]	years) belonging to 93 teams	Turing	26 yyaalka	- IG: 49899 h	- IG: 161	lower risk of overuse / fatigue injuries (RR = 0.47	
Norway	- IG: 41 teams and 837 players	Iwice	36 weeks	- CG: 45428 h	- CG: 215	[95%CI = 0.26-0.85]) and lower risk of serious	
	- CG: 52 teams and 1055 players					injury (RR = 0.55 [95% CI = 0.36-0.83]) compared	
						to the CG.	
	A total of 2020 female players (13-17					There was no difference in the overall injury rate	
Steffen et al. [47]	years belonging to 109 teams		32 weeks	- IG: 66423 h	- IG: 242	between the IG (3.6 injuries/1000h [95%CI: 3.2-	
Canada	- IG: 58 teams and 1073 players	1		- CG: 65725 h	- CG: 241	4.1]) and CG (3.7 injuries/1000h [95%Cl = 3.2-4.1]	
	- CG: 51 teams and 947 players					(RR = 1.00 [95%CI = 0.8-1.2]) or in the incidence	
						of any type of injury.	
	A total of 121 male players (13-17					FIFA 11+, unlike CG, showed a significant	
Longo et al. [48]	years) belonging to 11 teams			- IG: 23640 h	- IG: 14	reduction in the overall (0.9 vs 2.2 injuries/1000h)	
Italy	- IG: 7 teams and 80 players	3-4	36 weeks	- CG: 12648 h	- CG: 17	and lower extremity (0.7 vs 1.4 injuries/1000h)	
1000	- CC: 4 teams and 41 players				0011	injury incidence rates. Severe injuries were also	
						reduced (0 vs 0.5 injury/1000h).	
Steffen et al. [49]	A total of 226 female players (13-18	2_3	16 wooks	- IG1: 2230,4 h	- IG1:16	No significant inter-group differences in injury	
Canada	Canada years) belonging to 29 teams	-3 16 weeks	- IG2: 2698,8 h	- IG2: 14	incidence rates were reported. However, a sub-		

Table 1.3. Summary of studies investigating the injury-preventing effects of FIFA 11+, FIFA 11+ Kids and Harmoknee programs.

	 IG1 (Regular): 10 teams and 68 players IG2 (Comprehensive): 10 teams and 78 players CG: 9 teams and 80 players 			- CG: 2496 h	- CG: 16	analysis revealed that compared to players with low adherence, players with high adherence to FIFA 11+ had a 57% lower risk injury (RR = 0.43 [95% CI = 0.19-1.00]).
Owoeye et al. [50] Nigeria	A total of 416 male players (14-19 years) belonging to 20 teams - IG: 10 teams and 204 players - CG: 10 teams and 212 players	Twice	24 weeks	- IG: 51017 h - CG: 61045 h	- IG: 36 - CG: 94	The FIFA 11+ program significantly reduced the overall injury rate in the IG by 41% (RR = 0.59 [95% CI: 0.40-0.86; p = 0.006]) and injuries in lower limbs 48% (RR = 0.52 [95% CI: 0.34-0.82; p = 0.004]).
Rahlf & Zech et al. [51] Germany	A total of 185 players (14-17 years) belonging to 13 teams - IG1 (10 min): 6 teams and 77 players - IG2 (20 min): 7 teams and 108 players	Twice	40 weeks	- IG2: 12499 h - IG1: 8633 h	- IG2: 90 - IG1: 55	No significant differences between groups were found. Injury incidence in the group performing the program 10 min was 6.37/1000h compared to 7.20/1000h in the group performing the program 20 min.
			FIFA 11+	Kids		
Rössler et al. [42] Switzerland Czech Republic Germany Holland	A total of 3895 male and female players (9-12 years) belonging to 243 teams - IG: 128 teams and 2066 players - CG: 115 teams and 1829 players	Twice	36 weeks	- IG: 140716 h - CG: 152033 h	- IG: 139 - CG: 235	Compared to CG, IG reported 48% reduction in overall injury rate, 74% reduction in severe injury rate and 55% reduction in lower extremity injury rate (RR = 0.52 [95%CI = 0.32-0.86]).

Rössler et al. [46] Switzerland Czech Republic Germany Holland	A total of 1002 male and female players (7-12 years) belonging to 62 teams - IG: 37 teams and 2066 players - CG: 25 teams and 1829 players	Twice	44 weeks	- IG: 43777 h - CG: 32596 h	- IG: 42 - CG: 57	This exercise-based injury prevention program was efficacious in reducing the number of soccer injuries by 50% (RR = 0.50 [95%CI = $0.29-0.86$]).
Beaudouin et al. [43] Switzerland Czech Republic Germany Holland	A total of 3895 male and female players (7-13 years) belonging to - teams - IG: - teams and 2066players - CG: - teams and 1829 players	Twice	36 weeks	- IG: 140659 h - CG: 152089 h	- IG: 21 - CG: 50	There was a reduction of severe overall (RR = 0.42 [95% CI = 0.24 to 0.72]), match (RR = 0.41 [95%CI = 0.17 to 0.95]) and training injuries (RR = 0.42 [95% CI = 0.21 to 0.86]) in IG.
Zarei et al. [52] Iran	A total of 962 male players (7-14 years) belonging to 32 teams - IG: 15 teams and 443 players - CG: 17 teams and 519 players	Twice	32 weeks Harmok	- IG: 31934 h - CG: 32113 h	- IG: 30 - CG: 60	Injuries were reduced by half (RR = 0.50 [95%CI = 0.32- 0.78]).
Kiani et al. [36] Sweden	A total of 1506 female players (13-19 years) belonging to 97 teams - CG: 49 teams and 729 players	1-2	36 weeks	- IG: 66981 h - CG: 66505 h	- IG: 3 - CG: 13	Harmoknee presented a high reduction in the knee injury incidence (0.04 vs 0.2 injury/1000h) and specifically, on non-contact injuries in this joint (0.01 vs 0.15 injury/1000h) (RR = 0.06
	- IG: 48 teams and 777 players					[95%CI = 0.01-0.46]).

CG: control group; IG: intervention group; RR: relative risk; CI: confidence interval

1.4 EFFECTS OF FIFA 11+, HARMOKNEE AND FIFA 11+ KIDS PROGRAMS ON PHYSICAL PERFORMANCE AND NEUROMUSCULAR RISK FACTORS

Meta-analytical evidence shows that exercise-based injury prevention programs (including FIFA 11+, FIFA 11+ Kids and Harmoknee) are efficacious in reducing injuries in young athletes when they are systematically performed [45,53–56]. However, often it is difficult to stimulate club managers and coaches to perform such training programs just for the prevention of injuries [57]. As stated before, one important argument to convince manager, coaches, and players for the implementation of these exercise-based injury prevention programs might be the fact that they are able to simultaneously elicit positive (acute and chronic) effect on athletes' performance [32]. In addition, the knowledge of the effects elicited by these exercise-based injury prevention programs on physical and neuromuscular performance measures can help in identifying the potential mechanisms behind the documented reduction in injury incidence.

Acute and chronic physical performance and neuromuscular control adaptations resulting from the application of the FIFA 11+, Harmoknee and FIFA 11+ Kids in organized child and adolescent soccer are presented as follow.

1.4.1 Post-exercise (acute) effects of the FIFA 11+, FIFA 11+ Kids and Harmoknee programs on measures of physical and neuromuscular performance youth soccer

It is widely accepted that a "good" or "appropriate" warm-up program should be able to improve performance (via post activation potentiation, decrease in stiffness, rise in core temperature and resting oxygen consumption [58]) but should not be too demanding to cause detrimental effects due to fatigue related factors [59]. Therefore, the FIFA 11+, FIFA 11+ Kids and Harmoknee must demonstrate that they are able to elicit physiological acute changes (post-exercise effects) to positively affect the major physical and neuromuscular performance measures (i.e., sprinting, jumping, range of motion, etc.) before being considered as appropriate warm-up program to be performed prior to formal training and competition.

Furthermore, if the use of the FIFA 11+, FIFA 11+ Kids and/or Harmoknee is to be promoted at the expense of dynamic warm-up programs currently performed by most amateur young soccer teams, the magnitude of their hypothetical positive effects on physical and neuromuscular performance measures should be at least similar to those reported by the latter mentioned warm-up programs [28,52,60–64]. Otherwise, the FIFA 11+, FIFA 11+ Kids and Harmoknee should be implemented in everyday soccer sessions as training components (i.e., placed in the main part of the training sessions), based on their demonstrated superior positive effects on injury incidence rates in comparison with traditional practices, rather than being used as a pre-exercise warm-up routine.

However, analyzing the body of literature regarding the acute effects of the FIFA 11+, FIFA 11+ Kids and Harmoknee on sports performance, a smaller evidence base supporting their use as "good warm-up programs" appears to be available in comparison with the number of studies that have documented their efficacy as long-term interventions to reduce the injury incidence (table 1.3). To the best of the candidate knowledge, only Bizzini et al. [59], Silva et al. [61], Chen et al. [65], Akbari et al. [62], Pardos-Mainer et al. [66], Parsons et al. [67], and Trajkovic et al. [64] have examined the pre-exercise effects of the FIFA 11+ on various physical and neuromuscular performance variables in young (males and females) soccer players, showing improvements in 20 m sprint time, jump height, agility, balance (static and dynamic) and maximal strength comparable with those obtained with other dynamic warm-up routines reported in the literature. In addition, Bizzini et al. [59] also reported that the FIFA 11+ induced similar improvement in resting oxygen uptake, core temperature and lactate as those obtained with other warm-up routines.

However, no studies (to the candidate' knowledge) have analyzed the acute effects of the Harmoknee on physical and neuromuscular performance measures in young soccer players.

For its part, only Rössler et al. [28] have examined the pre-exercise effects of the FIFA 11+ kids on various physical performance variables in children soccer players, showing possibly beneficial effects in static and dynamic balance, jumping performance, and slalom dribbling.

Therefore, there is a clear need of studies aimed at exploring the acute (postexercise) effects of these three programs on physical and neuromuscular performance measures in order to elucidate whether (or not) the FIFA 11+, FIFA 11+ Kids and/or Harmoknee may be considered as appropriate warm-up routines.

1.4.2 Training (chronic) effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs on measures of physical and neuromuscular performance youth soccer

Some studies, but not many, have explored the training effects of both FIFA 11+ (mainly) and Harmoknee on some measures of neuromuscular performance (lower extremity balance [60] and core stability [67], knee proprioception [60], knee strength [60]) and sprinting and jumping abilities [61–64,67], which form part of the athletic physical performance spectrum, in adult soccer players (table 1.4). However, only six studies (to the best of the candidate's knowledge) have been published analyzing the training effects of these two injury prevention programs on measures of physical and neuromuscular performance in adolescent soccer players despite being the target population of both programs. For example, Zarei et al. [68] examined the long-term effects of the FIFA 11+ program on selected physical performance measures in adolescent male soccer players (n = 82). After 30 weeks of using the FIFA 11+ as a warm-up program, improvements in jump height, agility and sprint time were observed compared to a normal warm-up routine [68].

For its part, only one study has examined the pre-exercise effects of the FIFA 11+ Kids on various physical performance measures in children soccer players, showing possibly beneficial effects in static and dynamic balance, jumping performance and slalom dribbling [28].

Therefore, studies on the physical and neuromuscular performance effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids in children and adolescent soccer players are clearly warranted. This knowledge would help convince club managers and coaches of the need to implement the FIFA 11+, Harmoknee and FIFA 11+ Kids in their daily training sessions with the aims of reducing the number and severity of injury and improving performance in youth soccer players.

Reference	—					Outcome n	neasures		Key findings
Country	Participants	Design	Interventions	Sprint	Jump	Balance	Agility	Others	
]	FIFA 11+				
Daneshjoo et al. [60] Iran	36 male young soccer players IG = 12 (19.2 ± 0.9 y) CG = 12 (19.7 ± 1.6 y)	Cluster-RCT	FIFA 11+ 8 weeks 3 days per week	-	-	Y-Balance Static Balance open and closed eyes	-	Proprioception Isokinetic strength (knee)	Dynamic balance, proprioception and isokinetic strength improved in FIFA 11+ group compared with CG.
Silva et al. [61] Brazil	17 male young soccer players IG = 9 (18.3 ± 1.6 y) CG = 8 (18.3 ± 1.6 y)	Cluster-RCT	FIFA 11+ 9 weeks 3 days per week	Sil	SJ CMJ	iol	ec	a	IG showed significant improvements in both types of jumps (SJ and CMJ).
Akbari et al. [62] Iran	24 male elite-young soccer players IG = 12 (16.79 ± 1.18 y) CG = 12 (16.79 ± 1.18 y)	Cluster-RCT	FIFA 11+ 8 weeks 3 days per week	-	vj	Maguel	Herna	-	IG improves performance in the vertical jump test, but the improvement was not maintained after one month of halting training.
Zarei et al. [63] Iran	82 male adolescent soccer players $IG = 34 (15.0 \pm 0.7 y)$ $CG = 32 (15.2 \pm 0.6 y)$	Cluster-RCT	FIFA 11+ 30 weeks At least 2 days per week	20 and 40 yards	VJ CMJ	-	Illinois agility Dribbling sprint	Yo-Yo IR Sit and reach	IG showed superior results compared to CG in sprint time (20 yards), jump height (VJ and CMJ) and agility (Illinois agility).
Parsons et al. [67]	43 female young soccer players	Cluster-RCT	FIFA 11+ 36 weeks		LESS	Y-Balance	T-test	Static plank	IG increased their mean static plank hold time compared with

Table 1.4. Training (chronic) effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs on measures of physical and neuromuscular performance in youth soccer players.

Canada	IG = 25 (11.1 ± 1.6 y) CG = 18 (10.8 ± 2.2 y)		1-2 days per week		VJ				CG. Y-Balance scores worsened from pre- to postseason in both groups.
Trajkovic et al. [64] Serbia	36 male young soccer players IG = 19 (11.2 ± 0.8 y) CG = 17 (10.9 ± 0.8 y)	Cluster-RCT	FIFA 11+ 4 weeks 3 days per week	20 m	SLJ	-	Illinois agility	RSA Sit and reach	Agility and jump height improved in FIFA 11 group compared with control.
				Н	armoknee				
Daneshjoo et al. [60] Iran	36 male young soccer players IG = 12 (19.2 ± 0.9 y) CG = 12 (19.7 ± 1.6 y)	Cluster-RCT	Harmoknee 8 weeks 3 days per week	Sil	ol	Y-Balance Static Balance open and closed eyes	ec	Proprioception Isokinetic strength (knee)	Dynamic balance, proprioception and isokinetic strength improved in Harmoknee group compared with CG.
				FIF	A 11+ Kid	s			
Zarei et al. [52] Iran	31 male young soccer players IG = 16 (11.3 ± 0.9 y) CG = 15 (11.7 ± 0.7 y)	Cluster-RCT	FIFA 11+ Kids 10 weeks 2 days per week	-	-	-	-	Isokinetic strength (hip, knee and ankle)	The IG was beneficial compared to the CG regarding isokinetic strength of the hip adductors, knee flexors, as well as ankle evertors and invertors.
Zarei et al. [63] Iran	56 male adolescent soccer players IG = 19 (11.9 ± 1.9 y) CG = 23 (12.1 ± 1.1 y)	Cluster-RCT	FIFA 11+ Kids 10 weeks At least 2 days per week	20 and 40 yards	SLJ Hop	Y-Balance	Illinois agility Dribbling sprint	Plank Side plank	IG gained significantly better results in balance, jump height and sprint time compared to CG.

	122 children soccer		FIFA 11+ Kids		DI				Beneficial improvements in favor
Kossler et al.	players		10 weeks	20	DJ	V Palanas	Slalom V dribble		of FIFA 11+ Kids were observed in
[20] Switzerland	$IG = 56 (10.0 \pm 1.8 \text{ y})$	Cluster-KC1	At least 2 days	At least 2 days	SU I	1-Dalance		wan voney	jump height, balance, agility and
Switzenand	CG = 66 (10.1 ± 1.6 y)		per week		SLJ				wall volley performance.

IG: intervention group; CG: control group; RCT: randomized control trial; DJ: drop jump; CMJ: countermovement jump; SLJ: standing long jump; VJ: vertical jump; LESS: Landing error scoring system; RSA: repeated sprint ability.



1.5 LINES OF ACTION OF THE PRESENT PHD THESIS

The FIFA 11+, Harmoknee and FIFA 11+ Kids, when they are systematically performed as pre-exercise warm-ups (2-3 times a week) in training sessions and match play for at least 3 months, have demonstrated being powerful risk mitigation strategies in amateur youth soccer. However, despite their recognized injury-preventing effects, coaches and players do not usually embrace with enthusiasm these standardized warm-up programs as their adherence rates have been consistently reported as low in youth soccer. An argument that might help to persuade coaches and soccer players to incorporate these injury prevention programs in daily training sessions and matches is that they could be concurrently used to benefit sport performance. Furthermore, the knowledge of the effects elicited by these exercise-based injury prevention programs on physical and neuromuscular performance measures can help in identifying the potential mechanisms behind the documented reduction in injury incidence. However, very few studies have analyzed the effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs on major physical performance measures and neuromuscular risk factors. Therefore, the aim of the present doctoral thesis focuses on analysing the acute (i.e., post-exercise) (study 1) and training (i.e., chronic) (studies 2 and 3) effects of the FIFA 11+, Harmoknee and FIFA 11+ Kids programs on measures of physical and neuromuscular performance in youth soccer players.



CHAPTER 2

Research Objectives and Hypothesis



2

RESEARCH OBJECTIVES AND HYPOTHESES

2.1 GENERAL OBJECTIVE

Based on the limitations of the literature presented in the previous chapter, the general objective of the current doctoral thesis was *to analyze the acute and chronic effects of three neuromuscular training programs on parameters of physical performance and neuromuscular control in male young soccer players.*

In order to achieve this objective, three randomized controlled (experimental) studies were carried out. The first study described the acute effects of the FIFA 11+, Harmoknee and dynamic warm-up routines on several physical performance measures in amateur soccer players. The second study analyzed the training effects of the FIFA 11+ and Harmoknee on several parameters of physical performance in youth amateur soccer players. The last study examined the training effects of the FIFA 11+ Kids on several parameters of physical performance in male youth soccer players.

The titles of the three studies that comprised the current doctoral thesis are the following:

- **Study 1.** Acute effects of three neuromuscular warm-up strategies on several physical performance measures in football players.
- **Study 2.** Training effects of the FIFA 11+ and Harmoknee on several neuromuscular parameters of physical performance measures.
- **Study 3.** Training effects of the FIFA 11+ Kids on physical performance in youth football players: a randomized control trial.

2.2 SPECIFIC OBJECTIVES

The specific objectives have been structured depending on the three studies of this doctoral thesis:

Study 1. Acute effects of three neuromuscular warm-up strategies on several physical performance measures in football players.

1. To analyze the acute (post-exercise) effects of the FIFA 11+, Harmoknee and dynamic warm-up routines on several physical performance measures in amateur soccer players.

Study 2. Training effects of the FIFA 11+ and Harmoknee on several neuromuscular parameters of physical performance measures.

2. To analyze the training effects of the FIFA 11+ and Harmoknee on several parameters of physical performance measures in youth amateur soccer players.

Study 3. Training effects of the FIFA 11+ Kids on physical performance in youth football players: a randomized control trial.

3. To analyze the training effects of the FIFA 11+ Kids on several parameters of physical performance in male youth soccer players.

2.3 **Research hypotheses**

The following hypotheses were established for each of the three studies included in the current doctoral thesis:

Study 1. Acute effects of three neuromuscular warm-up strategies on several physical performance measures in football players.

1. The candidate hypothesizes that there would not be significant differences in the acute effects elicited by the three soccer-related warm-up protocols selected on the physical performance measures analyzed based on the fact that the duration (approximately 25 min) and type of exercises (neuromuscular exercises) in each routine are similar.

Study 2. Training effects of the FIFA 11+ and Harmoknee on several neuromuscular parameters of physical performance measures.

2. The candidate hypothesizes that FIFA 11+ and Harmoknee would show beneficial and superior effects on physical performance (particularly in balance, sprint and jumping measures) in comparison to the traditional practices as they include specific and novel exercises designated to improve these parameters (e.g., Nordic hamstring exercise, single-leg stance tasks, multidirectional [vertical, forward, backward, lateral] single leg jumps, running drills [alternate leg bounds, sub-maximal speed lineal running]).

Study 3. Training effects of the FIFA 11+ Kids on physical performance in youth football players: a randomized control trial.

3. It was hypothesized that FIFA 11+ Kids would show beneficial and superior effects on physical performance (particularly in balance and agility measures) in comparison to the traditional practices as they include specific and novel exercises designed to improve physical competency.





CHAPTER 3

19

Study 1



3

STUDY 1

ACUTE EFFECTS OF THREE NEUROMUSCULAR WARM-UP STRATEGIES ON SEVERAL PHYSICAL PERFORMANCE MEASURES IN FOOTBALL PLAYERS

by

Ayala F, Calderón-López A, Delgado-Gosálbez JC, Parra-Sánchez S, Pomares-Noguera C, Hernández-Sánchez S, López-Valenciano A, & De Ste Croix M.

3.1 Abstract

No studies have analyzed the acute effects of the FIFA 11+ and Harmoknee warmup programs on major physical performance measures. The aim of this study was to analyze the acute (post-exercise) effects of the FIFA 11+, Harmoknee and dynamic warmup routines on several physical performance measures in amateur soccer players. A randomized, crossover and counterbalanced study design was used to address the purpose of this study. A total of sixteen amateur soccer players completed the following protocols in a randomized order on separate days: a) FIFA 11+; b) Harmoknee; and c) dynamic warmup (DWU). In each experimental session, 19 physical performance measures (joint range of motion, hamstring to quadriceps [H/Q] strength ratios, dynamic postural control, 10 and 20 m sprint times, jump height and reactive strength index) were assessed. Measures were compared via a magnitude-based inference analysis. The results of this study showed no main effects between paired comparisons (FIFA 11+ vs. DWU, Harmoknee vs. DWU and Harmoknee vs. FIFA 11+) for joint range of motions, dynamic postural control, H/Q ratios, jumping height and reactive strength index measures. However, significant main effects (likely effects with a probability of >75–99%) were found for 10 (1.7%) and 20 (2.4%) m sprint times, demonstrating that both the FIFA 11+ and Harmoknee resulted in slower sprint times in comparison with the DWU. Therefore, neither the FIFA 11+ nor the Harmoknee routines appear to be preferable to dynamic warm-up routines currently performed by most soccer players prior to training sessions and matches.

Keywords: soccer, acute effects, warm-up, neuromuscular control, injury prevention.

3.2 INTRODUCTION

The FIFA 11+ [69] and Harmoknee [36] are two warm-up programs designed to prevent and reduce the number and severity of soccer-related injuries, particularly in amateur players. Both programs include running exercises and specific dynamic movements focusing on the major neuromuscular risk factors for lower extremity injuries (i.e., leg strength, balance, dynamic postural control, agility, knee control during cutting and landing, joint range of motion), based on scientific evidence and best practice [36,69]. In addition to the running and neuromuscular exercises, the FIFA 11+ and Harmoknee also include sprinting, multidirectional speed and plyometric exercises.

Recent cluster randomized controlled trials have demonstrated that the FIFA 11+ is effective in reducing lower extremity injury rates (mainly knee injuries) in teams practicing this warm-up at least twice a week for longer than three consecutive months [35,49,50,70–72]. Although with less rigor than the FIFA 11+, the effectiveness of the Harmoknee to reduce the incidence of lower extremity injuries (mainly knee injuries) has also been documented [36]. The potential training (chronic) effects behind the reported reduction in injury incidence produced by the FIFA 11+ and Harmoknee appear to be related to improvement in neuromuscular control of the trunk and lower extremities [60,73,74]. These positive results in injury incidence rates in combination with the fact that no additional or specific equipment (for example balance boards) is required have led some research groups and institutions to develop countrywide campaigns to implement the FIFA 11+ and Harmoknee in the everyday soccer training routines (especially at amateur levels) [57,75].

It is widely accepted that a "good" or "appropriate" warm-up program should be able to improve performance (via post activation potentiation, decrease in stiffness, rise in core temperature and resting oxygen consumption [58]) but should not be too demanding to cause detrimental effects due to fatigue related factors [59]. Therefore, both, the FIFA 11+ and Harmoknee must demonstrate that they are able to elicit physiological acute changes (post- exercise effects) to positively affect the major physical performance measures (i.e., sprinting, jumping, range of motion, etc.) before being considered as appropriate warm-up programs to be performed prior to formal training and competition. Furthermore, if the use of the FIFA 11+ and/or Harmoknee is to be promoted at the expense of dynamic warm-up programs currently performed by most amateur soccer teams, the magnitude of their hypothetical positive effects on physical performance measures should be at least similar to those reported by the latter mentioned warm-up programs [76–84]. Otherwise, the FIFA 11+ and Harmoknee should be implemented in everyday soccer sessions as training components (i.e., placed in the main part of the training sessions), based on their demonstrated superior positive effects on injury incidence rates in comparison with traditional practices, rather than being used as a pre-exercise warm-up routine.

However, analyzing the body of literature regarding the acute effects of the FIFA 11+ and Harmoknee on sports performance, a smaller evidence base supporting their use as "good warm-up programs" appears to be available in comparison with the number of studies that have documented their efficacy as long-term interventions to reduce the injury incidence [35,49,50,57,70–72,75]. To the best of our knowledge, only Bizzini et al. [59] have examined the pre-exercise effects of the FIFA 11+ on various sports-performance variables, showing improvements in 20 m sprint time, jump height and agility comparable with those obtained with other dynamic warm-up routines reported in the literature. In addition, Bizzini et al. [59] reported that the FIFA 11+ induced similar improvement in resting oxygen uptake, core temperature and lactate as those obtained with other warm-up routines. However, no studies (to the authors' knowledge) have analyzed the acute effects of the Harmoknee on physical performance measures.

Therefore, the main purpose of this study was to analyze the acute (post-exercise) effects of the FIFA 11+, Harmoknee and dynamic warm-up routines on several physical performance measures in amateur soccer players. We hypothesize that there would not be significant differences in the acute effects elicited by the three soccer-related warm-up protocols selected on the physical performance measures analyzed based on the fact that the duration (approximately 25 min) and type of exercises (neuromuscular exercises) in each routine are similar.

3.3 MATERIALS AND METHODS

3.3.1 Design

A randomized, post-test only crossover and counterbalanced study design, in which participants performed all experimental conditions, was used to address the aims of this study. The use of a pre- and post-test crossover design, in which participants performed a pre- and post- warm up physical performance assessment was not adopted because in a pilot study (n = 4 males and 2 females) the participants consistently reported that the preand post- warm up assessment procedure used was too long and subsequently they felt less able or fatigued to undertake the post-warm up assessment and hence, bias the results.

The independent variables were the three different intervention routines (FIFA 11+, Harmoknee and dynamic warm-up). The dependent variables included 19 physical performance measures (range of motion, dynamic postural control, conventional and functional hamstring-to-quadriceps strength ratios, 10 and 20 m sprint times, jumping height and reactive strength index).

Participants visited our laboratory on four occasions, with a week's rest interval between sessions. Each testing session was carried out 48–72 hours after finishing the previous competitive match (i.e., Tuesday or Wednesday) so that the players could have enough time for recovery. Furthermore, players did not carry out any training session throughout this rest-interval. To minimize circadian variation and other similar effects on physical performance, each participant carried out all experimental sessions at the same time of day (in the late afternoon or early evening, depending on the participants availability) and under the same environmental conditions (room temperature at 25°C).

The first visit was a practice/habituation session to the different testing procedures and warm-up exercises, and the following three visits were the experimental sessions. During each experimental session, participants began by completing one of the three interventions: the FIFA 11+, the Harmoknee or the dynamic warm-up. The order of interventions was randomized per person using a computer-based software program (<u>www.randomizer.org</u>) to avoid carry-over effects. The assessment of the physical performance measures was carried out 2–3 minutes (post-test) after the entire warm-up program was completed. This time elapse between the end of the warm-up and the start of the assessment of the physical performance measures was selected because: a) it reflects the typical period of time existing between the end of the warm-up and the start of a match in the amateur leagues; and b) to be consistent with similar previous studies [59,82].

The order of the tests was consistent throughout the experimental sessions and was established with the intention of minimizing any possible negative influence among variables (figure 3.1). The mean duration of each experimental session was 65.3 ± 8.6 min.

A priori, the overall duration of the post-warm up physical performance assessment (45 min approximately) carried out within each experimental session, although similar to previous studies [59,80,84-86], could appear to be enough to generate fatigue in the participants to bias the results (at least in the latter testing measures). However, the abovementioned pilot study also showed that the order of the tests, the number of trials performed, as well as the rest intervals within trials and among tests were sufficient in order not to produce meaningful musculoskeletal fatigue in the participants. Furthermore, a "damping phenomenon" (i.e., the physical demands required by the test battery may offset the potential effects of an intervention) was not expected due to the fact that the sportspecific dynamic movement components of the three warm ups included activities with very similar neuromuscular and energetic requirements to those needed to successfully perform in the tests. In addition, Ayala et al. [85] reported that the time course of the effects elicited by a dynamic warm up with similar characteristics (duration, type of exercises, intensity) to those used in the current study on some physical performance measures (i.e., 10 and 20 m sprint, jumping height) lasted more than 30 minutes while participants engaged in a simulated tennis match. Therefore, a decline in the magnitude of the effects elicited by the three interventions on physical performance throughout the length of the post- warm up assessments was not expected. Each experimental session was carried out under the strict supervision of the researchers.



Figure 3.1. Schematic representation of the study design.

3.3.2 Participants

Twenty-two participants, consisting of 12 men and 10 women who were amateur soccer players, took part in the current study. Participants belonged to five different soccer teams that were engaged in the Official Amateur Championships of the Spanish Soccer Federation (fourth [regional] Spanish division). Although all participants reported taking part in regional soccer leagues, none was involved in a systematic and specific strength-training program. Likewise, their training frequency and duration through the study was 3 days per week, 1.5 hours per session. To reduce the interference of uncontrolled variables, all the participants were instructed to maintain their habitual lifestyle and normal dietary intake before and during the study. The participants were told not to exercise on the day before a test and to consume their last (caffeine-free) meal at least 3 hours before the scheduled test time.

Other exclusion criteria were: (1) histories of neuromuscular diseases or musculoskeletal injuries specific to the shoulder, ankle, knee, or hip joints over the past 6 months; (2) missing one testing session during the data collection phase; (3) and presence of self-reported delayed onset muscle soreness at any testing session. Before any participation, experimental procedures and potential risks were explained fully to the participants both verbally and in writing. Written informed consent was obtained from the players. The Institutional Research Ethics Committee approved the study participant information sheet, design and testing protocols prior data collection.

Eight men (age = 19.1 ± 1.3 y; stature = 177.2 ± 6.4 cm; body mass = 71.4 ± 8.8 kg) and eight women (age = 20.1 ± 1.8 y; stature = 164.9 ± 6.9 cm; body mass = 60.1 ± 6.6 kg) classified as amateur soccer players completed this study. Four men and two women were excluded from the study because they missed one or more of the testing sessions.

3.3.3 Interventions

FIFA 11+

The FIFA 11+ consisted of three parts (table 3.1), the first of which involved running exercises (part 1). The second part covered six exercises, all of which comprised three levels of difficulty and were aimed at improving strength, balance, muscle control and core

stability (part 2). The third and final part consisted of advanced running exercises (part 3). All players were able to perform level II of difficulty for each exercise in part 2, which was confirmed during the familiarization session. However, no player was able to perform the level III of difficulty for each exercise. Therefore, the level II of difficulty was chosen for this study.

Exerci	se	Duration				
Part 1	Running exercises	8 minutes				
1.	Straight ahead					
2.	Hip out					
3.	Hip in					
4.	Circling partner	2 sets over 30 m each exercise				
5.	Shoulder contact					
6.	Quick forward & backwards	plateca				
Part 2	Strength—Plyometric—Balance	10 minutes				
7.	The bench: alternate legs	3 sets x 40 s (lifting 2 s each leg in turn)				
8.	Sideways bench: raise and lower hip	3 sets x 20 repetitions each side				
9.	Hamstrings: intermediate	1 set x 7 repetitions				
10.	Single-leg stance: throwing ball with partner	2 set x 30 s each leg				
11.	Squats: walking lunges	2 set x 10 repetitions each leg				
12.	Jumping: lateral jumps	2 set x 15 jumps (30 s approximately)				
Part 3	Running exercises	2 minutes				
13.	Across the pitch	2 sets x 30 m (70–80% maximum pace)				
14.	Bounding	2 sets x 30 m				
15.	Plant and cut	2 sets x 5 repetitions (80–90% maximum pace)				

Table 3.1. FIFA 11+ warm-up program.
Harmoknee

The Harmoknee warm-up program included 5 parts: warm-up, muscle activation, balance, strength and core stability, all of which can be combined and performed in a regular soccer training session (table 1.2). Total program duration was 20 to 25 minutes.

Dynamic warm-up

Ten different cards showing dynamic warm-up routines were given to each participant. Seven of the dynamic warm-up routines were extracted from the studies selected in the meta-analysis carried out by Bizzini et al. [59] in the context of the acute effects of dynamic warm-ups [78–81,84,87,88]. The other three warm-up routines were selected because they reflect the warm-up structure and content that, according to our experience, might be the most widely used in soccer and which contain the following components chronologically ordered: a) some active aerobic activities (including running, light calisthenics); b) dynamic stretching exercises of the major muscle groups; and c) soccer-specific movements incorporating a various range of motion exercises with skill-based drills executed at, or just below game intensity [82,89,90].

Those published warm-up routines that required additional equipment (for example, balance boards, dumb-bells, weight vests or barbells) were not considered for the purpose of this study because they do not represent the reality of most amateur clubs and teams. After that, each participant picked up the three cards that most closely replicated his/her habitual warm-up routine. Three points were given to the most closely related warm-up routine, two points to the second and one point to the last warm-up routine. The dynamic warm-up routine considered the 'standard' must have been picked up and scores with two or three points by at least 70% of the participants. In the case of more than one dynamic warm-up routine meeting the above-mentioned criterion, the one with the highest summation of points was selected as standard. Two dynamic warm-up routines were picked up by at least 70% of the participants [82,89], these being the dynamic warm-up routine designed by Taylor et al. [82], which reached the highest score and hence was considered as the standard soccer-related warm-up routine (table 3.2).

Exercise		Duration		
1.	High knees	3 set over 20 m		
2.	Butt flicks	3 set over 20 m		
3.	Carioca	3 set over 20 m each side		
4.	Dynamic hamstring swings	10 repetitions each leg		
5.	Dynamic groin swings	10 repetitions each leg		
6.	Arm swings: forwards and backwards	10 repetitions each direction		
7.	Faster high knees (shorter stride)	4 sets over 10 m		
8.	Swerving	2 sets over 30 m at 70% of maximum pace		
9.	Side stepping	2 sets over 30 m at 80% of maximum pace		
10.	Spiderman walks	1 set over 20 m		
11.	Sideways low squat walks	1 set x 10 steps each direction		
12.	Upper body rotations	10 repetitions each leg		
13.	Vertical jump	5 repetitions building in intensity		
14.	Run through 2 sets x 20 m at 70% of maximum p			
		2 sets x 20 m at 80% of maximum pace		
	Countermovement jump then 5 m sprint	1 set x 20 m at 90% of maximum pace		
15.		2 sets x 5 m at 90% of maximum pace		
		1 sets x 5 m at 95% of maximum pace		
16.	Sprint for 5 m then countermovement jump	2 sets x 5 m		

Table 3.2. Dynamic warm-up program*.

*: warm up program extracted from Taylor et al. [82]

3.3.4 Physical performance measures

Hip, knee and ankle range of motions

The passive hip flexion (passive straight leg raise test [figure a in supplementary file 3.1]), extension (Thomas test [figure b in supplementary file 3.1]) and abduction (passive hip abduction with knee flexed over the edge of the plinth test [figure c in supplementary file 3.1]); knee flexion (Modified Thomas test [figure d in supplementary file 3.1]) and ankle dorsiflexion (weightbearing lunge with knee extended [figure e in supplementary file 3.1] and flexed [figure f in supplementary file 3.1] tests) range of motions of the dominant and

non-dominant extremities were assessed following the methodology previously described [91]. Participants were instructed to perform, in a randomized order, two maximal trials of each range of motion test for each extremity. The mean score for each test was used in the subsequent analyses. The same researchers performed the ROM testing at all testing sessions.

Conventional and functional hamstring-to-quadriceps strength ratios

The assessment of the H/Qconv and H/QFUNCT strength ratios was carried out following the methodology described by Ayala et al. [92]. Briefly, a Biodex System-4 isokinetic dynamometer (Biodex Corp., Shirley, NY, USA) and its respective manufacture software were used to determine isokinetic concentric and eccentric torques during knee extension and flexion actions. Only the dominant leg was tested, as no meaningful differences between legs have been previously reported for sedentary and recreationally active adults [93,94]. The dynamometer was calibrated according to the manufacturer's instructions immediately before each test session and verified immediately after to ensure that no changes occurred in sensitivity.

Participants were secured supine on the dynamometer with the hip passively flexed at 10° – 20° (supplementary file 3.2). The axis of rotation of the dynamometer lever arm was aligned with the lateral epicondyle of the knee. The force pad was placed approximately 3 cm superior to the medial malleolus, with the foot in a relaxed position. Adjustable strapping across the pelvis, thigh proximal to the knee and foot localized the action of the musculature involved. The range of movement was set from 90° knee flexion (starting position) to 0° (0° was determined as maximal voluntary knee extension for each participant).

Before isokinetic testing, the participants performed a specific isokinetic warm-up consisting of three sub-maximal (self-perceived 50% effort) and two maximal concentric and eccentric knee extension and flexion actions at 120°/s.

The isokinetic examination was separated into two parts. The first part of the examination was the assessment of the knee extensor and knee flexor muscles during concentric/concentric (CON/CON) cycles with extension undertaken first. After a 5 min rest period the eccentric/ eccentric (ECC/ECC) testing cycle was performed.

In both testing methods, three cycles of knee flexions and extensions were performed at three preset constant angular velocities in the following order: 60, 180 and 240°/s for CON/ CON cycles; and 30, 60 and 180°/s for ECC/ECC cycles (slow to fast). The two testing parts (CON/CON and ECC/ECC) were separated by a 5 min rest interval and a rest of 30 s was allowed between action cycles. The number of maximal muscle actions and the rest period durations were chosen to minimize musculoskeletal fatigue, which is unlikely to occur (based on the participants' perceptions reported in the pilot study) with only three muscle actions at three velocities and 30 s rest between muscle actions and velocities and 5 min rest between testing modes (concentric and eccentric). For both concentric and eccentric actions, participants were encouraged to push–pull/resist as hard and as fast as possible and to complete the full range of motion. Participants were instructed to abort the test if they felt any discomfort or pain. During the test, all participants were given visual feedback from the system monitor.

They were also verbally encouraged by the investigator to give their maximal effort, and the instructions were standardized by using key words such as 'resist', 'push' and 'hard and fast as possible'.

The peak torque was extracted for each of the three trials performed at each velocity during the CON/CON and ECC/ECC cycles of reciprocal knee extension and flexion movements through the three testing sessions. For each isokinetic peak torque variable, the average of the three trials at each velocity was used for the subsequent calculation of the conventional and functional hamstring-to-quadriceps strength ratios due to the magnitude of the error component decreasing with increased trials [95].

Thus, the H/QCONV ratios were calculated as the ratio between the torques produced concentrically by hamstrings and quadriceps during the isokinetic tests at 60 (H/QCONV60) and 180°/s (H/QCONV180). H/QFUNC ratios were calculated as the ratio between the torques produced eccentrically by the hamstring and concentrically by the quadriceps muscles at 60 (H/QFUNC60) and 180°/s (H/QFUNC180). Finally, the H/QFUNC proposed by Croisier et al. [96] was also calculated as the ratio between the torques produced eccentrically by the hamstring at 30°/s and concentrically by the quadriceps muscles at 240°/s (H/QFUNC30/240).

Dynamic postural control

Dynamic postural control was evaluated using the Y-Balance test and following the guidelines proposed by Shaffer et al. [97]. Players were allowed a maximum of five trials to obtain three successful trials for each reach direction (anterior [figure a in supplementary file 3.3], posteromedial [figure b in supplementary file 3.3] and posterolateral [figure c in supplementary file 3.3]). Trials were discarded if the player failed to maintain unilateral stance on the platform, failed to maintain reach foot contact with the reach indicator on the target area while the reach indicator is in motion, used the reach indicator for stance support, or failed to return the reach foot to the starting position under control [97]. Specifically, the testing order was completed as dominant anterior, non-dominant anterior, dominant posteromedial, non-dominant posteromedial, dominant posterolateral, and nondominant posterolateral. The average of the three reaches was normalized by dividing by the previously measured leg length to standardize the maximum reach distance ([excursion distance/leg length] x100 = % maximum reach distance) [98]. Leg length was defined as the length measured in centimeters from the anterior superior iliac spine (ASIS) to the most distal portion of the medial tibial malleolus. To obtain a global measure of the balance test, data from each direction were averaged for calculating a composite score [99].

10- and 20-meters sprint time

Owing to their good reproducibility, linear sprint tests ranging from 10 to 20 m are used as general measures of linear acceleration and speed in soccer players. Time during a 20 m sprint in a straight line was measured by means of single beam photocell gates placed 1.0 m above the ground level (Time It; Eleiko Sport, Halmstad, Sweden). Each sprint was initiated from an individually chosen standing position, 50 cm behind the photocell gate, which started a digital timer. Each player performed three maximal 20 m sprints interspersed with 1 min of passive recovery, and the mean of the two fastest times achieved was retained.

Drop vertical jump height

A vertical drop jump (DJ) without arm swing was performed on a force platform (Kistler, Switzerland) according to Onate et al. [100]. Participants stood with feet shoulderwidth apart on a 28 cm high step, 30 cm from the contact platform. They were instructed to lean forward and drop from the step as vertically as possible, in an attempt to standardize landing height. Participants were required to land with one foot on each of the force plates, then immediately perform a maximal vertical jump, finally landing back on the contact platform. Participants were asked to keep their hands on their hips to prevent the influence of arm movements on vertical jump performance. Each participant performed five maximal jumps starting from a standing position, with at least 30 s of recovery between jumps. Participants were asked to jump as high as possible. The mean jump height of the best three jumps was used for statistical analysis.

Reactive strength index

The reactive strength index was calculated using the data from the drop vertical jumps performed by the participants and following this formula [101,102]:

Reactive strength index = Jumping height (mm) / ground contact time (ms)

The mean reactive strength index of the best three jumps was used for statistical analysis.

3.3.5 Statistical analysis

Descriptive statistics (means ± standard deviations) were calculated for all physical performance measures at post warm-up interventions.

Dependent sample t-tests were carried out to assess differences between limbs (dominant versus non-dominant) in dynamic postural control and ROMs.

Magnitude-based inferences of differences between-groups were calculated for each variable using a spreadsheet designed by Hopkins [103]. Each participant's change score between paired sessions (dynamic warm-up vs. FIFA 11+; dynamic warm-up vs. Harmoknee; FIFA 11+ vs. Harmoknee) was expressed as a percentage of baseline score via analysis of log-transformed values, to reduce bias arising from non-uniformity of error. Errors of measurement and individual responses expressed as coefficients of variation were also estimated. In addition, the analysis determines the probability that the true effects are substantial or trivial when a value for the smallest substantial change is entered.

Coefficients of variation (CV) determined the smallest substantial change for each of the variables. The CV (standard error of measure [SEM] or typical error of measure [TE] expressed as percentage) data reported by previous inter-session reliability studies for each variable were used for the magnitude-based inference analyses. Thus, substantial is an absolute change > 2.5% for measures of range of motion [91], 18.5% for measures of H/Q strength ratios [92], 3.0% for measures of dynamic postural control [104], 1% for 10 and 20 m sprint times [105], 6% for DJ height [106] and 13.9% for measures of reactive strength index [107]. The qualitative descriptors proposed by Batterham and Hopkins [108] were used to interpret the probabilities (clinical inferences based on threshold chances of harm and benefit of 0.5% and 25%) that the true effects are harmful, trivial or beneficial: <1%, almost certainly not; 1–4%, very unlikely; 5–24%, unlikely or probably not; 25–74%, possibly or maybe; 75–94%, likely or probably; 95–99%, very likely; >99%, almost certainly. This approach to qualitatively describe the inferences is based on where the confidence interval of the between-groups differences lies in relation to a 3-level (beneficial, trivial and harmful) scale of magnitudes. For example, whether a confidence interval is entirely within the beneficial range of the smallest substantial change, the effect is clearly beneficial (>99%, almost certainly). Contrarily, if the confident interval spans 2 levels; harmful and trivial or trivial and beneficial, then the inference is qualified with a descriptor that represents the likelihood that the true value will have the observed magnitude (probabilistic inference) [109]. The inference was deemed unclear when the 90% confidence interval of the pre-post change differences overlapped both beneficial and harmful levels.

This spreadsheet also provides estimates of the effect of an intervention adjusted to any chosen value of a covariate, thereby reducing the possibility for confounding effects (e.g., when a characteristic is unequal in the experimental and control groups). Thus, the sex of the participants was included as a covariate.

The current study considered a "substantial" main effect when a change was noted between paired-comparisons in physical performance measures that had reported a probability of the worthwhile differences of "likely" or higher (>75% positive or negative).

3.4 RESULTS

The statistical analysis showed no significant differences (p values from 0.53 to 0.95) in joint range of motions (hip flexion, extension and abduction; knee flexion and ankle dorsiflexion with knee flexed and extended) and dynamic postural control (anterior, posteromedial and posterolateral directions) outcomes between the dominant and non-

dominant limbs of the players. Consequently, the average score of both limbs for each unilateral variable was used for the subsequent statistical analysis. The post intervention results of variables are reported for descriptive purposes in table 3.3

Table 3.3. Post-intervention (Dynamic warm-up, Harmoknee and FIFA 11+) results (mean ± standard deviation [SD]) for physical performance outcomes.

	Variable	Dynamic warm-up	Harmoknee	FIFA 11+
Y-Balance test (cm) ^T				
-	Anterior distance	64.9 ±5.7	63.5 ±4.3	63.6 ±5.1
-	Posteromedial distance	101.7 ±8.6	103.9 ± 7.8	104.4 ± 8.4
-	Posterolateral distance	96.8 ±7.4	95.4 ±6.1	97.2 ±7.8
-	Composite	84.4 ±6.1	84.4 ±4.9	85.3 ±5.8
Sprint time (s)				
-	10 m	1.99 ± 0.17	2.07 ± 0.15	2.12 ±0.18
-	20 m	3.44 ±0.31	3.47 ± 0.27	3.53 ± 0.34
Vertical drop jump				
-	Height (cm)	27.1 ±3.1	22.9 ±2.6	24.6 ±2.9
-	Reactive strength index	1.04 ±0.29	1.07 ±0.27	1.05 ± 0.28
Hamstring-to-quadriceps strength ratios				
-	H/QCONV60	0.60 ± 0.06	0.60 ± 0.07	0.59 ± 0.08
-	H/Qconv180	0.64 ± 0.14	0.66 ±0.13	0.64 ± 0.14
-	H/Qfunct60	1.33 ± 0.34	1.23 ±0.29	1.29 ±0.18
-	H/Qfunct180	1.63 ± 0.41	1.58 ± 0.52	1.63 ±0.41
-	H/Qfunct30/240	1.99 ±0.62	1.92 ± 0.65	2.06 ± 0.47
Joint range of motion (°)				
-	Hip flexion	67.4 ±11.6	67.3 ±11.6	69.9 ±14.9
-	Hip extension	17.1 ±5.1	17.6 ±5.8	16.1 ±4.4
-	Hip abduction	50.4 ±8.7	51.4 ± 8.5	48.7 ± 8.6
-	Knee flexion	123.4 ± 14.9	124.7 ±13.1	124.2 ±11.6
-	Ankle dorsiflexion knee extended	37.9 ±8.1	37.8 ±7.2	37.7 ±8.2
-	Ankle dorsiflexion knee flexed	38.4 ±6.6	39.2 ±7.3	37.1 ±6.4

T: Normalized to limb length expressed as a percentage; s: seconds; cm: centimeter; ^o: degrees;

H: hamstring; Q: quadriceps; CONV: conventional; FUNCT: functional

The paired inter-interventions percentage differences with the corresponding 90% confidence interval for the different physical performance measures are displayed in figure 3.2 (FIFA 11+ warm-up vs. Harmoknee warm-up), figure 3.3 (Dynamic warm-up vs. Harmoknee warm-up) and figure 3.4 (FIFA 11+ warm-up vs. dynamic warm-up). No substantial differences (likely differences with a probability > 75%) were found between paired-comparisons for most of the physical performance measures. Only substantial differences (with a probability of 77%) were found for the sprint time outcomes, so that the FIFA 11+ and Harmoknee warm-ups resulted in slower sprint times in comparison with the dynamic-warm up for 20m (2.4%) and 10m (1.7%) respectively.





Figure 3.2. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the FIFA 11+ and Harmoknee. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

3.5 DISCUSSION

The primary findings of the current study reported that the acute (post-exercise) effects elicited by both the FIFA 11+ and Harmoknee warm-ups on most physical performance measures (with the exception of the sprint times) were similar (not substantial: probabilities of worthwhile differences <75%) to those found by the standard soccer-related dynamic warm-up routine. However, it should be noted that although not substantial and with a practical significance that is unclear, there appears to exist a tendency showing that the Harmoknee elicits superior improvements in the ROM measures than the FIFA 11+ (figure 3.1). Perhaps a probable explanation for this positive tendency in ROM improvements in favor of the Harmoknee may be due to the fact that the Harmoknee, in contrast to the FIFA 11+, includes a specific element of stretching exercises (part number two: muscle activation). On the other hand, it should also be noted that another tendency appears to exist showing that the standard dynamic warm-up routine elicits higher improvements in the H/QFUNC ratios compared to those reported by both the FIFA 11+ and Harmoknee. Although speculative and based on the subjective perception verbally expressed by the participants, the specific eccentric exercise present in both the FIFA 11+ and Harmoknee, the Nordic curl, may have produced a certain degree of fatigue in this sample of players who were not used to performing eccentric exercises in their daily training sessions. As this is the first study (to the authors' knowledge) to explore the postexercise effects of the FIFA 11+, Harmoknee and dynamic warm-up routines on joint ROM measures, dynamic postural control and conventional and functional H/Q ratios, we are not able to make direct comparisons. Consequently, until future studies address this issue, the above-mentioned tendencies should be taken with a high degree of caution.



Figure 3.3. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the Dynamic warm-up and Harmoknee. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

The current study also found that the standard dynamic warm-up routine elicited superior improvements in sprint times (1.7 and 2.4% for 10 and 20 m sprint times, respectively) when they were compared to those produced by the FIFA 11+ and Harmoknee routines (figures 3.2 and 3.3). These findings are not in agreement with the results reported by Bizzini et al. [59], who found that the magnitude of the effects elicited by the FIFA 11+ on sprint times (2.2%) were comparable with those reported in the literature for dynamic warm-up routines (=1.8%). A possible explanation for this discrepancy may be attributed to the different research design used in each study; Bizzini et al. [59] carried out a metaanalysis to compare the effects elicited by the FIFA 11+ with other warm-up routines previously published regarding soccer players, while we directly compared the effects elicited by the FIFA 11+ with a standard soccer-related dynamic warm-up routine. The higher positive effects in the sprint and jump height measures reported by the dynamic warm-up routine in comparison with the FIFA 11+ and Harmoknee routines might be partially due to enhanced activation of the history-dependent neuromuscular factors such as post-activation potentiation (PAP) and stretch-shortening cycle (SSC), which have been found mainly after isometric or resistance/weight-based exercises completed immediately before the task [84,110–112]. Thus, the higher stimuli of resistance-based exercises presented in the exercises belonging to the final part of the dynamic warm-up routine (vertical and countermovement jumps, spiderman walks and sideways low squat walks) may have led to a higher post-activation potentiation in comparison to the FIFA 11+ and Harmoknee. In addition, the previously mentioned possible presence of muscle fatigue as a consequence of the specific and novel eccentric exercise (i.e., Nordic curl) for the participants in both the FIFA 11+ and Harmoknee routines may also have contributed to this finding.

Although the current study is novel in several aspects (testing procedures, statistical analyses and design), some limitations should be noted. One limitation was the small sample size used in each group (interventions or controls). However, the sample size that was enrolled in each group was similar to previous warm-up studies [59,78–82,84,87,89,90] and allowed main effects to be found.

Another possible limitation of the current study is the sampling frame. The age distribution of participants (19.1 \pm 1.3 y) and their physical skills level (amateur) were narrow and so generalizability cannot be ascertained.

3.6 PRACTICAL APPLICATIONS

The main findings of the current study reported that the three interventions (FIFA 11+, Harmoknee and standard dynamic warm-up routines) elicited similar acute (postexercise) effects on several physical performance measures. However, superior positive acute effects on sprint measures were found in favor of the dynamic warm-up routine compared to both the FIFA 11+ and Harmoknee routines. Therefore, and based on the above-mentioned findings, neither the FIFA 11+ nor the Harmoknee routines appear superior to the dynamic warm-up routines currently performed by most soccer players prior to training sessions and matches. However, the FIFA 11+ and Harmoknee should be implemented in everyday soccer sessions as training components (i.e., placed in the main part of the training sessions), based on their demonstrated superior positive effects on injury prevalence rates in comparison with traditional practices.



Figure 3.4. Net effects (expressed as percentage) of the interventions (paired comparisons) on the physical performance measures analyzed for the Dynamic warm-up and FIFA 11+. The probabilities of an effect being harmful/trivial/beneficial are expressed as percentage values. Clinical inference is provided.

3.7 SUPPLEMENTARY FILES

Supplementary file 3.1

The passive hip flexion (passive straight leg raise test [figure a]), extension (Thomas test [figure b]) and abduction (passive hip abduction with knee flexed over the edge of the plinth test [figure c]); knee flexion (Modified Thomas test [figure d]) and ankle dorsiflexion (weight-bearing lunge with knee extended [figure e] and flexed [figure f] tests) range of motions assessment.



Supplementary file 3.2

Isokinetic testing position.



Supplementary file 3.3

Dynamic postural control assessment (anterior [figure a], posteromedial [figure b] and posterolateral [figure c] directions).



CHAPTER 4

Study 2





STUDY 2

TRAINING EFFECTS OF THE FIFA 11+ AND HARMOKNEE ON SEVERAL NEUROMUSCULAR PARAMETERS OF PHYSICAL PERFORMANCE MEASURES

by

Ayala F, Pomares-Noguera C, Robles-Palazón FJ, García-Vaquero MP, Ruiz-Pérez I, Hernández-Sánchez S, & De Ste Croix M.

4.1 Abstract

The main purpose of this study was to analyze the training effects of the FIFA 11+ and Harmoknee on several parameters of physical performance measures in youth amateur soccer players. 41 adolescent players were randomized within each team into 2 groups (team 1: control vs. FIFA 11+; team 2: control vs. Harmoknee). The FIFA 11+ and Harmoknee groups performed the program 3 times a week for 4 weeks; the control groups completed their usual warm-up routines. 13 physical performance measures (joint range of motion, dynamic postural control, single legged hop limb symmetry, sprint time, jumping height and agility) were assessed. All physical performance parameters were compared via a magnitude-based inference analysis. Significant between-group differences (in favor of the FIFA 11+ players) were found for dynamic postural control (anterior [2.5%] and posteromedial [7.2%] distances), single legged hop limb symmetry (side-to-side symmetry during a triple hop test [8.3%]), 10 (8.4%) and 20 (1.8%) m sprint times and jumping height (9.1%) neuromuscular outcomes. For the Harmoknee, significant differences (in comparison to its paired control group) were found only for 10 (2.7%) and 20 (2.9%) m sprint times and jumping height (9.7%). Therefore, the main findings of this study suggest exchanging traditional warm-up programs for the FIFA 11+ in male youth soccer players based on its superior effects on some neuromuscular parameters (sprinting, jumping and stability) of physical performance.

Keywords: soccer, muscle, warm-up, neuromuscular control, sport-related injuries.

4.2 INTRODUCTION

Soccer (associated football) is one of the most popular sports in the world. Despite the numerous health benefits, participation in a physically demanding sport such as soccer can lead to greater exposure to causal factors of injury [24,26]. The increased risk of injury (mainly in the lower extremities) produced by playing soccer is especially relevant in cases in which growth and maturation are not yet completely developed, such as in childhood and adolescence [113]. Specifically, injury incidence in adolescent populations has recently been aligned to peak height velocity, when rapid growth is evident [31,114]. Furthermore, recent epidemiology studies have reported that the frequency and severity of injuries among youth soccer players is striking in comparison to other sports [115–117]. Particularly, injury rates in youth soccer vary from 5.6 injuries per 1000 h of exposure [118] to 37.6 injuries per 1000 match hours [119]. Consequently, participation at a young age if a child or adolescent is unable to continue to participate because of the effects of injury [120].

Therefore, there is a clear necessity to develop and implement measures aimed at preventing and reducing the number and severity of soccer-related injuries in youth players. The FIFA 11+ [69] and Harmoknee [36] are 2 training programs designed to prevent injuries in youth soccer players. Both programs include running exercises and specific dynamic movements focusing on enhancing the most important and modifiable injury risk factors (e. g., poor physical performance in strength, power, balance, speed, proprioception and joint range of motion [53]) and based on scientific evidence and best practice [36,69]. Recent cluster randomized controlled trials have demonstrated that the FIFA 11+ is effective in reducing lower extremity injury rates in male [50,71] and female [35,72] youth players. Although with a smaller evidence base than the FIFA 11+, the effectiveness of the Harmoknee to reduce the incidence of lower extremity injuries (mainly knee injuries) has also been documented in female youth players [36].

Although the main purpose of both FIFA 11+ and Harmoknee is injury prevention, the knowledge of training effects elicited by these training programs on physical performance can help in identifying the potential mechanisms behind the reported reduction in injury incidence [121]. Although some studies have explored the training effects of both programs (mainly FIFA 11+) on some measures of neuromuscular performance (lower extremity balance [49,122] and core stability [121], knee proprioception

[122], knee strength [121,123,124]), and sprinting and jumping ability [121,125,126] (that form part of the athletic physical performance spectrum), none of them have used adolescents players as a sample despite being the target population of both programs.

Therefore, the main purpose of this study was to analyze the training effects of the FIFA 11+ and Harmoknee on several parameters of physical performance in youth amateur soccer players. We hypothesized that these 2 programs would show beneficial and superior effects on physical performance (particularly in balance, sprint and jumping measures) in comparison to the traditional practices as they include specific and novel exercises designated to improve these parameters (e. g.: Nordic hamstring exercise, single-leg stance tasks, multidirectional [vertical, forward, backward, lateral] single leg jumps, running drills [alternate leg bounds, sub- maximal speed lineal running]).

4.3 METHODS

4.3.1 Sample size estimation

The sampling package (sample size estimation, contrast of hypothesis, comparing groups means, independent groups) of the statistical software Epidat 4.1 was used to calculate the sample size needed to detect meaningful changes. The change in the hip flexion ROM was selected as the primary outcome variable for the sample size estimation as pilot studies carried out in our laboratory using different population cohorts suggested that this variable has high inter-session variability.

Thus, an estimation of the desired sample size was made based on the expected difference in the primary outcome variable (hip flexion ROM) of 1.5° (standard deviation of 0.95°) assuming equal variances, with confidence limits of 95 %, a statistical power of 80 % in a Student's t-test for independent groups and an alpha level of 0.05. The analysis reported that a minimal sample size of 8 participants would be required. Considering the possible level of dropout in this type of intervention, an additional 25 % was added to the minimal sample size calculation (10 participants per group) to ensure an appropriate final sample size.

4.3.2 Participants

A total of 60 male youth amateur soccer players were contacted to take part in the study. Participants were recruited from 3 different soccer teams that were engaged in the Official Amateur Championships of the Spanish Soccer Federation (first national juvenile league). The participants met 4 inclusion/exclusion criteria: 1) had no history of impairments to the knee, thigh, hip, or lower back in the 6 months prior to the study; 2) all playing positions except goalkeepers; 3) all participants were free of delayed onset muscle soreness (DOMS) at any testing session; and 4) participants were excluded from the data analysis if they a) missed more than 2 consecutive or 3 non-consecutive training sessions and/or b) one testing session.

Before any participation, experimental procedures and potential risks were fully explained to the participants in verbal and written form, and written informed consent was obtained from players, their parent/guardian and coaches. The Institutional Research Ethics committee approved the study protocol prior data collection, conforming to the recommendations of the Declaration of Helsinki and the IJSM guidelines [127].

Of the 60 players contacted, all players (n = 20) who belonged to one team were excluded from the study because the post-test session was not completed. Therefore, 41 youth male amateur soccer players (age: 16.8 ± 0.7 y; body mass: 70.2 ± 3.5 kg; stature: 173.9 ± 6.7 cm; years playing soccer: 6.1 ± 2.2 y) from 2 different soccer teams completed this study.

4.3.3 Research design

A parallel, 2-group, pre-post, randomized controlled trial with double baseline (2 pre-test sessions) was used to address the purposes of this study.

The study was conducted in Spain and began in February 2015 and was completed in April 2015. In Spain, the first national juvenile league has 2 different rest periods (winter [2–3 weeks for Christmas holidays] and spring [2 weeks for Easter holidays] breaks) so the season is divided into 3 main terms/macrocycles. The 3 terms have approximately the same number of weeks (from 10 to 12 weeks) and matches (from 10 to 12 matches; one every weekend). The time frame of the study was selected so that the study could start after the winter break and could be completed before the play-off/play-out phases. The second term of the season was chosen rather than the first term in order to be sure that the players selected to each team was definitive and stable within the testing period. Further, the study was not carried out in the third term of the season with the aim of reducing the dropout rate of players' that could be expected due to the secondary school final exams (this decision was made based on coaches' experience).

The independent variables were the 3 different intervention programs (control [traditional or regular warm-up], FIFA 11+ and Harmoknee). The dependent variables included 13 physical performance measures (range of motion [hip, knee and ankle joints], dynamic postural control [measured throughout the Y-balance test], single legged hop limb symmetry, sprint times [10 and 20 m], jumping height and agility).

Prior to the intervention phase, the participants' baseline value for each dependent variable was determined using 2 identical testing sessions separated by a week rest-interval. Each testing session was carried out 48–72 h after finishing the previous competitive match (i.e., Tuesday or Wednesday) so that the players could have enough time for recovery. In addition, players did not carry out any training session throughout this rest-interval. Tests were conducted within the time frame of a regular training session at the same time of the day (in the late afternoon or early evening, depending on the team's training schedule). All the tests were carried out on an outdoor training pitch (3G artificial surface). The total testing procedure lasted approximately 2 h for one team. After these 2 pretest sessions were completed, participants were randomized within each team into 2 groups (team 1: control [n = 11] vs. FIFA 11+ [n = 10]; team 2: control [n = 10] vs. Harmoknee [n = 10]) using a computer-based software program. One of the researchers without any contact or knowledge of the players completed the allocation and randomization. Therefore, no allocation concealment mechanisms were necessary.

For the following 4 weeks (intervention phase), the participants completed only one of the 3 intervention programs 3 days a week as part of their weekly training sessions. As the FIFA 11+ and Harmoknee were initially proposed as training programs that should be performed during the pre-exercise warm-ups, the participants who were allocated in the intervention groups carried out the FIFA 11+ or the Harmoknee instead of their traditional or regular warm-up routines. However, prior to the matches played every single weekend (n = 4), all players performed their traditional warm-up routines (this situation was imposed by the coaches).

The training period of 4 weeks was selected: (a) to match the typical duration of each of the 2 mesocycle within each macrocycle of the regular season in this population; (b) to ensure that both the testing and intervention phases of this study were developed during the same period of the season in each team; and (c) it has been defined as the minimum period needed to find improvements in ROM [128], dynamic postural control [129], sprinting and jumping [103] after performing a specific training program.

A trained rehabilitation specialist was assigned to each team for administrating the FIFA 11+ (team 1) and Harmoknee (team 2), and for checking the warm-ups and assisting the coaches during the normal warm-up (control group). All players in the intervention groups attended a workshop designed to demonstrate how to perform the exercises correctly. In order to prevent contamination of the control groups the training pitch was divided into 2 equal parts, so that the players who belonged to the control group performed their regular warm-up in one part while the players who belonged to the intervention group (FIFA 11+ or Harmoknee) performed their new warm-up in another part of the pitch. Two days after the intervention phase, the post intervention assessments were carried out following the same procedure completed during the baseline-testing phase. The testers who conducted the baseline and post intervention assessments were blinded to group assignment.

4.3.4 Testing procedure

During each testing session, participants began by completing a standardized warmup routine consisting of 4–5 min of self-paced low- to moderate-intensity running including forward/backwards movements, sidestepping and general mobilization (i.e., arm circles, leg kicks). After these participants performed 6–8 min of dynamic stretching (i.e., straight leg march, forward lunge with opposite arm reach, forward lunge with an elbow instep, lateral lunge, trunk rotations, multidirectional skippings) performing 3 sets, from low to high intensity, with a 15 s rest period between each set. Following this dynamic stretching routine, participants performed 3 sets of ballistic exercises with a 15 s rest period between each set. Exercises included single hop jumps (5 repetitions), alternate leg bounds (multidirectional x 5 repetitions) and short (5–15 m) accelerations and decelerations in different directions (3 repetitions forwards and 5 repetitions side to side). The assessment of the dependent variables was carried out 3–5 min after the standardized warm-up. The order of the tests was consistent through the experimental sessions and is displayed in figure 4.1.

Dynamic postural control

Dynamic postural control was evaluated using the Y-Balance test and following the guidelines proposed by Shaffer et al. [97]. Players were allowed a maximum of 5 trials to obtain 3 successful trials for each reach direction (anterior [supplementary file 4.1a], posteromedial [supplementary file 4.1b] and posterolateral [supplementary file 4.1c]). Trials were discarded if the player failed to maintain unilateral stance on the platform, failed to maintain reach foot contact with the reach indicator on the target area while the reach indicator is in motion, used the reach indicator for stance support, or failed to return the reach foot to the starting position under control [97]. Specifically, testing order was completed as dominant anterior, non-dominant anterior, dominant posteromedial, nondominant posteromedial, dominant posterolateral, and non-dominant posterolateral. The average of the 3 reaches were normalized by dividing by the previously measured leg length to standardize the maximum reach distance ((excursion distance/leg length) x100 =% maximum reach distance) [98]. Leg length was defined as the length measured in centimeters from the anterior superior iliac spine to the most distal portion of the medial tibial malleolus. To obtain a global measure of the balance test, data from each direction were averaged for calculating a composite score [99]. The dominant lower extremity was defined as the participant's kicking leg (self-reported).



Figure 4.1. Schematic representation of the study design.

Hip, knee and ankle range of motions

The passive hip flexion (passive straight leg raise test [supplementary file 4.2a]), knee flexion (Modified Thomas test [supplementary file 4.2b]) and ankle dorsiflexion (weightbearing lunge with knee extended test [supplementary file 4.2c]) range of motions of the dominant and non-dominant extremities were assessed following the methodology previously described [91,130]. Participants were barefoot and instructed to perform, in a randomized order, 2 maximal trials of each ROM test for each extremity. An ISOMED inclinometer (Portland, Oregon) with a telescopic arm was used as the key measure for all ROM measures. The mean score for each test was used in the subsequent analyses. The same researchers performed the ROM testing at all testing sessions.

Single legged hop limb symmetry

Side-to-side symmetry in jumping distance were evaluated using 2 single-legged hop tests: 1) the single hop for distance (single hop); and 2) triple hop for distance (triple hop).

The single hop was performed with the participant standing on the leg to be tested, hopping as far as possible, and landing on the same leg (supplementary file 4.3a). The triple hop for distance was performed with the participant standing on 1 leg and performing 3 consecutive hops as far as possible (supplementary file 4.3b). Both tests were considered successful if the landing was stable. To be considered a valid trial, the landing must be on one limb, under complete control of the participant. If the participant landed with early touchdown of the contralateral limb, had loss of balance, touched the wall, or had additional hops after landing, the hop was repeated. The hop distance was measured to the nearest centimeter from the starting line to the player's heel with a standard tape measure. Each player performed 3 practice trials and 3 test trials for each of the different hops. The single hop was performed first, followed by the triple hop. Players were instructed to rest between hops whenever needed. For the calculation of the limb symmetry index (LSI) during each of the hop tests, the average values of the 3 successful trials were used. We calculated LSI by using the following formula: (dominant lower extremity/non-dominant lower extremity) × 100 % [131].

10 and 20 m sprint

Owing to its good reproducibility, linear sprint tests ranging from 10 to 20 m are used as general measures of linear speed in soccer players [132]. Time during 10 and 20-m sprint in a straight line was measured by means of single beam photocell gates placed 1.0 m above ground level (Time It; Eleiko Sport, Halmstad, Sweden). Each sprint was initiated from an individually chosen standing position, 50 cm behind the photocell gate, which started a digital timer. Each player performed 2 maximal 10 and 20-m sprint trials interspersed with 3 min of passive recovery, and the fastest time achieved for each distance was recorded.

Vertical drop jump

A vertical drop jump (DJ) without arm swing was performed on a contact platform (Ergojump®, Finland) according to Onate et al. [100]. Participants stood with feet shoulderwidth apart on a 28-cm- high step, 30 cm from the contact platform. They were instructed to lean forward and drop from the step as vertically as possible, in an attempt to standardize landing height. Participants were required to land with one foot on each of the contact platform, then immediately perform a maximal vertical jump, finally landing back on the contact platform. Participants were asked to keep their hands on their hips to prevent the influence of arm movements on vertical jump performance. Each participant performed at least 5 maximal jumps starting from a standing position, with at least 1 min of recovery between jumps. Participants were asked to jump as high as possible. The mean jump height of the best 3 jumps was used for statistical analysis.

Agility

The Illinois agility test is commonly used in measuring agility in soccer [126,133,134]. The length of the zone is 10 m, while the width (distance between the start and finish points) is 5 m. 4 cones were placed in the center of the testing area at a distance of 3.3 m from one another (supplementary file 4.4). The participants started the test lying face down, with their hands at shoulder level. The trial started on the "go" command, and the participants began to run as fast as possible. The trial was completed when the players crossed the finish line without having knocked any cones over. Time was measured using a photocell system (Time It; Eleiko Sport, Halmstad, Sweden). 3 trials were performed by each player with the best time used for analysis [126,133].

4.3.5 Interventions

Control group

Coaches were asked to administer their normal warm-up routines trying to match the duration of the FIFA 11+ and the Harmoknee (20–25 min). The traditional warm-up differed between teams but included a combination of running, stretching, technical exercises with the ball and small-sided games.

FIFA 11+

The FIFA 11+ consisted of 3 parts: the first of which involved running exercises (part 1); the second part covered 6 exercises, all of which comprised 3 levels of difficulty and were aimed at improving strength, balance, muscle control and core stability (part 2). The third and final part consisted of advanced running exercises (part 3). For more details see the manual and instructions freely available on the official website (FIFA-Medical and Assessment Research Centre [F-MARC], FIFA 11+, http://www.f-marc.com/11plus [accessed September 1, 2014]). The players completed The FIFA 11+ 3 times a week for 4 weeks substituting their normal warm-up routine. All players were able to perform level II of difficulty for each exercise in part 2, as confirmed during the introductory workshop. No player was able to perform the level III of difficulty for each exercise and therefore the level II of difficulty was chosen for this study.

Harmoknee

The Harmoknee warm-up program included 5 parts: warm-up, muscle activation, balance, strength and core stability, all of which can be combined and performed in a regular soccer training session (Kiani, Ashkan, Harmoknee, http://www.harmoknee.com/ [accessed June 7, 2014]. Total program duration was 20–25 min [36]). Similar to the FIFA 11+, Harmoknee was also performed 3 times per week for 4 weeks substituting their normal warm-up routine.

4.3.6 Statistical analysis

The distribution of raw data sets was checked using the Kolmogorov-Smirnov test and demonstrated that all data had a normal distribution (p > 0.05).

Dependent sample t-tests were carried out to assess differences between limbs (dominant vs. non-dominant) in dynamic postural control and ROM in pre-test 1 and pre-test 2 measures. In cases where no significant differences were found, the mean value of both limbs was used for the subsequent analyses. Dependent t-tests were also carried out to assess baseline inter-session differences (pre-test 1 vs. pre-test 2) for each dependent variable. If no significant differences were found, the mean value of both testing sessions for each variable was used to assess the effects of the intervention programs. Independent sample t-tests were run to evaluate baseline differences between the groups belonging to the same team (paired comparison: team 1, control vs. FIFA 11+; team 2, control vs. Harmoknee) for each dependent variable (mean of the 2 baseline measures).

Magnitude-based inference analysis of the interventions (FIFA 11+, Harmoknee and control) were estimated using a spreadsheet designed by Hopkins [103] via Student t-test with unequal-variances computed for change scores between paired sessions (team 1: control vs. FIFA 11+; team 2: control vs. Harmoknee) at each testing moment (pre-test [baseline], post-test) for each variable. Alpha was set at p < 0.05. Each participant's change score between pre and post-tests was expressed as a percentage of baseline score via analysis of log-transformed values, to reduce bias arising from non-uniformity of error. In addition, the analysis determines the probability that the true effects are substantial or trivial when a value for the smallest substantial change is entered. This spreadsheet also provides estimates of the effect of an intervention adjusted to any chosen value of a covariate, thereby reducing the possibility for confounding effects (e.g., when a characteristic is unequal in the experimental and control groups). Thus, the baseline pretest value (mean of the 2 pre-test measures) of each dependent variable was included as a covariate to avoid the phenomenon of regression to the mean and thereby obtaining a better estimation of the effects of the FIFA 11+ and Harmoknee interventions in comparison with their paired control groups.

Coefficients of variation (CV) determined the smallest substantial change for each of the variables. The CV (standard error of measure [SEM] or typical error of measure [TE] expressed as percentage) data reported by previous inter-session reliability studies for each variable were used for the magnitude-based inference analyses. Thus, substantial is an absolute change > 2.5% for measures of range of motion [91], 3% for measures of dynamic postural control [104], 1.0% for 10 and 20 m sprint times [105], 1.2% for Illinois agility [135]

and 6% for DJ height [128]. For those variables (single and triple hop measures) where no studies have analyzed the inter-session reliability, 0.20 standardized units (that is a fraction of the between-participants standard deviation at baseline) was chosen as the substantial change [136]. The default of 0.20 gives chances that the true effect is at least small [109].

The qualitative descriptors proposed by Hopkins [109] were used to interpret the probabilities that the true affects are harmful, trivial or beneficial: <1%, almost certainly not; 1–4%, very unlikely; 5–24%, unlikely or probably not; 25–74%, possibly or may be; 75–94%, likely or probably; 95–99%, very likely; >99%, almost certainly. This approach to qualitatively describe the inferences is based on where the confidence interval of the between-groups differences lies in relation to a 3-level (beneficial, trivial and harmful) scale of magnitudes. For example, whether a confidence interval is entirely within the beneficial range of the smallest substantial change, the effect is clearly beneficial (> 99%, almost certainly). Contrarily, if the confident interval spans 2 levels; harmful and trivial or trivial and beneficial, then the inference is qualified with a descriptor that represents the likelihood that the true value will have the observed magnitude (probabilistic inference) [109]. The inference was deemed unclear when the 90% confidence interval of the pre-post change differences overlapped both beneficial and harmful levels. Effect sizes, which are standardized values that permit the determination of the magnitude of differences between groups or experimental conditions [136], were also calculated for each of the variables using the method previously described by Cohen [136]. Cohen [136] assigned descriptors to the effect sizes (d) such that effect sizes less than 0.4 represented a small magnitude of change while 0.41–0.7 and greater than 0.7 represented moderate and large magnitudes of change, respectively. Analyses were completed using SPSS version 20 (SPSS Inc, Chicago, IL, USA) and an online spreadsheet (Hopkins, Will, "pre-post parallel groups trial spreadsheet", http://www.sportsci.org [accessed April 14, 2015]).

4.4 **RESULTS**

The statistical analysis showed no significant differences (p values from 0.43 to 0.87) in Y-balance test (anterior, posteromedial and posterolateral directions) and ROM (hip flexion, knee flexion and ankle dorsiflexion) outcomes between the dominant and non-dominant limbs of the players at either the pre-test 1 or the pre-test 2 sessions. Consequently, the average score of both limbs for each unilateral variable was used for the

subsequent statistical analysis. No significant differences (p values from 0.22 to 0.93; d < 0.4) were found between the values obtained in both pre-test sessions in each variable so that the average score was used as criterion of reference (real baseline score). In addition, there were no paired inter-group differences at baseline for any dependent variable (p values ranging from 0.27 to 0.65).

The pre (baseline) and post intervention results of each group are reported for descriptive purposes in table 4.1 (FIFA 11+ and control) and table 4.2 (Harmoknee and control). The paired inter-group differences after the intervention phase (4 weeks) with the corresponding 90% confidence interval for the physical performance measures are displayed in figures 4.2 and 4.3 for the FIFA 11+ and Harmoknee respectively.

There were substantial (possible and likely substantial differences with a probability ranging from 60 to 91%; d 0.22 to 0.49), and significant (p < 0.05) pre-post change differences between control vs. FIFA 11+ (in favor of the latter) for LSI during triple hop (pre-post change differences ranging from -12.5 to -3.9%), anterior (pre-post change differences ranging from 0.9 to 13.8%) and posteromedial (pre-post change differences ranging from -1.2 to 6.5%) distances reached through the Y-balance test, 10 (pre-post change differences ranging from -18.5 to 3.0%) and 20 m (pre- post change differences ranging from -5.7 to 1 %) sprint times and DJ height (pre-post change differences ranging from -8.5 to 22 %) (figure 4.2). However, no main effects were observed (p > 0.05; trivial effect with a probability of 55–98%; d < 0.2) between control vs. FIFA 11 + for all the ROM measures, LSI during single hop, posterolateral distance, agility and composite score for the Y-balance test. Main effects of the Harmoknee in comparison to its paired control group (likely substantial differences with a probability ranging from 79 to 93%; d ranging from 0.33 to 0.55; p < 0.05) were found for knee flexion ROM (pre-post change differences ranging from -5.4 to -1.3%) 10 (pre-post change differences ranging from -5.3 to -0.1%) and 20 m (prepost change differences ranging from -5.5 to -0.3%) sprint times and DJ height (pre-post change differences ranging from 1.9 to 18%) (figure 4.3).



Percentage difference (%)

-20

Effects of injury prevention programs on physical performance and neuromuscular control

Hip flexion (possibly trivial difference [57%]) Knee flexion (likely trivial difference [76%]) Ankle dorsiflexion (possibly trivial difference [55%])

Single hop (LSI score) (possibly trivial difference [48%]) Triple hop (LSI score) (very likely worthwhile difference [98%])

Anterior distance (likely worthwhile difference [89%]) Posteromedial distance (possibly worthwhile difference [60%]) Posterolateral distance (possibly trivial difference [73%]) Composite (possibly worthwhile difference [55%])

10 m sprint time (likely worthwhile difference [87%])20 m sprint time (possibly worthwhile difference [62%])

Agility Illinois test (possible trivial difference [66%])

Drop jump (possible worthwhile difference [70%])

Figure 4.2. Net effects (expressed as percentage) of the intervention on neuromuscular parameters of physical performance for the "FIFA

11+" and control group.



Percentage difference (%)

Figure 4.3. Net effects (expressed as percentage) of the intervention on neuromuscular parameters of physical performance for the Harmoknee

and control group.
Physical performance measure		FIFA 11+							Control						
		Baseline (pre-test)		Post	-test	% Diff	% Difference		Baseline (pre-test)		Post-test		erence		
		Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD		
Range of motion (°)															
•	Hip flexion	79.3	± 4.5	77.8	± 6.7	-1.9	± 6.5	79.9	± 8.7	76.5	± 7.4	-4.3	± 6.9		
•	Knee flexion	136.1	± 7.7	134.6	±7.6	-1.1	± 5.8	135.3	± 7.0	134.1	± 8.3	-0.9	± 4.3		
•	Ankle dorsiflexion	36.4	± 2.1	39.8	± 3.4	9.3	± 10.1	36.7	± 4.0	36.5	± 2.5	-0.5	± 2.1		
Single legged hop limb symmetries		5													
•	Single hop	91.6	± 11.1	92.7	± 8.7	1.2	± 3.4	98.6	± 8.3	98.8	± 2.3	0.2	± 2.7		
•	Triple hop	94.3	±12.9	90.8	±13.2	-3.7	± 7.8	97.3	± 4.6	102.7	± 6.6	5.5	± 9.5		
Y-Balance test (cm) ^T															
•	Anterior distance	67.5	±6.1	67.3	± 6.3	-0.3	±7.5	70.5	± 5.3	64.8	± 5.2	-8.1	±10.3		
•	Posteromedial distance	112.9	± 4.8	115.3	± 2.8	2.1	± 4.3	112.0	± 7.8	109.2	± 6.3	-2.5	± 5.1		
•	Posterolateral distance	109.2	± 5.3	109.5	± 4.9	0.3	± 3.7	108.3	± 8.5	106.5	± 5.7	-1.7	± 5.4		
•	Composite	96.3	± 4.0	97.3	± 2.7	1.0	± 2.4	96.8	± 6.7	93.2	± 4.9	-3.7	± 5.9		
Sp	rint time (s)														
•	10 m	1.93	± 0.13	2.03	± 0.23	5.20	± 7.80	1.82	± 0.28	2.09	± 0.33	14.80	± 21.10		
•	20 m	3.30	± 0.18	3.24	± 0.15	-1.80	± 6.20	3.28	± 0.15	3.31	± 0.23	0.90	± 5.50		
Ag	ility (s)	16.7	±1.2	16.3	± 0.8	-2.4	± 5.6	16.6	± 1.0	16.4	± 1.4	-1.0	± 3.0		
Vertical drop jump (cm)		26.7	± 2.2	26.6	± 3.5	-0.4	±7.3	24.1	± 2.2	22.4	± 5.5	-7.1	± 17.3		

Table 4.1. Baseline and post-intervention (FIFA 11+ and Control) results (mean ± standard deviation [SD]) for physical performance outcomes. The percentage differences between pre and post-test average values are also reported.

^T: Normalized to limb length expressed as a percentage; s: seconds; cm: centimetre; ²: degrees

Physical performance measure			Control										
		Baseline (pre-test)		Post	-test	% Diffe	% Difference		Baseline (pre-test)		Post-test		erence
		Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
Range of motion (²)													
•	Hip flexion	68.5	± 6.2	69.7	± 6.4	1.8	± 3.6	66.4	± 4.0	67.5	± 3.6	1.7	± 3.1
•	Knee flexion	131.6	± 8.8	130.6	± 8.5	-0.8	± 1.7	133.2	± 4.7	137.1	± 7.6	2.9	± 2.3
•	Ankle dorsiflexion	41.9	± 2.1	41.9	± 2.0	0.0	± 3.5	41.3	±1.8	41.9	± 4.0	1.5	±7.8
Single legged hop limb symmetrie		S											
•	Single hop	100.9	± 6.6	98.0	± 6.1	-2.9	± 6.7	99.9	± 2.1	99.7	± 5.4	-0.2	± 2.9
•	Triple hop	100.9	± 6.3	99.1	± 1.8	-1.8	± 4.7	98.8	± 4.9	94.0	± 5.5	-4.9	± 8.6
Y-Balance test (cm) ^T													
•	Anterior distance	62.7	± 5.2	64.8	±7	3.3	± 6.6	59.6	± 5.2	61.2	± 2.9	2.7	± 7.2
•	Posteromedial distance	95 <mark>.0</mark>	± 4.1	94.5	± 5.9	-0.5	± 4.7	91.4	± 4.6	91.2	± 3.8	-0.2	± 3.1
•	Posterolateral distance	93.0	± 4.1	93.0	± 6.3	0.0	± 3.3	92.4	± 5.1	90.8	± 7.5	-1.7	±7.1
•	Composite	83.6	± 2.8	84.1	± 3.7	0.6	± 3.1	81.0	± 5.0	81.3	± 4.4	0.3	± 2.1
Sp	rint time (s)												
•	10 m	1.81	± 0.05	1.81	± 0.07	0.00	± 2.4	1.86	± 0.04	1.91	± 0.07	2.70	± 3.80
•	20 m	3.25	± 0.09	3.17	± 0.05	-2.50	± 2.1	3.34	± 0.05	3.36	± 0.13	0.60	± 3.10
Ag	;ility (s)	16.6	± 0.3	16.4	± 0.3	-1.1	±1.4	17.2	± 0.4	17.1	± 0.5	-0.6	± 2.6
Vertical drop jump (cm)		23.7	± 3.6	26.1	± 4.1	10.1	± 9.1	24.4	±1.2	24.5	± 2.0	0.4	± 6.7

Table 4.2. Baseline and post-intervention (Harmoknee and Control) results (mean ± standard deviation [SD]) for physical performance outcomes.

 The percentage differences between pre and post-test average values are also reported.

^T: Normalized to limb length expressed as a percentage; s: seconds; cm: centimetre; ²: degrees

4.5 DISCUSSION

The findings of the current study indicate that the training stimuli provided by the implementation of the FIFA 11+ 3 times per week for 4 weeks (12 sessions) appear to be sufficient to elicit substantial improvements in several (dynamic postural control, single legged hop limb symmetry [triple hop], sprint times and jumping height), but not all (joint [hip and knee flexion and ankle dorsiflexion] range of motion and agility measures), of the physical performance parameters analyzed.

No studies appear to have examined the effectiveness of the FIFA 11+ on range of motion, dynamic postural control, sprint, agility and jump physical performance parameters in adolescent amateur soccer players and hence, direct comparisons are not possible. However, previous studies using different soccer player cohorts [121,125,126] have reported conflicting training effects of the FIFA 11+ on dynamic postural control, sprint, jump and agility measures. Thus, whereas Daneshjoo et al. [125], and Kilding et al. [126] reported benefits (compared to traditional warm-up) in jumping height and 20 m sprint time after performing the FIFA 11+ in young adults (age: 18.9 ± 1.4 years) and preadolescent (age: 10.4 ± 1.4 years) soccer players respectively; Impellizzeri et al. [121] did not find meaningful improvements in the dynamic postural control, sprint time (10 and 20 m), jumping height (countermovement jump) and agility in adult players (age: 23.7 ± 3.7 years). These differences in population could be a possible explanation for this discrepancy among the results reported. It is possible that the FIFA11+ is not rigorous enough for adult soccer players to reach a training effect and improve physical performance after implementing the FIFA11+ into their regular training. The timing of the interventions carried out in these studies [121,125,126] and also in our study was the same, (e.g., during the regular season) and hence this aspect may have very little influence on these conflicting results.

As this is the first study (to the authors' knowledge) to explore the effects of the FIFA 11+ on single legged hop limb symmetry and joint ROM measures, we are not able to make comparisons. The absence of improvements in the joint ROM measures after performing the FIFA 11+ was expected because this program does not include any specific group of exercises a priori designed to enhance joint ROM (i.e., passive hip flexion, knee flexion and ankle dorsiflexion ROMs). Consequently, our findings would appear to suggest that the

inclusion of more specific dynamic stretching exercises, emphasizing major lower extremity joint movements, might be required if a major component of the FIFA 11+ is to improve joint ROM and reduce the likelihood of sustaining a muscle strain [137,138].

The current study also found that the Harmoknee injury prevention program only elicited positive training effects (in comparison with its paired control group) after being implemented for 4 weeks on sprint times (-2.7 to -2.9%) and jumping height (9.7%)measures. The magnitude of the above-mentioned improvements in the sprint time and jumping height after the Harmoknee program were higher than those previously reported by Daneshjoo et al. [125]. Daneshjoo et al. [125] reported an improvement of -0.5% and 1.9% for the sprint time and jumping height respectively after implementing the Harmoknee program 3 times per week for 8 weeks in a group of male professionals under 21y-old soccer players. Perhaps, a possible explanation for these conflicting results might be attributed to the different levels of sport performance of the population used in each study (amateurs vs. professionals). It can be argued that given the high level of professional under 21 players, the Harmoknee may not be challenging enough for them to see a training effect [139]. In addition, the results of the current study did not show meaningful improvements in the dynamic postural control, in contrast to the findings reported by Daneshjoo et al. [122], who showed a significant increase (approximately 5.6%) in dynamic postural control. Aside from the different populations used, and in contrast to the other neuromuscular performance variables (sprint and jumping), the training stimuli needed to achieve meaningful improvement in postural control after the implementation of the Harmoknee program might be higher. Thus, while Daneshjoo et al. [122] applied the Harmoknee program for 8 weeks (total number of sessions = 24), we implemented the Harmoknee for just 4 weeks (total number of sessions = 12). Thus, given the intensity and volume of the Harmoknee program used in the current study, 4 weeks (12 sessions) might not be enough to elicit a training response. We were not able to compare the effects of the Harmoknee on the rest of the physical performance parameters selected, as the current study was the first in addressing this issue.

The authors of the current study speculate that the reported improvements in physical performance elicited by the FIFA 11+ program may contribute to a reduction of injury risk in the long-term. Thus, the mechanisms behind the previously reported reduction in injury incidence by the FIFA 11+ in youth soccer players [35,49,50,71,72] might

be particularly associated with enhancements in neuromuscular control and dynamic postural control. However, the mechanisms behind the effectiveness of the Harmoknee to reduce the incidence of lower extremity injuries [36] might be only related to enhancements in neuromuscular control. It has been suggested that for youth athletes, an effective plyometric training load should be approximately 74–88 feet contacts completed within each session, which should be performed through the combination of 4–6 exercises geared toward developing both safe jumping and landing mechanics and also to stress stretch-shortening cycle activity [140,141]. In addition, the training frequency should be at least 2–3 non-consecutive days per week [141]. However, despite the fact that the youth amateur players involved in the current study undertake less plyometric training load e. g., within session (< 60 foot contacts completed through 2–3 exercises) and weekly (3 training session) compared with those competing at a higher level e.g., within session (> 74 foot contacts completed through 4–6 exercises and a weekly frequency of 5 training sessions), it appears that the training stimuli of the FIFA 11+ and Harmoknee may be enough to induce benefits in jumping and sprint performance.

Although the current study is novel in several aspects (testing procedures, statistical analyses and design), some limitations should be noted. One limitation is the lack of a placebo group, defined as a separate group, that in contrast to a typical control group, receives a new sham "placebo" intervention (e.g., upper limb strength training) which is specifically designed to have no real effects on the dependent variables analyzed. In fact, we only used paired control groups as in previous similar studies [121,122,124–126] but we could not control for potential effects of expectations generated by the enrolment in an experimental intervention. However, to avoid or minimizing the placebo effect in measures influenced by motivation or learning, a double baseline (2 pre-test sessions) design was used. Therefore, while we cannot exclude that the trends towards a better performance which can be observed from the inspection of the forest plots (mainly in the FIFA 11+ intervention) are partially or totally due to a placebo effect, the results of the tests were less likely influenced. Another important limitation was the small sample size used in each group (interventions or controls). However, the sample size that was enrolled in each group was similar to previous studies [122,124-126] and allowed main effects were identified. Another possible limitation of the current study is the sample used. The age distribution of participants (16.8 ± 0.7y) and their physical skills level (amateur) narrow the generalizability of these results. In order to minimize the error associated with the players who belonged to

the control group copying and performing any new exercise included in the intervention groups during their regular warm-up, the players of the intervention groups performed their new warm-up in a separate part of the pitch. In addition, a trained rehabilitation specialist was assigned to each team for administrating the interventions and for checking that the control groups did not perform exercise that were not part of their normal warmup. However, we cannot totally exclude the possibility that the players of the control groups might have performed exercise included in the interventions outside of their regular training sessions. In particular, although during the warm-up prior to the normal training sessions they were in a different part of the pitch, they might have been conscious (via eye contact) that their team mates belonging to the intervention groups were performing different warm-up exercises. In addition, it is possible that the players belonging to both groups in each team (control and intervention groups) might have spoken about the interventions during the rest periods of the training sessions or even in the locker room. Furthermore, it is also possible that the players of the control group might have felt less motivated to perform their traditional warm-up routine because they knew they were in the control group and not in the intervention group. Finally, comparison between intervention groups (FIFA 11+ and Harmoknee) were not carried out due the fact that it was impossible to reach an agreement between the coaches of the teams regarding the minutes spent in the different training components (technical-tactical, interval-training, intermittent training, small-sided games, ...) in each training session. Future studies that investigate the effects of longer interventions than that conducted in the current study (> 4 weeks) on several physical performance variables using randomized control trial with placebo control group designs are needed to understand better potential mechanisms behind the reported reduction in injury incidence reported by the FIFA 11+ and Harmoknee programs. In addition, further research is needed to identify the exercises really necessary for preventing injuries in order to develop more time-efficient prevention programs which can be more easily implemented into daily routines.

In conclusion, the findings of the current study reinforce that even short-term adherence to an injury prevention program can have positive effects on performance variables. Thus, the results of this study showed that performing the FIFA 11+ 3 times per week for 4 weeks elicits substantial improvements (in comparison with its paired control group) in dynamic postural control, single legged hop limb symmetry, sprint time and jumping height. Thus, we can speculate that an enhancement in physical performance (neuromuscular control and dynamic postural control) is a possible key mechanism for explaining the injury prevention effect of FIFA 11+. In contrast to the FIFA 11+, the Harmoknee only reported meaningful effects on sprint and jumping ability suggesting improvements in neuromuscular control as the mechanism behind its documented effectiveness to reduce the incidence of lower extremity injuries. Our findings would suggest that the FIFA 11+ is more effective in reducing injury risk factors (poor physical performance in postural control and neuromuscular control) than the Harmoknee program in adolescent male amateur soccer players. We would therefore advocate exchanging traditional warm-up programs for the FIFA 11+ in male youth soccer players.



4.6 SUPPLEMENTARY FILES

Supplementary file 4.1

Dynamic postural control assessment (anterior [a], posteromedial [b] and posterolateral [c] directions).





Supplementary file 4.2

The passive hip flexion (passive straight leg raise test [a]), knee flexion (Modified Thomas test [b]) and ankle dorsiflexion (weight-bearing lunge with knee extended test [c]) range of motion assessment.





Supplementary file 4.3

Single legged hop limb symmetry assessment (single hop for distance [a]; and triple hop for distance [b]).



UNIVERSITAS Miguel Hernández

Supplementary file 4.4

Illinois agility run dimensions and completion route. m = meters.





CHAPTER 5

Study 3



5

STUDY 3

TRAINING EFFECTS OF THE FIFA 11+ KIDS ON PHYSICAL PERFORMANCE IN YOUTH SOCCER PLAYERS: A RANDOMIZED CONTROL TRIAL

by

Pomares-Noguera C, Ayala F, Robles-Palazón FJ, Alomoto-Burneo JF, López-Valenciano A, López-Elvira JL, Hernández-Sánchez S, & De Ste Croix M.

5.1 Abstract

The objective of this study was to analyze the training effects of the FIFA 11+ Kids on several parameters of physical performance in male youth soccer players. Twenty-three youth players were randomized within each team into two groups (control vs. intervention). The intervention group performed the FIFA 11+ Kids program 2 times a week for 4 weeks; the control groups completed their normal warm-up routines. Thirteen physical performance measures (range of motion (hip, knee, and ankle joints), dynamic postural control (measured throughout the Y balance test), 20 m sprint time, slalom dribble with a ball, agility, vertical jumping height [counter movement jump (CMJ) and drop jump (D]), horizontal jump distance, accuracy when volleying a ball [measured throughout the Wall Volley test]} were assessed. All physical performance parameters were compared via magnitude-based inference analysis. Significant between-group differences in favor of the FIFA 11+ players were found for dynamic postural control {anterior [mean and 90% confidence intervals (CI) = 1 cm, from -1.6 to 3.5 cm] and posteromedial (mean and 90% CI = 5.1 cm, from -1.8 to 12 cm) and posterolateral (mean and 90% CI = 4.8 cm, from 0.6 to 9.0 cm) distances}, agility run (mean and 90% CI = 0.5 s, from -0.9 to 0 s), vertical jump height [CMJ (mean and 90% CI = 3.1 cm, from 0.2 to 6.1 cm) and DJ (mean and 90% CI = 1.7 cm, from -0.5 to 3.9 cm)], and horizontal jump distance (mean and 90% CI = 2.5 cm, from -8 to 15 cm). The control groups showed better performance in 20 m sprint time (mean and 90% CI = -0.05 s, from -0.11 to 0.07) and wall volley tests (mean and 90% CI = 0.2, from -0.2to 0.6) compared to the intervention group. The main findings of this study suggest that

just 4 weeks of implementation of the FIFA 11+ Kids produce improved physical performance compared with traditional warm-up routines in youth soccer players.

Keywords: youth athletes, agility, injury prevention, jumping, warm-up.



5.2 INTRODUCTION

Soccer (associated football) is a physically demanding sport that entails sudden acceleration and deceleration, rapid changes of directions, jumping and landing tasks, as well as many situations in which players are involved in tackling to keep possession of or to win the ball [22,23]. These high-intensity situations result in a notable increase in injury risk and are especially relevant in children where individual growth and maturation may predispose youth players to a higher risk [24,26,27]. Epidemiology studies have reported that the frequency and severity of injuries among youth soccer players is striking in comparison to other sports [115–117]. In particular, a recent epidemiological study of young children (7–12 years of age) reported injury incidence rates of 0.61 per 1,000 h of exposure and an increase in the incidence rate with age [142]. While some studies have explored position-related differences in injury incidence [143], it should be noted that there may be differing movement demands based on soccer position; however, no studies have directly explored this within youth soccer.

Therefore, it would appear important to implement effective injury prevention programs early to counter potential injury-related risks. Several injury prevention programs have been designed with the aim of preventing and reducing the number and severity of soccer-related injuries in adolescent players, such as the FIFA 11+, knee injury prevention program and preventing injury and enhancing performance program [36,38,69,144]. All of these programs include running exercises and specific dynamic movements focusing on enhancing physical competence and performance, as it has been suggested as a primary injury risk factors in youth athletes [145] and are based on the injury profile and biological status (i.e., maturation phase) of the target population (>14 years of age). The effectiveness in reducing non-contact overall and overuse lower extremity injury rates of these abovementioned injury prevention program have been documented in male and female youth (aged 13–19 years) players [146].

However, these injury prevention program may not be suitable for younger players (<13 years of age) since maturation seems to affect the incidence and characteristics of injury. Specifically, injury incidence in youth populations has been shown to be higher in adolescents than pre/early pubertal players [30]. Thus, injury incidence has recently been aligned to peak height velocity (adolescence), when rapid disproportional growth is

evident [31]. Furthermore, younger soccer players seem to have more fractures and bone stress, fewer strains and sprains, and more injuries of the upper body than adolescent players [142]. These considerations have led some authors to develop a warm-up program designed to prevent injuries and reduce the number and severity of soccer-related injuries in children who take age-specific injury characteristics and physical maturity into account [28]. The warm-up program "FIFA 11+ Kids" is intended to prevent and reduce the number and severity of soccer-related injuries by enhancing children's fundamental and sport-specific motor skills through a range of evidence-based exercises. In particular, the FIFA 11+ Kids focus on: (a) spatial orientation, anticipation, and attention, particularly while dual-tasking (to avoid unintended contact with other players or objects); (b) body stability and movement coordination (more general than specific neuromuscular or proprioceptive training); and (c) learning appropriate fall techniques (to minimize the consequences of unavoidable falls). Accordingly, this warm-up program may contribute to achieving the aims of the earlier stages [(1) Active Start, (2) FUNdamentals, and (3)Learn to Train] of the long-term athletic development conceptual model for late specialization sports [147,148].

Only one large-scale cluster-randomized controlled trial has evaluated the effectiveness of the FIFA 11+ Kids showing a 38% reduction in injuries, with severe injuries reduced by more than 50%, compared to a control group (data available at https://www.fifamedicinediploma.com/lessons/prevention-fifa11-kids/).

However, the training effects elicited on movement skills and physical performance on the potential mechanisms behind the reported reduction in injury incidence remain to be elucidated. To the best of our knowledge, only some authors have examined the preexercise effects of the FIFA 11+ Kids on various physical performance variables in children soccer players, showing possibly beneficial effects in static and dynamic balance, jumping performance, and slalom dribbling [28].

Therefore, the main purpose of this study was to analyze the training effects of the FIFA 11+ Kids on several parameters of physical performance in young soccer players. We hypothesized that this new program would show beneficial and superior effects on physical performance (particularly in balance and agility measures) in comparison to the traditional practices as they include specific and novel exercises designed to improve physical competency.

5.3 MATERIALS AND METHODS

5.3.1 Research design and participants

A total of 26 male youth soccer players (age: 11.8 ± 0.3 years; stature: 144.7 ± 5.1 cm; body mass: 39.4 ± 5.5 kg) took part in the current study. Participants were recruited from two different soccer teams that were engaged in the Official Amateur Championships of the Spanish Soccer Federation (under 12 years regional league). All participants were classified as pre-peak height velocity (PHV) (-3 years to -1 years from PHV) and met three inclusion criteria: (1) had no history of impairments to the knee, thigh, hip, or lower back in the 6 months prior to the study; (2) all participants were free of self-reported delayed onset muscle soreness at any testing session (self-reported); and (3) participated in two training sessions per week (1.5–2 h per session). In addition, two exclusion criteria were also established: (1) missed two consecutive or three non-consecutive training sessions and (2) missed one testing session [149].

All experimental procedures and potential risks were explained fully both verbally and in writing to the participants, and written informed consent was obtained from players, their parent/guardian, and coaches. The Institutional Research Ethics committee of the Miguel Hernandez University of Elche (DPS.FAR.01.14), conforming to the recommendations of the Declaration of Helsinki, approved the study protocol prior data collection.

Twenty-three young soccer players from two different soccer teams completed this study. Three players who belonged to one team were excluded from the study because they missed more than three non-consecutive training sessions. The participants' characteristics can be observed in table 5.1.

A parallel, two-group, pre-post, randomized controlled trial with double baseline (two pre-test sessions) was used to address the purposes of this study.

The study was conducted between February and April 2015. In Spain, the under 12 local league has two rest periods [winter (2–3 weeks for Christmas holidays) and spring (2 weeks for Easter holidays) breaks] so the season is divided into three main terms/macrocycles. The three terms have approximately the same number of weeks (8–10 weeks) and matches (8–10 matches; one every weekend). The time frame of the study was

selected so that the study started after the winter break and could be completed before the Easter break. The second term of the season was chosen rather than the first term in order to be sure that the players selected to each team was definitive and stable within the testing period. Further, the study was not carried out in the third term of the season with the aim of reducing the dropout rate of players' that could be expected due to the primary school final exams.

0 1	1 /
Age (years)	11.8 ± 0.3
Stature (cm)	144.7 ± 5.1
Body mass (kg)	39.4 ± 5.5
Body mass index (Kg/m²)	18.7 ± 2.1
Maturation offset (years from PHV)	-2.4 ± 0.4

Table 5.1. Demographic variables for the children soccer players*.

* All values are mean \pm SD

The independent variables were the two different warm-up programs [control (traditional or regular warm-up) and intervention (FIFA 11+ Kids)]. The dependent variables included 13 physical performance measures {range of motion (hip, knee, and ankle joints), dynamic postural control (measured throughout the Y balance test), 20 m sprint time, vertical jump height [counter movement jump (CMJ) and drop jump (DJ)], horizontal jump distance, accuracy when volleying a ball (measured throughout the wall volley test), slalom dribble and agility}.

Prior to the intervention phase, the participants' baseline value for each dependent variable was determined using two identical testing sessions separated by a week rest-interval. Each testing session was carried out 48–72 h after finishing the previous competitive match (i.e., Tuesday or Wednesday) so that the players had enough time for recovery. In addition, players did not carry out any training session throughout this rest-interval. Tests were conducted within the time frame of a regular training session at the same time of the day (in the late afternoon). All the tests were carried out on an outdoor training pitch (3G artificial surface). The total testing procedure lasted approximately 2 h for one team. After the two pre-test sessions were completed, participants were randomized within each team into two groups [team 1: control (n = 6) vs. FIFA 11+ Kids (n = 6); team 2: control (n = 4) vs. FIFA 11+ Kids (n = 7)] using a computer-based software program. One of the researchers without any contact or knowledge of the players

completed the allocation and randomization. Therefore, no allocation concealment mechanisms were necessary.

For the following 4 weeks (intervention phase), the participants completed only one of the two intervention programs 2 days a week as part of their weekly training sessions. Prior to competitive matches, all players performed their normal warm-up routines (this was imposed by the coaches of both teams).

The training period of 4 weeks was selected (a) to match the typical duration of each of the two mesocycle that make up the three macrocycles of the regular season and (b) to ensure that both the testing and intervention phases of the study were completed during the same period of the season in each team.

A Master in Sports Science student was assigned to both teams for administrating the FIFA 11+ Kids and for checking that the coaches delivered their normal warm-ups in the control group. All players in the intervention groups attended a workshop designed to demonstrate how to perform the exercises correctly. In order to prevent contamination of the control groups, the training pitch was divided into two equal parts, so that the players who belonged to the control group performed their regular warm-up in one part while the players who belonged to the intervention group performed their new warm-up in another part of the pitch. Two days after the intervention phase, the post-intervention assessments were carried out following the same procedure completed during the baseline-testing phase. Due to organizational reasons, the same two testers (one tester conducted the tests while the other tester recorded the data) who conducted the baseline and post-intervention assessments were not blinded to group assignment.

5.3.2 Testing procedure

During each testing session, participants began by completing a standardized warmup routine consisting of 4–5 min of self-paced low- to moderate-intensity running including forward/backward movements, sidestepping, and general mobilization (i.e., arm circles, leg kicks). After this, participants performed 6–8 min of dynamic stretching (i.e., straight leg march, forward lunge with opposite arm reach, forward lunge with an elbow instep, lateral lunge, trunk rotations, multidirectional skippings) performing three sets, from low to high intensity, with a 15 s rest period between each set. The assessments of the dependent variables were carried out 3–5 min after the standardized warm-up. The order of the tests was consistent through the experimental sessions and is displayed in figure 5.1.

Dynamic Postural Control

Dynamic postural control was evaluated using the Y balance test. Players were allowed a maximum of five trials to obtain three successful trials for each reach direction (anterior, posteromedial, and posterolateral). Trials were discarded if the player failed to maintain unilateral stance on the platform, failed to maintain reach foot contact with the reach indicator on the target area while the reach indicator is in motion, used the reach indicator for stance support, or failed to return the reach foot to the starting position under control [97]. Specifically, testing order was completed as dominant anterior, non-dominant anterior, dominant posterolateral. The best of the three reaches was normalized by dividing by leg length to standardize the maximum reach distance ((excursion distance/leg length) $\times 100 = \%$ maximum reach distance) [98]. Leg length was defined as the length measured in centimeters from the anterior superior iliac spine to the most distal portion of the medial tibial malleolus. To obtain a global measure of the balance performance, data from each direction was averaged to determine a composite score [98].

20-m Sprint

Time during a 20-m sprint in a straight line was measured by means of single beam photocell gates placed 0.3 m above the ground level (Time It; Eleiko Sport, Halmstad, Sweden). Each sprint was initiated from an individually chosen standing position, 30 cm behind the photocell gate, which started a digital timer. Each player performed three maximal 20 m sprints interspersed with 3 min of passive recovery, and the fastest time achieved was retained [150].

Agility

The Illinois agility test is commonly used in measuring agility in soccer [28,76,126]. The length of the zone is 10 m, while the width (distance between the start and finish points) is 5 m. Four cones were placed in the center of the testing area at a distance of 3.3 m from one another. Four cones were used to mark the start, finish, and two turning points. The participants started the test lying face down, with their hands at shoulder level. The trial started on the "go" command, and the participants began to run as fast as possible.



Figure 5.1. Schematic representation of the study design.

The trial was completed when the players crossed the finish line without having knocked any cones over. Time was measured using a photocell system (Time It; Eleiko Sport, Halmstad, Sweden). Each player performed three trials with the best score (time) used for analysis.

Slalom Dribble

The slalom dribble course was 20 m in length. Participants ran with the ball in a zigzag fashion around five cones placed in a straight line 4.5 m away from one another. The run time was measured using two photoelectric timing gates. Participants started 0.3 m in front of the starting line and performed four repetitions with the best score (time) used for analysis.

Wall Volley Test

The wall volley test required players to pass the ball through the air against a wall, control the rebound, and make as many direct air-borne passes against the wall as possible, within a time limit of 30 s. The outcome was the absolute number of correct rebounds [151]. The player was placed in a field which was 2 m wide and 0.5 m away from the wall. Only rebounds accomplished while standing in the sector were counted. After two familiarization tests, participants completed two repetitions and the best score was used for analysis.

Standing Long Jump

The participant stood behind the starting line and was instructed to push off vigorously and jump as far as possible. The participant had to land with the feet together and to stay upright. Arm swing was not permitted. Jump distance was measured from the take off line to the back of the heel. After two familiarization tests, participants performed three repetitions. The best score of the three repetitions was selected for the subsequent analysis.

Counter Movement Jump

A CMJ without arm swing was performed on a contact platform (Ergojump1,

Finland) [152]. During the CMJ, the participants first stood upright, then squatted to a selfselected depth of approximately 90° knee flexion, and jumped immediately as high as possible. Players were asked to keep their hands on their hips to prevent the influence of arm movements on vertical jump performance. In addition, players were allowed to perform a countermovement with the lower limbs before jumping. Each player performed five maximal CMJs interspersed with 45 s of passive recovery, and the best jump height was recorded.

Drop Jump

A vertical DJ without arm swing was performed on a contact platform (Ergojump®, Finland) [100]. Participants stood with feet shoulder-width apart on a 28-cm-high step, 30 cm from the contact platform. They were instructed to lean forward and drop from the step as vertically as possible, in an attempt to standardize landing height. Participants were required to land with one foot on each of the force plates, then immediately perform a maximal vertical jump, finally landing back on the contact platform. Participants were asked to keep their hands on their hips to prevent the influence of arm movements on vertical jump performance. Each participant performed at least five maximal jumps starting from a standing position, with at least 1 min of recovery between jumps. Participants were asked to jump as high as possible with the shortest contact time with the ground. The best jump height was used for statistical analysis.

Hip, Knee, and Ankle Range of Motions

Passive hip flexion [passive straight leg raise test], knee flexion [Modified Thomas test], and ankle dorsiflexion [weight-bearing lunge with knee extended test] range of motion of the dominant and non-dominant extremities were assessed [91]. Participants were barefoot and instructed to perform in a randomized order and two maximal trials of each range of motion test for each extremity. The best score for each test was used in the subsequent analyses. The same researchers performed the ROM testing at all testing sessions.

5.3.3 Interventions

Control Group

Coaches were asked to administer their normal warm-up routines trying to match the duration of the FIFA 11+ Kids (15–20 min). The traditional warm-up differed slightly between teams but included a combination of running, stretching, technical exercises with the ball and games.

FIFA 11+ Kids

The FIFA 11+ Kids consisted of seven different exercises: a running game, two jumping exercises, a balance/coordination task, two exercises targeting body stability, and an exercise to improve falling technique [28]. The program has a modular structure and consists of three skill levels with progressive load. The players completed the FIFA 11+ Kids two times a week for 4 weeks substituting their normal warm-up routine. All players were able to perform the level II of difficulty for each exercise in part 2 properly as confirmed during the habituation workshop. However, no player was able to perform the level III of difficulty for each exercise, and therefore, the level II of difficulty was used.

5.3.4 Statistical analysis

The distribution of raw data sets was checked using the Kolmogorov–Smirnov test and demonstrated that all data had a normal distribution (p > 0.05).

Dependent sample *t*-tests were carried out to assess differences between limbs (dominant vs. non-dominant) in dynamic postural control and range of motion pre-test 1 and pre-test 2. In cases where no significant differences were found, the mean value of both limbs was used for the subsequent analyses. Dependent *t*-tests were also carried out to assess baseline intersession differences (pre-test 1 vs. pre-test 2) for each dependent variable. If no significant differences were found, the mean value of both testing sessions for each variable was used to assess the effects of the intervention programs. Contrarily, if significant differences were found, the highest value of both testing sessions for each variable was used for the magnitude-based inference analysis of the interventions. Independent sample t-tests were run to evaluate baseline differences between the groups (intervention vs. control) for each dependent variable (mean of the two baseline measures). Magnitude-based inference analysis of the warm-up programs (intervention and control)

were estimated using a spreadsheet via Student's t-test with unequal variances computed for change scores between paired sessions (intervention vs. control) at each testing time [pre-test (baseline), post-test] for each variable [103]. Alpha was set at p < 0.05. Each participant's change score between pre- and post-tests was expressed as a percentage of baseline score via analysis of log-transformed values, to reduce bias arising from nonuniformity of error.

This approach of data analysis uses confidence intervals (CI) to calculate the probability that a difference is of practical relevance or trivial when a value for the smallest worthwhile change is entered. A difference score of at least 0.2 of the between-participant SD (representing a small effect) was considered to be practically worthwhile [153]. The qualitative descriptors were used to interpret the probabilities that the true affects are harmful, trivial, or beneficial: <1%, almost certainly not; 1–4%, very unlikely; 5–24%, unlikely or probably not; 25–74%, possibly or may be; 75–94%, likely or probably; 95–99%, very likely; >99%, almost certainly [109]. This spreadsheet also provides estimates of the effect of an intervention adjusted to any chosen value of the covariate, thereby reducing the possibility for confounding of the effect when a characteristic is unequal in the experimental and control groups: thus, the baseline pre-test value (mean of the two pretest measures) of each dependent variable was included to avoid the phenomenon of regression to the mean and thereby obtaining a better estimation of the effects of the warm-up programs in comparison with their paired control groups.

5.4 **Results**

t-Tests demonstrated no significant differences (p values from 0.06 to 0.97) in Y balance test (anterior, posteromedial, and posterolateral directions and composite score) and ROM (hip flexion, knee flexion, and ankle dorsiflexion) outcomes between the dominant and non-dominant limbs of the players at either pre-test 1 or pre-test 2. Consequently, the average score of both limbs for each unilateral variable was used for the subsequent statistical analysis. No significant differences (p values from 0.08 to 0.97) were found between the scores obtained in both pre-test sessions in each variable (with the exception of the Illinois test [mean pre-test 1 vs. pre-test 2 difference = 0.24 ± 0.42 s; p = 0.01]) so the average score was used as criterion of reference (baseline score) (table 5.2). In addition, there were no paired inter-group differences at baseline for any dependent

variable (p values ranging from 0.24 to 0.89). The pre (baseline) and post-intervention results of each group are reported for descriptive purposes in table 5.3 (FIFA 11+ Kids and control).

The inter-groups differences after the intervention phase (4 weeks) with the corresponding 90% CI for the physical performance measures are displayed in figure 5.2. Very likely and likely beneficial effects favoring the intervention group were observed in CMJ height and DJ height, respectively. Possibly beneficial effects in favor of the intervention group were observed in Y balance test (anterior, posteromedial, and posterolateral distances and composite score), agility run, standing long jump, and knee flexion ROM measures. Likely trivial effects were found for Slalom dribble and for hip flexion and ankle dorsiflexion ROM measures. The control group showed better performance in 20 m sprint time and wall volley tests compared to the intervention group.

5.5 DISCUSSION

The findings of the current study indicate that the training stimuli provided by the implementation of the FIFA 11+ Kids two times per week for 4 weeks (8 sessions) appears to be sufficient to elicit small to moderate improvements in some [dynamic postural control, agility run, and jumping (standing long jump, CMJ, and DJ) measures] but not all [20 m sprint time, slalom dribble, wall volley, and ROMs (with the exception of the knee flexion ROM) measures] of the physical performance parameters analyzed.

Similar improvements in dynamic postural control (Y balance anterior distance), agility run and jumping measures have been reported after the implementation of the FIFA 11+ Kids twice a week for 10 weeks in a large cohort of young soccer players [28]. However, and in contract to our results, same authors found improvements in slalom dribble and wall volley measures. Perhaps, a possible explanation for these conflicting results might be attributed to the different duration of the intervention phases (10 vs. 4 weeks). Therefore, given the intensity and volume of the FIFA 11+ Kids program used in the current study, 4 weeks (8 sessions) might not be enough to elicit training responses in sprinting times and soccer-specific coordinative tasks measures (slalom dribble and wall volley).

Variable	Pre-	test 1	Pre-t	est 2	Di	P level	
Y balance test ^T							
 Anterior distance 	75.1	± 12.1	75.8	± 10.7	0.7	(-2.5 to 3.9)	0.63
 Posteromedial distance 	107.7	±14.0	108.9	± 12.9	1.2	(-2.1 to 4.6)	0.44
 Posterolateral distance 	110.5	± 17.9	111.5	± 15.6	1.0	(-2.3 to 4.4)	0.52
 Composite 	97.8	± 13.3	98.8	± 12.2	1.0	(-1.2 to 3.2)	0.65
20 m sprint time (s) ^b	3.83	± 0.18	3.81	± 0.18	-0.02	(-0.09 to 0.04)	0.51
Agility run (s) ^b	20.0	± 0.8	19.8	± 0.8	-0.24	(-0.43 to 0.06)	0.01
Slalom dribble (s) ^b	6.12	± 0.34	6.14	± 0.39	0.02	(-0.16 to 0.20)	0.82
Wall Volley (n)	3.0	± 0.9	2.7	± 0.4	-0.3	(-0.6 to 0.1)	0.08
Standing long jump (cm)	142.6	± 10.8	142.6	±14.9	0.0	(-3,6 to 3.7)	0.97
Counter movement jump (cm)	23.2	± 3.4	22.9	± 2.5	-0.3	(-0.7 to 0.1)	0.59
Drop jump (cm)	19.7	± 2.9	20.5	± 3.3	0.8	(-0.2 to 1.9)	0.13
Range of motion (²)							
Hip flexion	71.6	± 5.2	71.7	± 5.3	0.1	(-0.1 to 0.4)	0.27
Knee flexion	129.8	± 5.9	129.9	± 6.1	0.1	(-0.2 to 0.5)	0.53
 Ankle dorsiflexion 	31.3	± 3.9	31.3	± 3.8	0.0	(-0.4 to 0.4)	1.00

Table 5.2. Descriptive pre-tests statistics (mean \pm standard deviation) and differences between pre-test 2 and pre-tests 1 (mean \pm 95% confidence intervals).

.

°: degrees; s: seconds; cm: centimeters; ^b: smaller values represent better results; ^T: normalized to limb length expressed as a percentage



Figure 5.2. Standardized mean differences (90% confidence interval) between the FIFA 11+ Kids and control group in the physical performance parameters analyzed.

	FIFA 11+ Kids							Control						
Physical performance measure	Baseline		Post	Post-test		Difference		Baseline		Post-test		oifference		
	Mean	SD	Mean	SD	Mean	95%CI	Mean	SD	Mean	SD	Mean	95%CI		
Y Balance test ^T														
 Anterior distance 	75.8	±14.2	77.2	± 12.8	1.4	(-0.2 to 3.2)	75.1	±14.2	75.6	± 5.6	0.5	(-1.5 to 2.7)		
 Posteromedial distance 	109.5	± 14.4	112.9	±14.9	3.5	(-1.1 to 8.4)	106.9	±11.3	106.3	± 9.9	-0.6	(-6.9 to 4.0)		
Posterolateral distance	113.1	±19.8	116.2	± 17.6	3.1	(-0.1 to 7.1)	108.3	± 10.7	107.4	± 9.9	-0.9	(-3.7 to 1.2)		
Composite	99.5	± 15.5	102.1	±14.3	2.7	(-0.1 to 5.7)	96.7	±7.5	96.4	±7.4	-0.3	(-3.3 to 2.1)		
20 m sprint time (s) ^b	3.79	± 0.21	3.78	± 0.25	-0.01	(-0.05 to 0.03)	3.87	± 0.06	3.83	± 0.11	-0.04	(-0.1 to -0.01)		
Agility run (s) ^b	20.0	± 0.8	19.5	± 0.8	-0.4	(-0.62 to -0.17)*	19.7	± 0.7	19.8	± 0.9	0.07	(-0.37 to 0.47)		
Slalom dribble (s) ^b	6.13	± 0.32	5.42	± 0.33	-0.71	(-0.86 to -0.56)*	6.13	± 0.29	5.43*	± 0.38	-0.07	(-0.9 to -0.5)		
Wall Volley (n)	2.9	± 0.6	3.1	± 0.5	0.2	(-0.09 to 0.41)	2.8	± 0.6	3.2	± 0.7	0.4	(0.04 to 0.75)		
Standing long jump (cm)	143.0	± 15.5	146.1	±19.6	3.1	(-5.5 to 10.7)	142.1	± 6.8	141.7	± 10.2	-0.3	(-10.9 to 19.1)		
Counter movement jump (cm)	22.7	± 2.9	24.5	± 5.3	1.8	(0.6 to 4.1)	23.5	± 2.7	21.8	± 3.2	-1.7	(-3.1 to 1.5)		
Drop jump (cm)	19.8	± 3.2	21.6	± 4.2	1.8	(0.3 to 3.2)	20.5	± 2.4	20.4	± 3.2	-0.1	(-1.7 to 1.6)		
Range of motion (^o)														
 Hip flexion 	72.1	± 6.0	73.8	±6.6	1.7	(-1.1 to 4.9)	71.1	± 4.2	72.1	± 4.0	1.0	(-0.3 to 2.2)		
 Knee flexion 	129.7	± 6.8	131.1	± 6.8	1.4	(-0.7 to 3.4)	130.1	± 4.9	129.1	± 4.4	-1.0	(-3.5 to 1.9)		
 Ankle dorsiflexion 	30.9	± 3.9	32.5	± 4.5	2.6	(-0.4 to 3.3)	31.9	± 3.9	33.6	± 3.9	1.7	(-0.5 to 5.1)		

Table 5.3. Baseline (pre-test) and post-intervention (FIFA 11+ and Control) results (mean ± standard deviation [SD]) for physical performance outcomes.The differences between pre- and post-test average values are also reported (mean and 95% confidence intervals [CI]).

^T: Normalized to limb length expressed as a percentage; s: seconds; cm: centimetres; ⁹: degrees; ^b: smaller values represent better results; *: p < 0.05

Force platforms have been used as a criterion measure in scientific setting to assess postural control by interpretation of parameters derived from the center of pressure (COP) such as velocity and area of COP displacement [154]. However, because of its high cost and the need for sophisticated instruments, qualified technicians, and time constraints, the use of this method is limited in sports settings. Consequently, the field-based Y balance test has been recommended as an alternative to force platforms for estimating dynamic postural control in sports setting because: (a) it offers sufficient challenge for dynamic postural control as the subject must maintain balance on a single limb, whilst the other limb carries out a series of reaching tasks (operationally valid); (b) its procedure is simple to administer; (c) instructions are easy to follow; (d) scores are easy to explain; (e) the movements require minimal skills training; and (f) large numbers can be tested in a short period of time [98]. Furthermore, the Y balance test has been shown to be sensitive enough: (a) to detect dynamic postural control deficits in patients with chronic ankle instability [155], patellofemoral pain syndrome [156] and anterior cruciate ligament deficiency [157]; (b) to identify athletes at high risk of non-contact lower extremity injury [158,159]; and (c) to monitor the rehabilitation and return to play processes [160]. Furthermore, the test has also been shown to have high intra and inter-tester reliability [161]. Studies looking at the influence of the FIFA 11+ in adult males have reported improvements in dynamic postural control, agility run, jump performance, and balance [69]. The very likely positive effects of jump performance in the current study show probable improvements in power in this group of young children. The possibly beneficial effects of the FIFA 11+ Kids on balance and agility found in the current study are important as balance and coordination develop throughout maturation [162]. To our knowledge this is the first study to show improvements in balance and agility in prepubertal children following the introduction of a short, focused warm-up program for only 4 weeks. These findings highlight the benefits of such programs to development movement competency in young children, which is essential for athletic performance and which forms the foundation of the youth physical development model [163] and also for injury prevention.

As this is the first study (to the authors' knowledge) to explore the effects of the FIFA 11+ Kids on ROM measures, we are not able to make comparisons. The absence of improvements in the joint ROM measures for the ankle and hip, after performing the FIFA 11+ Kids was expected because this program does not include any specific group of exercises a priori designed to enhance their ROM. However, the possibly beneficial increase

in knee ROM is encouraging, given that restricted knee flexion ROM on landing is a risk factor for high ligament loading.

The findings of the current study speculate that the reported improvements in physical performance elicited by the FIFA 11+ Kids program may contribute to a reduction of injury risk in the long-term application. Thus, the mechanisms behind the promising reduction in injury risk by adhering to the FIFA 11+ Kids in prepubertal soccer players might be particularly associated with enhancements in tendon stiffness, balance coordination and physical competency.

5.6 PERSPECTIVES AND LIMITATIONS

An important limitation was the small sample size used in each group (interventions or controls). However, despite the sample size that was enrolled in each group significant main effects were still found. In order to minimize the error associated with the players who belonged to the control group copying and performing any new exercise included in the intervention groups during their regular warm-up, the players of the intervention groups performed their new warm-up in a separate part of the pitch. In addition, a trained rehabilitation specialist was assigned to each team for administrating the interventions and for checking that the control groups did not perform exercise that were not part of their normal warm-up. However, we cannot totally exclude the possibility that the players of the control groups might have performed exercise included in the interventions outside of their regular training sessions. Likewise, and in order to avoid the possible expectation bias of the intervention and control groups, participants did not received information regarding which warm-up program (FIFA11+ Kids or traditional warm-up) was expected to achieve better results in the dependent variables. As we mentioned before, we can neither excluded the possibility that the participants might have sought information about the potential effects of the FIFA+ Kids program on physical performance reported in previous studies. However, neither trainers nor any member of the research staff reported any suspicion about this issue and hence, we consider that this potential source of bias was negligible.

Future studies that investigate the effects of longer interventions than that conducted in the current study (>4 weeks) on several physical performance variables using randomized control trial designs are needed to understand better potential mechanisms behind the reported reduction in injury incidence reported by the FIFA 11+ Kids.

5.7 CONCLUSION

Given the improvements in jump performance, balance, and agility, our study would advocate the introduction of these essential movement competency skills in prepubertal children. It would appear that a structured warm-up routine, such as the FIFA 11+ Kids, focusing on balance, jumping/landing, coordination, and stability should be advocated at an early age. These movements should be focused and delivered early in any long-term athlete development model to enhance movement literacy in young athletes [164].



CHAPTER 6

Epilogue


6

EPILOGUE

6.1 GENERAL CONCLUSIONS

The studies included in the present doctoral thesis:

- a) Describe the acute (post-exercise) and training (long lasting) effects of the FIFA 11+, Harmoknee and dynamic warm-up routines on major physical performance and neuromuscular control measures in male young (adolescents) soccer players.
- b) Analyze the training effects elicited by the FIFA 11+ Kids on measures of physical performance in male children soccer players.
- c) Provide robust arguments in favor of the implementation of the FIFA 11+ Kids and FIFA 11+ warm-up strategies, as alternatives to traditional practices, in everyday training sessions and competitions in children and adolescent soccer players, respectively.

Overall, the main findings of the current doctoral thesis may help clinicians, coaches and sports science specialists in the decision-making process of injury prevention and sport performance optimization.

The following summarizes the major contributions of this thesis:

Study 1. *Acute effects of three neuromuscular warm-up strategies on several physical performance measures in football players.*

- 1. Both the FIFA 11+ and the Harmoknee programs provoke similar acute effects on several physical performance and neuromuscular control measures to the warm-up routines currently performed by most young soccer players prior to training sessions and matches.
- 2. The FIFA 11+ and Harmoknee may be considered "good" or "appropriate" warm-up programs to be performed prior to formal training and competition in youth soccer as they have demonstrated eliciting physiological acute changes (post-exercise effects) to positively affect the major physical performance

measures (i.e., sprinting and jumping) and neuromuscular injury risk factors (i.e., dynamic postural control, H/Q strength ratios).

Study 2. *Training effects of the FIFA 11+ and Harmoknee on several neuromuscular parameters of physical performance measures.*

- 3. The training stimuli provided by the implementation of the FIFA 11+, three times per week for four weeks (12 sessions), appears to be sufficient to cause substantial improvements in several physical performance (sprint time and jumping height) and neuromuscular (dynamic postural control, single legged hop limb symmetry [triple hop]) measures in young amateur soccer players.
- 4. Enhancements in physical competency and neuromuscular control are possible key mechanisms for explaining the well-documented injury prevention effect of FIFA 11+.
- 5. The Harmoknee program, when it is regularly performed (three times per week for four weeks) in training sessions as substitute to traditional warm-ups, may produce significant improvements in sprint and jump performance in young amateur soccer players.
- 6. The FIFA 11+ seems to be more effective in reducing injury risk factors (poor physical performance in postural control and neuromuscular control) than the Harmoknee and traditional warm-up routines in adolescent male amateur soccer players.

Study 3. *Training effects of the FIFA 11+ Kids on physical performance in youth football players: a randomized control trial.*

- 7. The training stimuli provided by the implementation of the FIFA 11+ kids, two times per week for four weeks (eight sessions), substituting normal warm-up routines in children soccer appears to be sufficient to elicit small to moderate improvements in some (dynamic postural control, agility run, and jumping [standing long jump, CMJ, and DJ]) physical performance parameters.
- 8. The reported improvements in jump performance (tendon stiffness), postural control and agility may partially explain the mechanism behind the recently shown injury prevention effect of FIFA 11+ kids in children soccer players.

6.2 LIMITATIONS AND RECOMMENDATIONS FOR FUTURE RESEARCH

This doctoral thesis is not without limitations. Most of them have been discussed in each of the different studies previously presented (chapters 3 to 5). In addition, this section highlights some general limitations that can be used as a starting point for future research:

- To investigate the training effects of longer interventions using larger sample sizes. Future studies should analyze the long-lasting effects of longer interventions using a large number of participants than that conducted in the current study (>4 weeks and 50 participants per group) on several physical performance and neuromuscular control variables using randomized control trial with placebo control group designs to understand better potential mechanisms behind the reported reduction in injury incidence by the FIFA 11+, Harmoknee and FIFA 11+ Kids programs.
- 2. To examine the efficacy of structured prevention exercises programs for individuals at high and low risk of injury. In order to achieve a better understanding of the training effects elicited by structured prevention exercises programs (e.g., FIFA 11+, Harmoknee and FIFA11+ Kids) on injury risk, future studies should explore whether these programs are more beneficial (or not) for high-risk versus low-risk individuals within at-risk groups, for a range of injury risk factors (e.g., knee abduction moment on landing and kinematics, leg stiffness and unilateral dynamic balance).
- 3. To explore grass-root soccer coaches' reasons for not implementing standardized injury prevention programs in their daily training sessions and matches. Available data consistently indicate that standardized injury prevention programs (including FIFA 11+ and FIFA 11+ Kids) are effective only when high compliance rates are maintained. Coaches play a major role in encouraging and ensuring that participants of their teams adopt appropriate safe practices. Therefore, future studies should identify the main factors and barriers that influence grass-root coaches' decision-making processes for implementing or not injury prevention programs in their soccer teams and explore strategies to better support players and coaches in this issue.

4. To analyze and compare the effects of standardized and individualized injury prevention programs on neuromuscular risk factors and sport performance parameters in young soccer players. It has been suggested that the design and implementation of tailored exercise programs to each player needs (in terms of physical and neuromuscular performance, maturity status and sex) may be a better injury risk mitigation strategy than the use of the standardized injury prevention programs currently available in the literature. Therefore, an interesting future research line may be the study of the efficacy of short-duration exercise programs purposely designed to address each player individual needs in terms of injury risk and sport performance outcomes.



CAPÍTULO 7

Epílogo



and a lot of the second second

(in the

TED N



EPÍLOGO

7.1 CONCLUSIONES GENERALES

Los estudios incluidos en la presente tesis doctoral:

- a) Describen los efectos agudos (post-ejercicio) y crónicos (de larga duración) de las rutinas FIFA 11+, Harmoknee y dynamic warm-up sobre las principales medidas de rendimiento físico y control neuromuscular en jugadores de fútbol jóvenes (adolescentes).
- Analizan los efectos crónicos provocados por el FIFA 11+ Kids sobre medidas de rendimiento físico en jugadores de fútbol niños varones.
- c) Proporcionan argumentos sólidos a favor de la implementación de las estrategias de calentamiento FIFA 11+ Kids y FIFA 11+ como alternativas a las prácticas tradicionales, en las sesiones de entrenamiento diarias y competiciones en jugadores de fútbol niños y adolescentes, respectivamente.

En general, los principales hallazgos de la actual tesis doctoral pueden ayudar a los médicos, entrenadores y especialistas en ciencias del deporte en el proceso de toma de decisiones para la prevención de lesiones y a la optimización del rendimiento deportivo.

A continuación, se resumen las principales contribuciones de esta tesis:

Estudio 1. Efectos agudos de tres estrategias de calentamiento neuromuscular sobre algunas medidas de rendimiento físico en jugadores de fútbol.

- 1. Tanto el programa FIFA 11+ como el Harmoknee provocan efectos agudos similares en varias medidas de rendimiento físico y control neuromuscular a las rutinas de calentamiento que actualmente realizan la mayoría de los jugadores de fútbol jóvenes antes de las sesiones de entrenamiento y partidos.
- El FIFA 11+ y Harmoknee pueden considerarse programas de calentamiento "buenos" o "apropiados" que se deben realizar antes del entrenamiento formal y la competición en el fútbol juvenil, ya que han demostrado que provocan cambios fisiológicos agudos (efectos posteriores al ejercicio) al afectar de

manera positiva a las principales medidas de rendimiento físico (es decir, carreras de velocidad y saltos) y a los factores de riesgo de lesión neuromuscular (es decir, control postural dinámico y ratios de fuerza H/Q).

Estudio 2. Efectos del entrenamiento del FIFA 11+ y Harmoknee sobre varios parámetros neuromusculares de medidas de rendimiento físico.

- 3. Los estímulos del entrenamiento proporcionados por la implementación del FIFA 11+, tres veces por semana durante cuatro semanas (12 sesiones), parecen ser suficientes para provocar mejoras sustanciales en varias medidas de rendimiento físico (tiempo de sprint y altura de salto) y neuromusculares (control postural dinámico, simetría de las extremidades en el salto unipodal [triple salto]) en jugadores de fútbol jóvenes amateurs.
- 4. Las mejoras en la competencia física y control neuromuscular son posibles mecanismos clave para explicar el documentado efecto de prevención de lesiones del FIFA 11+.
- 5. El programa Harmoknee, cuando se realiza con regularidad (tres veces por semana durante cuatro semanas) en sesiones de entrenamiento como sustituto de los calentamientos tradicionales, puede producir mejoras significativas en el rendimiento del sprint y del salto en jugadores de fútbol jóvenes amateurs.
- 6. El FIFA 11+ parece ser más efectivo para reducir los factores de riesgo de lesión (bajo rendimiento físico en el control postural y control neuromuscular) que el Harmoknee y las rutinas de calentamiento tradicionales en jugadores de fútbol jóvenes amateurs.

Estudio 3. *Efectos del entrenamiento del FIFA 11+ Kids sobre el rendimiento físico en jugadores de fútbol jóvenes: un ensayo de control aleatorizado.*

7. Los estímulos del entrenamiento proporcionados por la implementación del FIFA 11+ Kids, dos veces por semana durante cuatro semanas (ocho sesiones), sustituyendo a las rutinas de calentamiento tradicionales en niños parecen ser suficientes para provocar mejoras de pequeñas a moderadas en algunos parámetros de rendimiento físico (control postural dinámico, carrera de agilidad y saltos [salto de longitud de pie, CMJ y DJ]). 8. Las mejoras reportadas en el rendimiento del salto (rigidez de los tendones), el control postural y la agilidad, pueden explicar parcialmente el mecanismo detrás del efecto preventivo de lesiones recientemente demostrado del FIFA 11+ Kids en jugadores de fútbol infantiles.

7.2 LIMITACIONES Y RECOMENDACIONES PARA FUTURAS INVESTIGACIONES

Esta tesis doctoral no está exenta de limitaciones. La mayoría de ellas han sido discutidas en cada uno de los diferentes estudios presentados anteriormente (capítulos 3 a 5). Además de esto, esta sección destaca algunas de las limitaciones generales que se pueden utilizar como punto de partida para futuras investigaciones:

- 1. Investigar los efectos del entrenamiento de intervenciones más longevas utilizando tamaños de muestra más grandes. Los estudios futuros deben analizar los efectos a largo plazo de intervenciones más prolongadas con un número mayor de participantes que el realizado en el estudio actual (>4 semanas y 50 participantes por grupo) sobre varias variables de rendimiento físico y control neuromuscular mediante un ensayo de control aleatorizado con grupo control placebo diseñado para comprender mejor los posibles mecanismos detrás de la reducción informada de la incidencia de lesión por los programas FIFA 11+, Harmoknee y FIFA 11+ Kids.
- 2. Examinar la eficacia de los programas de ejercicios preventivos estructurados para personas con alto y bajo riesgo de lesión. Para lograr una mejor comprensión de los efectos del entrenamiento provocados por los programas de ejercicios de prevención estructurados (por ejemplo, el FIFA 11+, Harmoknee y FIFA 11+ Kids) sobre el riesgo de lesión, los estudios futuros deberían explorar si estos programas son más beneficiosos (o no) para los individuos de alto vs bajo riesgo de lesión dentro de los grupos de riesgo para una variedad de factores de riesgo de lesión (por ejemplo, el momento de abducción de la rodilla y su cinemática al aterrizar, la rigidez de la pierna y el equilibrio dinámico unilateral).
- 3. *Explorar las razones de los entrenadores de fútbol base para no implementar programas estandarizados de prevención de lesiones en sus sesiones de entrenamientos diarias y partidos.* Los datos disponibles indican consistentemente que los programas estandarizados de prevención de lesiones (incluidos el FIFA 11+ y FIFA 11+

Kids) son efectivos solo cuando se mantienen altas tasas de cumplimiento. Los entrenadores desempeñan un papel importante a la hora de alentar y garantizar que los participantes de sus equipos adopten prácticas seguras adecuadas. Por lo tanto, los estudios futuros deben identificar los principales factores y barreras que influyen en los procesos de toma de decisiones de los entrenadores de fútbol base para implementar o no programas de prevención de lesiones en sus equipos de fútbol y explorar estrategias para apoyar mejor a sus jugadores y entrenadores en este tema.

4. Analizar y comparar los efectos de los programas de prevención de lesiones estandarizados e individualizados sobre los factores de riesgo neuromuscular y los parámetros de rendimiento deportivo en futbolistas jóvenes. Se ha sugerido que el diseño y la implementación de programas de ejercicio personalizados para las necesidades de cada jugador (en términos de rendimiento físico y neuromuscular, maduración y sexo) puede ser una mejor estrategia de mitigación del riesgo de lesiones que el uso de programas estandarizados de prevención de lesiones actualmente disponibles en la literatura. Por lo tanto, una línea de investigación futura interesante puede ser el estudio de la eficacia de los programas de ejercicios de corta duración diseñados expresamente para abordar las necesidades individuales de cada jugador en términos de riesgo de lesión y resultados de rendimiento deportivo.





8

REFERENCES

- 1 Ekelund U, Anderssen SA, Froberg K, *et al.* Independent associations of physical activity and cardiorespiratory fitness with metabolic risk factors in children: the European youth heart study. *Diabetologia* 2007;**50**:1832–40. doi:10.1007/s00125-007-0762-5
- 2 Kriemler S, Manser-Wenger S, Zahner L, *et al.* Reduced cardiorespiratory fitness, low physical activity and an urban environment are independently associated with increased cardiovascular risk in children. *Diabetologia* 2008;**51**:1408–15. doi:10.1007/s00125-008-1067-z
- 3 Ortega F, Ruiz J, Castillo M, *et al.* Physical fitness in childhood and adolescence: A powerful marker of health. *Int J Obes (Lond)* 2008;**32**:1–11. doi:10.1038/sj.ijo.0803774
- 4 Janssen I, LeBlanc AG. Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. *Int J Behav Nutr Phys Act* 2010;**7**:40. doi:10.1186/1479-5868-7-40
- 5 World Health Organization (WHO). Benefits of regular physical activity. https://www.euro.who.int/en/health-topics/disease-prevention/nutrition/ahealthy-lifestyle/benefits-of-regular-physical-activity.
- 6 Kjonniksen L, Anderssen N, Wold B. Organized youth sport as a predictor of adulthood physical activity. *Scand J Med Sci Sports* 2008;19:646–54. doi:10.1111/j.1600-0838.2008.00850.x
- Telama R, Yang X, Viikari J, *et al.* Physical activity from childhood to adulthood: A
 21-year tracking study. *Am J Prev Med* 2005;28:267–73.
 doi:https://doi.org/10.1016/j.amepre.2004.12.003
- Federation International of Football Association (FIFA). FIFA Big Count 2006: 270
 Million People Active in Football. *FIFA Commun Div Inf Serv* 2007;**31**:1–12.
- 9 Bangsbo J, Junge A, Dvorak J, *et al.* Executive summary: Football for health prevention and treatment of non-communicable diseases across the lifespan

through football. *Scand J Med Sci Sports* 2014;**24 Suppl 1**:147–50. doi:10.1111/sms.12271

- 10 Malm C, Jakobsson J, Isaksson A. Physical Activity and Sports—Real Health Benefits: A Review with Insight into the Public Health of Sweden. *Sports* 2019;7:127. doi:10.3390/sports7050127
- Howie EK, McVeigh JA, Smith AJ, et al. Organized Sport Trajectories from Childhood to Adolescence and Health Associations. Med Sci Sports Exerc 2016;48:1331-1339. doi:10.1249/mss.00000000000894
- 12 Rice S, Purcell R, Silva S, *et al.* The Mental Health of Elite Athletes: A Narrative Systematic Review. *Sport Med* 2016;**46**. doi:10.1007/s40279-016-0492-2
- Schwellnus M, Soligard T, Alonso J, et al. How much is too much? (Part 2) International Olympic Committee consensus statement on load in sport and risk of illness. Br J Sports Med 2016;50:1043–52. doi:10.1136/bjsports-2016-096572
- 14 Soligard T, Schwellnus M, Alonso J, et al. How much is too much? (Part 1) International Olympic Committee consensus statement on load in sport and risk of injury. Br J Sports Med 2016;50:1030–41. doi:10.1136/bjsports-2016-096581
- 15 Endres B, Kerr Z, Stearns R, *et al.* Epidemiology of Sudden Death in Organized Youth Sports in the United States, 2007–2015. *J Athl Train* 2019;**54**. doi:10.4085/1062-6050-358-18
- 16 Egger F, Scharhag J, Kästner A, et al. FIFA Sudden Death Registry (FIFA-SDR): a prospective, observational study of sudden death in worldwide football from 2014 to 2018. Br J Sports Med. doi:10.1136/bjsports-2020-102368
- DeFroda SF, McDonald C, Myers C, *et al.* Sudden Cardiac Death in the Adolescent Athlete: History, Diagnosis, and Prevention. *Am J Med* 2019;**132**:1374–80. doi:10.1016/j.amjmed.2019.05.025
- 18 Joy E, Kussman A, Nattiv A. 2016 update on eating disorders in athletes: A comprehensive narrative review with a focus on clinical assessment and management. *Br J Sports Med* 2016;**50**:154–62. doi:10.1136/bjsports-2015-095735
- Brenner JS. Overuse Injuries, Overtraining, and Burnout in Child and Adolescent Athletes. *Pediatrics* 2007;119:1242–5. doi:10.1542/peds.2007-0887
- 20 Clark A, Mach N. Exercise-induced stress behavior, gut-microbiota-brain axis and diet: a systematic review for athletes. *J Int Soc Sports Nutr* 2016;**13**:43.

doi:10.1186/s12970-016-0155-6

- 21 Jones S, Almousa S, Gibb A, et al. Injury Incidence, Prevalence and Severity in High-Level Male Youth Football: A Systematic Review. Sport Med 2019;49:1879–99. doi:10.1007/s40279-019-01169-8
- Faude O, Rößler R, Junge A. Football injuries in children and adolescent players: are there clues for prevention? *Sports Med* 2013;43:819—837. doi:10.1007/s40279-013-0061-x
- Krustrup P, Aagaard P, Nybo L, *et al.* Recreational football as a health promoting activity: a topical review. *Scand J Med Sci Sports* 2010;20 Suppl 1:1–13. doi:10.1111/j.1600-0838.2010.01108.x
- 24 Maffulli N, Longo UG, Gougoulias N, *et al.* Long-term health outcomes of youth sports injuries. *Br J Sports Med* 2010;**44**:21–5. doi:10.1136/bjsm.2009.069526
- 25 Schmikli S, Backx F, Kemler E, *et al.* National Survey on Sports Injuries in the Netherlands: Target Populations for Sports Injury Prevention Programs. *Clin J Sport Med* 2009;19:101–6. doi:10.1097/JSM.0b013e31819b9ca3
- 26 Venturelli M, Schena F, Zanolla L, *et al.* Injury risk factors in young soccer players detected by a multivariate survival model. *J Sci Med Sport* 2011;14:293–8. doi:https://doi.org/10.1016/j.jsams.2011.02.013
- 27 Vanderlei FM, Bastos FN, Tsutsumi GYC, et al. Characteristics and contributing factors related to sports injuries in young volleyball players. BMC Res Notes 2013;6:415. doi:10.1186/1756-0500-6-415
- 28 Rössler R, Donath L, Bizzini M, et al. A new injury prevention programme for children's football--FIFA 11+ Kids--can improve motor performance: a clusterrandomised controlled trial. J Sports Sci 2016;34:549-556. doi:10.1080/02640414.2015.1099715
- 29 Robles-Palazón F. Epidemiology and prediction models of injuries in male youth football players (Doctoral Thesis). 2021.
- 30 Renshaw A, Goodwin PC. Injury incidence in a Premier League youth soccer academy using the consensus statement: a prospective cohort study. *BMJ Open Sport Exerc Med* 2016;2. doi:10.1136/bmjsem-2016-000132
- Rumpf M, Cronin J. Injury Incidence, Body Site, and Severity in Soccer Players Aged
 6-18 Years: Implications for Injury Prevention. *Strength Cond J* 2012;**34**:20–31.

- 32 Maffulli N, Longo UG, Spiezia F, *et al.* Aetiology and Prevention of Injuries in Elite Young Athletes. In: *Medicine and Sport Science*. 2011. 187–200. doi:10.1159/000321078
- 33 Soligard T, Steffen K, Palmer D, *et al.* Sports injury and illness incidence in the Rio de Janeiro 2016 Olympic Summer Games: A prospective study of 11274 athletes from 207 countries. *Br J Sports Med* 2017;**51**:1265–71. doi:10.1136/bjsports-2017-097956
- 34 Junge A, Lamprecht M, Stamm H, *et al.* Countrywide campaign to prevent soccer injuries in Swiss amateur players. *Am J Sports Med* 2011;**39**:57–63. doi:10.1177/0363546510377424
- 35 Soligard T, Myklebust G, Steffen K, et al. Comprehensive warm-up programme to prevent injuries in young female footballers: cluster randomised controlled trial. BMJ 2008;337:a2469. doi:10.1136/bmj.a2469
- 36 Kiani A, Hellquist E, Ahlqvist K, *et al.* Prevention of soccer-related knee injuries in teenaged girls. *Arch Intern Med* 2010;**170**:43–9. doi:10.1001/archinternmed.2009.289
- 37 Waldén M, Atroshi I, Magnusson H, et al. Prevention of acute knee injuries in adolescent female football players: cluster randomised controlled trial. BMJ Br Med J 2012;344:e3042. doi:10.1136/bmj.e3042
- 38 LaBella CR, Huxford MR, Grissom J, et al. Effect of neuromuscular warm-up on injuries in female soccer and basketball athletes in urban public high schools: cluster randomized controlled trial. Arch Pediatr & amp; Adolesc Med 2011;165:1033-1040. doi:10.1001/archpediatrics.2011.168
- 39 Mandelbaum BR, Silvers HJ, Watanabe DS, et al. Effectiveness of a Neuromuscular and Proprioceptive Training Program in Preventing Anterior Cruciate Ligament Injuries in Female Athletes: 2-Year Follow-up. Am J Sports Med 2005;33:1003–10. doi:10.1177/0363546504272261
- 40 Malina R, Morano P, Barron M, *et al.* Incidence and Player Risk Factors for Injury in Youth Football. *Clin J Sport Med* 2006;**16**:214–22. doi:10.1097/00042752-200605000-00005
- 41 Rommers N, Rössler R, Verhagen E, *et al.* A Machine Learning Approach to Assess Injury Risk in Elite Youth Football Players. *Med Sci Sport Exerc* 2020;**52**:1. doi:10.1249/MSS.00000000002305
- 42 Rössler R, Junge A, Bizzini M, et al. A Multinational Cluster Randomised Controlled

Trial to Assess the Efficacy of '11+ Kids': A Warm-Up Programme to Prevent Injuries in Children's Football. *Sport Med* 2018;**48**. doi:10.1007/s40279-017-0834-8

- 43 Beaudouin F, Rössler R, aus der Fünten K, *et al.* Effects of the 11+ Kids injury prevention programme on severe injuries in children's football: a secondary analysis of data from a multicentre cluster-randomised controlled trial. *Br J Sports Med* 2019;**53**:1418–23. doi:10.1136/bjsports-2018-099062
- 44 Zarei M, Abbasi H, Namazi P, et al. The 11+ Kids warm-up programme to prevent injuries in young Iranian male high-level football (soccer) players: A clusterrandomised controlled trial. J Sci Med Sport 2020;23:469–74. doi:https://doi.org/10.1016/j.jsams.2019.12.001
- 45 Thorborg K, Krommes KK, Esteve E, *et al.* Effect of specific exercise-based football injury prevention programmes on the overall injury rate in football: a systematic review and meta-analysis of the FIFA 11 and 11+ programmes. *Br J Sports Med* 2017;**51**:562–71. doi:10.1136/bjsports-2016-097066
- 46 Rössler R, Verhagen E, Rommers N, et al. Comparison of the '11+ Kids' injury prevention programme and a regular warmup in children's football (soccer): a cost effectiveness analysis. Br J Sports Med 2019;53:309 LP – 314. doi:10.1136/bjsports-2018-099395
- 47 Steffen K, Myklebust G, Olsen OE, *et al.* Preventing injuries in female youth football
 A cluster-randomized controlled trial. *Scand J Med Sci Sports* 2008;**18**:605–14. doi:10.1111/j.1600-0838.2007.00703.x
- 48 Longo UG, Loppini M, Berton a., *et al.* The FIFA 11+ Program Is Effective in Preventing Injuries in Elite Male Basketball Players: A Cluster Randomized Controlled Trial. *Am J Sports Med* 2012;**40**:996–1005. doi:10.1177/0363546512438761
- 49 Steffen K, Emery CA, Romiti M, et al. High adherence to a neuromuscular injury prevention programme (FIFA 11+) improves functional balance and reduces injury risk in Canadian youth female football players: A cluster randomised trial. Br J Sports Med 2013;47:794–802. doi:10.1136/bjsports-2012-091886
- 50 Owoeye OBA, Akinbo SRA, Tella BA, *et al.* Efficacy of the FIFA 11+ warm-up programme in male youth football: A cluster randomised controlled trial. *J Sport Sci Med* 2014;**13**:321–8.
- 51 Rahlf A, Zech A. Comparison of 10 vs. 20 min neuromuscular training for the

prevention of lower extremity injuries in male youth football: A cluster randomised controlled trial. *J Sports Sci* 2020;**38**:1–9. doi:10.1080/02640414.2020.1776459

- 52 Zarei M, Abbasi H, Daneshjoo A, et al. The Effect of the "11+ Kids" on the Isokinetic Strength of Young Football Players. Int J Sports Physiol Perform 2019;15:1–19. doi:10.1123/ijspp.2018-0827
- 53 Rössler R, Donath L, Verhagen E, *et al.* Exercise-Based Injury Prevention in Child and Adolescent Sport: A Systematic Review and Meta-Analysis. *Sport Med* 2014;44:1733–48. doi:10.1007/s40279-014-0234-2
- Al Attar W, Soomro N, Pappas E, *et al.* How Effective are F-MARC Injury Prevention Programs for Soccer Players? A Systematic Review and Meta-Analysis. *Sport Med* 2016;46:205–217. doi:10.1007/s40279-015-0404-x
- 55 Neto T, Freitas S, Marques M, *et al.* Effects of lower body quadrant neural mobilization in healthy and low back pain populations: A systematic review and meta-analysis. *Man Ther* 2017;**27**:14–22. doi:10.1016/j.msksp.2016.11.014
- 56 Sadigursky D, Braid J, Lira D, et al. The FIFA 11+ injury prevention program for soccer players: A systematic review. BMC Sports Sci Med Rehabil 2017;9:18. doi:10.1186/s13102-017-0083-z
- 57 Bizzini M, Junge A, Dvorak J. Implementation of the FIFA 11+ football warm up program: How to approach and convince the Football associations to invest in prevention. *Br J Sport Med* 2013;**47**:803–6. doi:10.1136/bjsports-2012-092124
- 58 Bishop D. Warm up I: potential mechanisms and the effects of passive warm up on exercise performance. *Sport Med* 2003;**33 (6)**:439-54.
- 59 Bizzini M, Impellizzeri FM, Dvorak J, *et al.* Physiological and performance responses to the 'FIFA 11+' (part 1): Is it an appropriate warm-up? *J Sports Sci* 2013;**31**:1481–90. doi:10.1080/02640414.2013.802922
- 60 Daneshjoo A, Mokhtar AH, Rahnama N, *et al.* The effects of comprehensive warmup programs on proprioception, static and dynamic balance on male soccer players. *PLoS One* 2012;7:e51568. doi:10.1371/journal.pone.0051568
- 61 Silva J, Silva J, Salvador P, *et al.* The effect of FIFA 11+ on vertical jump performance in soccer players. *Rev Bras Cineantropometria Desempenho Hum* 2015;**17**:733–41. doi:10.5007/1980-0037.2015v17n6p733
- 62 Akbari H, Sahebozamani M, Daneshjoo A, et al. Effect of the FIFA 11+ Programme

on Vertical Jump Performance in Elite Male Youth Soccer Players. *Montenegrin J Sport Sci Med* 2018;7:17–22. doi:10.26773/mjssm.180903

- 63 Zarei M, Namazi P, Abbasi H, et al. The Effect of Ten-Week FIFA 11+ Injury Prevention Program for Kids on Performance and Fitness of Adolescent Soccer Players. Asian J Sports Med 2018;9:e61013. doi:10.5812/asjsm.61013
- 64 Trajković N, Gušić M, Molnar S, et al. Short-Term FIFA 11+ Improves Agility and Jump Performance in Young Soccer Players. Int J Environ Res Public Health 2020;17:2017. doi:10.3390/ijerph17062017
- 65 Chen YS, Lai W-L, Hou C-W, *et al.* Effect of post warm-up resting interval on static and dynamic balance, and maximal muscle strength followed by the FIFA 11+ and dynamic warm-up exercises. *J Sports Med Phys Fitness* 2018;**59**. doi:10.23736/S0022-4707.18.08247-6
- 66 Pardos Mainer E, Casajus J, Gonzalo-Skok O. Adolescent female soccer players' soccer-specific warm-up effects on performance and inter-limb asymmetries. *Biol* Sport 2019;36:199–207. doi:10.5114/biolsport.2019.85453
- 67 Parsons J, Carswell J, Nwoba I, *et al.* Athlete perceptions and physical performance effects of the FIFA 11+ program in 9-11 year-old female soccer players: a cluster randomized trial. *Int J Sports Phys Ther* 2019;**14**:740–52. doi:10.26603/ijspt20190740
- 68 Zarei M, Abbasi H, Daneshjoo A, et al. Long-term effects of the 11+ warm-up injury prevention programme on physical performance in adolescent male football players: a cluster-randomised controlled trial. J Sports Sci 2018;36:1–8. doi:10.1080/02640414.2018.1462001
- 69 Bizzini M, Junge A, Dvorak J. 11+ Manual. A complete warm-up program to prevent injuries. *Zurich FIFA Med Assess Res Cent* 2011.
- 70 Grooms DR, Palmer T, Onate JA, *et al.* Soccer-specific warm-up and lower extremity injury rates in collegiate male soccer players. *J Athl Train* 2013;48:782–9. doi:10.4085/1062-6050-48.4.08
- Silvers H, Mandelbaum B, Bizzini M, *et al.* the Efficacy of the Fifa 11+ Program in the Collegiate Male Soccer Player (Usa). *Br J Sports Med* 2014;48:662.1-662. doi:10.1136/bjsports-2014-093494.272
- 72 Van Beijsterveldt AMC, Van De Port IGL, Krist MR, *et al.* Effectiveness of an injury prevention programme for adult male amateur soccer players: A cluster-

randomised controlled trial. Br J Sports Med 2012;46:1114–8. doi:10.1136/bjsports-2012-091277

- 73 Daneshjoo A, Mokhtar AH, Rahnama N, et al. Effects of the 11+ and Harmoknee warm-up programs on physical performance measures in professional soccer players. J Sport Sci Med 2013;12:489–96.
- Impellizzeri FM, Bizzini M, Dvorak J, *et al.* Physiological and performance responses to the FIFA 11+ (part 2): a randomised controlled trial on the training effects. *J Sports Sci* 2013;**31**:1491–502. doi:10.1080/02640414.2013.802926
- 75 Fuller CW, Junge A, Dorasami C, *et al.* 11 for Health, a football-based health education programme for children: A two-cohort study in Mauritius and Zimbabwe. *Br J Sports Med* 2011;**45**:612–8. doi:10.1136/bjsm.2011.084905
- 76 Amiri-Khorasani M, Osman NAA, Yusof A. Acute Effect of Static and Dynamic Stretching on Hip Dynamic Range of Motion During Instep Kicking in Professional Soccer Players. J Strength Cond Res 2011;25:1647–52.
- Brown PI, Hughes MG, Tong RJ. The Effect of Warm-Up on High-Intensity, Intermittent Running Using Nonmotorized Treadmill Ergometry. J Strength Cond Res 2008;22. doi:doi: 10.1519/JSC.0b013e31816a5775
- Fletcher IM, Monte-Colombo MM. An investigation into the effects of different warm-up modalities on specific motor skills related to soccer performance. *J strength Cond Res* 2010;**24**:2096–2101. doi:10.1519/jsc.0b013e3181e312db
- 79 Gelen E. Acute effects of different warm-up methods on sprint, slalom dribbling, and penalty kick performance in soccer players. J strength Cond Res 2010;24:950– 956. doi:10.1519/jsc.0b013e3181cb703f
- 80 Little T, Williams AG. Effects of differential stretching protocols during warm-ups on high-speed motor capacities in professional soccer players. *J Strength Cond Res* 2006;20:203–7. doi:10.1519/R-16944.1
- 81 Sayers AL, Farley RS, Fuller DK, et al. The Effect of Static Stretching on Phases of Sprint Performance in Elite Soccer Players. J Strength Cond Res 2008;22. doi:10.1519/JSC.0b013e318181a450
- 82 Taylor KL, Sheppard JM, Lee H, *et al.* Negative effect of static stretching restored when combined with a sport specific warm-up component. *J Sci Med Sport* 2009;**12**:657–61. doi:10.1016/j.jsams.2008.04.004

- 83 Till KA, Cooke C. Postactivation potentiation on sprint and jump performance. J Strength Cond Res 2009;23:1960–7.
- 84 Zois J, Bishop DJ, Ball K, *et al.* High-intensity warm-ups elicit superior performance to a current soccer warm-up routine. J Sci Med Sport 2011;14:522–8. doi:10.1016/j.jsams.2011.03.012
- 85 Ayala F, Moreno-Perez V, Vera-Garcia FJ, *et al.* Acute and time-course effects of traditional and dynamic warm-up routines in young elite junior tennis players. *PLoS One* 2016;**11**:1–14. doi:10.1371/journal.pone.0152790
- 86 Aguilar AJ, DiStefano LJ, Brown CN, et al. A dynamic warm-up model increases quadriceps strength and hamstring flexibility. J strength Cond Res 2012;26:1130– 1141. doi:10.1519/jsc.0b013e31822e58b6
- 87 Rønnestad BR, Ellefsen S. The Effects of Adding Different Whole-Body Vibration Frequencies to Preconditioning Exercise on Subsequent Sprint Performance. J Strength Cond Res 2011;25. doi:10.1519/JSC.0b013e318215f298
- 88 Sander A, Keiner M, Schlumberger A, et al. Effects of functional exercises in the warm-up on sprint performances. J Strength Cond Res 2013;27:995–1001. doi:10.1519/JSC.0b013e318260ec5e
- Bishop D, Middleton G. Effects of static stretching following a dynamic warm-up on speed, agility and power. J Hum Sport Exerc 2013;8:391–400. doi:10.4100/jhse.2012.82.07
- 90 Samson M, Button DC, Chaouachi A, *et al.* Effects of dynamic and static stretching within general and activity specific warm-up protocols. *J Sport Sci Med* 2012;**11**:279– 85.
- 91 Cejudo A, Sainz de Baranda P, Ayala F, *et al.* Test-retest reliability of seven common clinical tests for assessing lower extremity muscle flexibility in futsal and handball players. *Phys Ther Sport* 2015;**16**:107–13. doi:10.1016/j.ptsp.2014.05.004
- 92 Ayala F, De Ste Croix M, Sainz De Baranda P, *et al.* Absolute reliability of hamstring to quadriceps strength imbalance ratios calculated using peak torque, joint anglespecific torque and joint ROM-specific torque values. *Int J Sports Med* 2012;**33**:909– 16. doi:10.1055/s-0032-1311586
- 93 Dervišević E, Hadžić V, Karpljuk D, *et al*. The influence of different ranges of motion testing on the isokinetic strength of the quadriceps and hamstrings. *Isokinet Exerc Sci*

2006;14:269-78. doi:10.3233/IES-2006-0236

- 94 Impellizzeri FM, Bizzini M, Rampinini E, *et al.* Reliability of isokinetic strength imbalance ratios measured using the Cybex NORM dynamometer. *Clin Physiol Funct Imaging* 2008;**28**:113–9. doi:10.1111/j.1475-097X.2007.00786.x
- 95 Sole G, Hamrén J, Milosavljevic S, et al. Test-Retest Reliability of Isokinetic Knee Extension and Flexion. Arch Phys Med Rehabil 2007;88:626–31. doi:10.1016/j.apmr.2007.02.006
- 96 Croisier J-L, Forthomme B, Namurois M-H, et al. Hamstring Muscle Strain Recurrence and Strength Performance Disorders. Am J Sports Med 2002;30:199–203. doi:10.1177/03635465020300020901
- 97 Shaffer SW, Teyhen DS, Lorenson CL, *et al.* Y-balance test: A reliability study involving multiple raters. *Mil Med* 2013;**178**:1264–70. doi:10.7205/MILMED-D-13-00222
- 98 Gribble PA, Hertel J, Plisky P. Using the star excursion balance test to assess dynamic postural-control deficits and outcomes in lower extremity injury: A literature and systematic review. J Athl Train 2012;47:339–57. doi:10.4085/1062-6050-47.3.08
- 99 Filipa A, Byrnes RN, Paterno M, *et al.* Neuromuscular training improves performance on the star excursion balance test in young female athletes. *J Orthop Sports Phys Ther* 2010;**40** 9:551–8.
- 100 Onate J, Cortes N, Welch C, *et al.* of the Landing Error Scoring System. 2016;**19**:41– 56.
- 101 Flanagan EP, Comyns TM. The use of contact time and the reactive strength index to optimize fast stretch-shortening cycle training. *Strength Cond J* 2008;**30**:32–8. doi:10.1519/SSC.0b013e318187e25b
- 102 Flanagan E, Ebben W, Jensen RL. Reliability of the Reactive Strength Index and Time to Stabilization During Depth Jumps. J Strength Cond Res 2008;22:1677–82.
- 103 Hopkins WG, Marshall SW. A spreadsheet for deriving a confidence interval, mechanistic inference and clinical inference from a P Value. *Sportscience* 2007;11:16– 20.
- 104 Munro AG, Herrington LC. Between-session reliability of the star excursion balance test. *Phys Ther Sport* 2010;**11**:128–32. doi:10.1016/j.ptsp.2010.07.002

- 105 Duthie GM, Pyne DB, Ross AA, et al. The reliability of ten-meter sprint time using different starting techniques. J Strength Cond Res 2006;20:246–51. doi:10.1519/R-17084.1
- 106 Arteaga R, Dorado C, Chavarren J, et al. Reliability of jumping performance in active men and women under different stretch loading conditions. J Sports Med Phys Fitness 2000;40:26-34.http://europepmc.org/abstract/MED/10822906
- 107 Lloyd RS, Oliver JL, Hughes MG, et al. Reliability and validity of field-based measures of leg stiffness and reactive strength index in youths. J Sports Sci 2009;27:1565-1573. doi:10.1080/02640410903311572
- 108 Batterham AM, Hopkins WG. Making meaningful inferences about magnitudes. Int J Sports Physiol Perform 2006;1:50-57.http://europepmc.org/abstract/MED/19114737
- Hopkins WG. Probabilities of Clinical or Practical Significance. Sportscience 2002;6.
 http://www.sportsci.org/jour/0201/wghprob.htm
- 110 Faigenbaum AD, McFarland JE, Schwerdtman JA, *et al.* Dynamic warm-up protocols, with and without a weighted vest, and fitness performance in high school female athletes. *J Athl Train* 2006;**41**:357–63. doi:10.1016/s0162-0908(08)79073-1
- 111 Needham RA, Morse CI, Degens H. The acute effect of different warm-up protocols on anaerobic performance in elite youth soccer players. J Strength Cond Res 2009;23:2614–20. doi:10.1519/JSC.0b013e3181b1f3ef
- 112 Turki O, Chaouachi A, Behm DG, *et al.* The effect of warm-ups incorporating different volumes of dynamic stretching on 10- and 20-m sprint performance in highly trained male athletes. *J Strength Cond Res* 2012;**26**:63–72. doi:10.1519/JSC.0b013e31821ef846
- 113 Bastos FN, Vanderlei FM, Vanderlei LCM, et al. Investigation of characteristics and risk factors of sports injuries in young soccer players: A retrospective study. Int Arch Med 2013;6:2–7. doi:10.1186/1755-7682-6-14
- 114 van der Sluis A, Elferink-Gemser MT, Brink MS, et al. Importance of Peak Height Velocity Timing in Terms of Injuries in Talented Soccer Players. Int J Sports Med 2015;36:e6. doi:10.1055/s-0035-1549903
- 115 Fridman L, Fraser-Thomas JL, McFaull SR, et al. Epidemiology of sports-related injuries in children and youth presenting to Canadian emergency departments from 2007-2010. BMC Sports Sci Med Rehabil 2013;5:1–6. doi:10.1186/2052-1847-5-30

- 116 Hootman JM, Dick R, Agel J. Epidemiology of collegiate injuries for 15 sports: Summary and recommendations for injury prevention initiatives. *J Athl Train* 2007;**42**:311–9.
- 117 Schiff M, Mack C, Polissar N, *et al.* Soccer injuries in female youth players: comparison of injury surveillance by certified athletic trainers and internet. *J Athl Train* 2010;**45 3**:238–42.
- 118 Johnson A, Doherty PJ, Freemont A. Investigation of growth, development, and factors associated with injury in elite schoolboy footballers: prospective study. *BMJ* 2009;**338**. doi:10.1136/bmj.b490
- 119 Brink MS, Visscher C, Arends S, *et al.* Monitoring stress and recovery: New insights for the prevention of injuries and illnesses in elite youth soccer players. *Br J Sports Med* 2010;44:809–15. doi:10.1136/bjsm.2009.069476
- Longo UG, Lamberti A, Maffulli N, *et al.* Tissue engineered biological augmentation for tendon healing: A systematic review. *Br Med Bull* 2011;98:31–59. doi:10.1093/bmb/ldq030
- 121 Impellizzeri FM, Bizzini M, Dvorak J, *et al.* Physiological and performance responses to the FIFA 11+ (part 2): a randomised controlled trial on the training effects. *J Sports Sci* 2013;**31**:1491–502. doi:10.1080/02640414.2013.802926
- 122 Daneshjoo A, Mokhtar AH, Rahnama N, et al. The Effects of Comprehensive Warm-Up Programs on Proprioception, Static and Dynamic Balance on Male Soccer Players. PLoS One 2012;7:1–10. doi:10.1371/journal.pone.0051568
- 123 Brito J, Figueiredo P, Fernandes L, et al. Isokinetic strength effects of FIFA's 'the 11+' injury prevention training programme. *Isokinet Exerc Sci* 2010;**18**:211–5. doi:10.3233/IES-2010-0386
- 124 Daneshjoo A, Mokhtar AH, Rahnama N, et al. The Effects of Injury Preventive Warm-Up Programs on Knee Strength Ratio in Young Male Professional Soccer Players. PLoS One 2012;7:1–7. doi:10.1371/journal.pone.0050979
- 125 Daneshjoo A, Mokhtar AH, Rahnama N, *et al.* Effects of the 11+ and Harmoknee warm-up programs on physical performance measures in professional soccer players. *J Sport Sci Med* 2013;**12**:489–96.
- 126 Kilding AE, Tunstall H, Kuzmic D. Suitability of FIFA's 'The 11' training programme for young football players Impact on physical performance. *J Sport Sci*

Med 2008;7:320-6.

- Harriss DJ, Macsween A, Atkinson G. Ethical Standards in Sport and Exercise
 Science Research: 2020 Update. *Int J Sports Med* 2019;40:813–7. doi:10.1055/a-1015-3123
- 128 Arteaga R, Dorado C, Chavarren J, *et al.* Reliability of jumping performance in active men and women under different stretch loading conditions. *J Sports Med Phys Fitness* 2000;**40**:26–34.
- 129 Hoch M, Silkman C. Balance training for patients with chronic ankle instability. *Athl Ther Today* 2010;**15**:9–11. doi:10.1123/att.15.1.9
- 130 Ayala F, De Sainz Baranda P, De Ste Croix M. Effect of active stretch on hip flexion range of motion in female professional futsal players. J Sports Med Phys Fitness 2010;50:428–35.
- 131 Schmitt LC, Paterno M V., Hewett TE. The impact of quadriceps femoris strength asymmetry on functional performance at return to sport following anterior cruciate ligament reconstruction. J Orthop Sports Phys Ther 2012;42:750–9. doi:10.2519/jospt.2012.4194
- 132 Meyers RW, Oliver JL, Hughes MG, *et al.* Reliability of the Spatiotemporal Determinants of Maximal Sprint Speed in Adolescent Boys Over Single and Multiple Steps. *Pediatr Exerc Sci* 2015;27:419–26.
- 133 Amiri-Khorasani M, Sahebozamani M, Tabrizi KG, et al. Acute effect of different stretching methods on Illinois agility test in soccer players. J strength Cond Res 2010;24:2698-2704. doi:10.1519/jsc.0b013e3181bf049c
- 134 Katis A, Kellis E. Effects of small-sided games on physical conditioning and performance in young soccer players. *J Sport Sci Med* 2009;**8**:374–80.
- 135 Hachana Y, Chaabène H, Nabli MA, et al. Test-retest reliability, criterion-related validity, and minimal detectable change of the Illinois agility test in male team sport athletes. J Strength Cond Res 2013;27:2752–9. doi:10.1519/JSC.0b013e3182890ac3
- 136 Cohen J. Statistical power analysis for the behavioral sciences. 2nd editio. Hillsdale, NJ: Lawrence Erlbaum 1988.
- 137 Arnason A, Sigurdsson SB, Gudmundsson A, et al. Risk Factors for Injuries in Football. Am J Sports Med 2004;32. doi:10.1177/0363546503258912

- Witvrouw E, Danneels L, Asselman P, *et al.* Muscle flexibility as a risk factor for developing muscle injuries in male professional soccer players. A prospective study. *Am J Sports Med* 2003;**31**:41–6. doi:10.1177/03635465030310011801
- 139 Cometti G, Maffiuletti NA, Pousson M, et al. Isokinetic strength and anaerobic power of elite, subelite and amateur French soccer players. Int J Sports Med 2001;22:45–51. doi:10.1055/s-2001-11331
- 140 Impellizzeri FM, Rampinini E, Castagna C, *et al.* Effect of plyometric training on sand versus grass on muscle soreness and jumping and sprinting ability in soccer players. *Br J Sports Med* 2008;**42**:42–6. doi:10.1136/bjsm.2007.038497
- 141 Lloyd R, Radnor JM, Croix MDS, et al. Changes in Sprint and Jump Performances After Traditional, Plyometric, and Combined Resistance Training in Male Youth Pre- and Post-Peak Height Velocity. J Strength Cond Res 2016;30:1239–47.
- Rössler R, Junge A, Chomiak J, *et al.* Soccer Injuries in Players Aged 7 to 12 Years: A
 Descriptive Epidemiological Study Over 2 Seasons. *Am J Sports Med* 2015;44:309–17.
 doi:10.1177/0363546515614816
- 143 Ekstrand J, Gillquist J. Soccer injuries and their mechanisms: a prospective study. 1983;**15**:267–70.
- Gilchrist J, Mandelbaum BR, Melancon H, et al. A Randomized Controlled Trial to Prevent Noncontact Anterior Cruciate Ligament Injury in Female Collegiate Soccer Players. Am J Sports Med 2008;36:1476–83. doi:10.1177/0363546508318188
- 145 Myer G, Faigenbaum A, Ford K, et al. When to Initiate Integrative Neuromuscular Training to Reduce Sports-Related Injuries and Enhance Health in Youth? Curr Sports Med Rep 2011;10:155–66.
- 146 Herman K, Barton C, Malliaras P, et al. The effectiveness of neuromuscular warmup strategies, that require no additional equipment, for preventing lower limb injuries during sports participation: a systematic review. BMC Med 2012;10:75. doi:10.1186/1741-7015-10-75
- 147 Balyi I, Hamilton A. Long-term Athlete Development: Trainability in Childhood and Adolescence. Windows of Opportunity. Optimal Trainability. Victoria: National Coaching Institute British Columbia & Advanced Training and Performance Ltd 2004. 194 p.
- 148 Lloyd RS, Oliver JL. The Youth Physical Development Model: A New Approach to

Long-Term Athletic Development. *Strength Cond J* 2012;**34**:61–72. doi:10.1519/SSC.0b013e31825760ea

- Mirwald RL, Baxter-Jones ADG, Bailey DA, et al. An assessment of maturity from anthropometric measurements. *Med Sci Sports Exerc* 2002;34:689-694. doi:10.1097/00005768-200204000-00020
- 150 Lockie RG, Schultz AB, Callaghan SJ, et al. Reliability and Validity of a New Test of Change-of-Direction Speed for Field-Based Sports: the Change-of-Direction and Acceleration Test (CODAT). J Sports Sci Med 2013;12:88–96.
- 151 Reilly T, Holmes M. A preliminary analysis of selected soccer skills. *Phys Ed Rev* 1983;6:64–71.
- Bosco C, Luhtanen P, Komi P V. A simple method for measurement of mechanical power in jumping. *Eur J Appl Physiol Occup Physiol* 1983;50:273–82. doi:10.1007/BF00422166
- Hopkins WG, Marshall SW, Batterham AM, et al. Progressive statistics for studies in sports medicine and exercise science. *Med Sci Sports Exerc* 2009;41:3–12. doi:10.1249/MSS.0b013e31818cb278
- Prieto TE, Myklebust JB, Hoffmann RG, et al. Measures of postural steadiness:
 Differences between healthy young and elderly adults. *IEEE Trans Biomed Eng* 1996;43:956–66. doi:10.1109/10.532130
- 155 Olmsted L, Carcia C, Hertel J, *et al.* Efficacy of the Star Excursion Balance Tests in Detecting Reach Deficits in Subjects With Chronic Ankle Instability. *J Athl Train* 2003;**37**:501–6.
- Earl JE, Hertel J, Denegar CR. Patterns of Dynamic Malalignment, Muscle Activation, Joint Motion, and Patellofemoral-Pain Syndrome. *J Sport Rehabil*;14:216–33. doi:10.1123/jsr.14.3.216 10.1123/jsr.14.3.216
- Herrington L, Hatcher J, Hatcher A, et al. A comparison of Star Excursion Balance Test reach distances between ACL deficient patients and asymptomatic controls. *Knee* 2009;16:149–152. doi:10.1016/j.knee.2008.10.004
- 158 Bressel E, Yonker JC, Kras J, *et al.* Comparison of static and dynamic balance in female collegiate soccer, basketball, and gymnastics athletes. *J Athl Train* 2007;**42**:42–46.
- 159 Plisky PJ, Rauh MJ, Kaminski TW, et al. Star Excursion Balance Test as a Predictor of

Lower Extremity Injury in High School Basketball Players. *J Orthop Sport Phys Ther* 2006;**36**:911–9. doi:10.2519/jospt.2006.2244

- Clagg S, Paterno M V, Hewett TE, *et al.* Performance on the Modified Star Excursion
 Balance Test at the Time of Return to Sport Following Anterior Cruciate Ligament
 Reconstruction. *J Orthop Sport Phys Ther* 2015;45:444–52. doi:10.2519/jospt.2015.5040
- Hertel J, Braham RA, Hale SA, *et al.* Simplifying the star excursion balance test: Analyses of subjects with and without chronic ankle instability. *J Orthop Sports Phys Ther* 2006;**36**:131–7. doi:10.2519/jospt.2006.36.3.131
- 162 Payne V, Isaacs L. *Human Motor Development: A Lifespan Approach*. London: : Routledge 2016.
- 163 Lloyd RS, Oliver JL, Faigenbaum AD, *et al.* Long-Term Athletic Development, Part
 2: Barriers to Success and Potential Solutions. *J Strength Cond Res* 2015;29:1451–64. doi:10.1519/01.JSC.0000465424.75389.56
- 164 Lloyd RS, Oliver JL, Faigenbaum AD, et al. Long-Term Athletic Development- Part
 1: A Pathway for All Youth. J Strength Cond Res 2015;29:1439–50. doi:10.1519/JSC.000000000000756